# MANAGING QUALITY IN HIGHER EDUCATION: A STAKEHOLDER PERSPECTIVE

Thesis submitted for the Degree of

Doctorate of Education

at The University Of Leicester

by

Shoba Zachariah

**School of Education** 

**University of Leicester** 

October 2007

#### **ABSTRACT**

This study proposes a framework for managing quality in higher education based on the key quality values of students, academics and employers. The proposed framework of the stakeholder determinants of quality (SDQ) is unique in that it is the only approach to quality that seeks to address differences in quality values between stakeholder groups while it builds on similarities in their views. The methodology employs sequential methodological triangulation and includes individual interviews and a larger survey. The sample includes students and academic staff on Business and IT undergraduate programmes from six of the largest private sector higher education institutions in Muscat, Oman and some of the largest private and public sector employers.

The findings reveal strong congruence on many criteria between academic staff and employers, including the importance of developing core transferable skills, student transformation and empowerment and high academic standards. Student responses indicate a lack of congruence on those criteria that focus on student input and participation in the learning process, which are highly rated by academic staff and employers. There are a number of criteria in which there is congruence between the three groups, most significantly, the importance of the teaching and learning function. Students' engagement with the learning process through the lecturers' ability to motivate students' interest, facilitate subject knowledge, stimulate thought and develop transferable skills are considered the most critical issues in managing quality by all three groups. The study suggests that an approach to quality that is based on an understanding of key values of the main participants will facilitate shared understanding and quality consciousness within institutions in comparison to current quality assurance regimes that are externally imposed.

	Content	Page
	Abstract	2
	Acknowledgements	7
Chapter 1	Introduction	8
1.1	Rationale and Context of The Study	9
1.1.1	Conceptual Context	9
1.1.2	Physical Context	11
1.2	Research Aims and Objectives	13
1.3	Structure of the Thesis	14
Chapter 2	A Review of the Literature on Quality in Higher Education	15
2.1	Defining Quality in Higher Education	15
2.2	Stakeholders in Higher Education	22
2.3	Relevance of Existing Quality Management Models to Higher Education	27
2.4	Functional Quality Versus Technical Quality	36
2.4.1	Input	39
2.4.2	Process	43
2.4.3	Output	49
2.5	Studies on Quality Values	53
2.6	Conclusion	58
Chapter 3	Research Methodology	60
3.1	Research Questions	60
3.2	Research Design	61
3.3	Research Methods	63
3.3.1	Qualitative Data: Semi-Structured Interviews	63
3.3.2	Qualitative Data: Sample	65
3.3.3	Qualitative Data: Data Collection and Analysis	66
3.3.4	Quantitative Data: Survey Questionnaire	69
3.3.5	Quantitative Data: Sample	72
3.3.6	Quantitative Data: Data Collection and Analysis	73
3.3.6.1	Consistency estimates	74
3.3.6.2	Kruskal Wallis Test	75
3.3.6.3	Mann Whitney Test	76
3.4	Validity, Reliability, Generalisability and Ethical Issues	76
Chapter 4	Findings And Discussion: Qualitative Data	83
4.1	Emerging Themes	83
4.1.1	Teaching and Learning	85
4.1.1.a	Faculty	86

4.1.1.b	Teaching Methods	88
4.1.1.c	Focus of Lecturers on Subject Knowledge	90
4.1.1.d	Focus of Lecturers on Developing Skills	91
4.1.1.e	Contribution of Other Students	94
4.1.2	Curriculum	95
4.1.3	Learning Resources	97
4.1.4	Outcomes	97
4.1.4.a	Assessment Methods	98
4.1.4.b	Student Performance in Assessments	99
4.1.4.c	Assessing Subject Specific Knowledge versus Assessing Skills	100
4.1.5	Admissions Policies	101
4.1.5.a	Student Aptitude	101
4.1.5.b	Attitude and Commitment of Students	101
4.1.5.c	Range and Variety of Programmes on Offer	102
4.1.5 d	Strict Admissions Criteria	102
4.1.6	Institutional Factors	103
4.1.6.a	Institutional Reputation	103
4.1.6.b	Academic Standards	103
4.1.6.c	Links with Industry	104
4.1.6.d	Extracurricular activities	104
4.1.6.e	Attendance Requirements	105
4.2	Stakeholder Determinants of Quality	105
4.3	Conclusion	108
Chapter 5	Findings And Discussion: Quantitative Data	109
5.1	Profile of Respondents	109
5.2	Quality Dimensions	110
5.3	Quality Criteria	110
5.3.1	Admission Policies	110
5.3.2	Institutional Factors	113
5.3.3	Curriculum	114
5.3.4	Resources	116
5.3.5	Teaching and Learning	117
5.3.6	Outcomes	121
5.4	Cross-Verification	123
5.4.1	Standards versus Meeting Students Expectations	123
5.4.2	Examinations	124
5.4.3	Student evaluation of teaching	125
5.4.4	Institutional versus Student Responsibility	125

5.4.5	Quality of Student Intake	126
5.4.6	Value Added versus High Quality Student Intake	127
5.4.7	Defining Quality in Higher Education	127
5.4.8	Purpose of Higher Education	129
5.5	Differences between Students based on Subject	130
5.6	Differences between Academic staff based on Subject	130
5.7	Discussion	132
5.7.1	Defining Quality in Higher Education	132
5.7.2	Purpose of Higher Education	135
5.7.3	Academic Standards	137
5.7.4	Quality of Intake and the Value Added View of Quality	138
5.7.5	Core Transferable Skills for the Workplace	140
5.7.6	Curriculum	142
5.7.7	Teaching and Learning	144
5.7.8	Assessments	147
5.5.9	Resources	150
5.7.10	Students as Co-Participants	150
5.7.11	Differences between Academic Staff	152
5.8	Conclusion	154
Chapter 6	Framework of Stakeholder Determinants of Quality in Higher Education	155
6.1	SDQ Framework	155
6.2	Student Institutional Socialisation	160
6.2.1	Managing Student expectations	160
6.2.2	Student Motivation	162
6.3	Learner Transformation	162
6.3.1	Curriculum focus on Core Transferable Skills	163
6.3.2	Balance between Subject and Skills by Teaching Staff	163
6.3.3	Balance between Subject and Skills Focus in Assessments	164
6.4	Management Commitment	164
Chapter 7	Conclusion	166
7.1	Significance and Strengths of the Study	166
7.2	Future Implications	170
7.3	Limitations	171
7.4	Scope for Further Research	171
	References	173
Appendix 1	Table 19: Differences between academic staff on subject area: Mann Whitney Test	189
Appendix 2	Interview Schedule	190
Appendix 3	Student Questionnaire	191

Appendix 4	Academic Staff Questionnaire	195
Appendix 5	Employer Questionnaire	199
	LIST OF TABLES	
Table 1	Internal consistency of scales	75
Table 2	Demographic characteristics: Overall	110
Table 3	Demographic characteristics: By Stakeholder Group	110
Table 4	Quality Dimensions: Descending order of Means	111
Table 5	Admission Policies: Kruskal-Wallis test	112
Table 6	Institutional Factors: Kruskal-Wallis Test	113
Table 7	Curriculum: Kruskal-Wallis Test	114
Table 8	Resources: Kruskal-Wallis Test	116
Table 9	Teaching and Learning: Kruskal-Wallis Test	117
Table 10	Outcomes: Kruskal-Wallis Test	121
Table 11	Standards vs. student expectations	124
Table 12	Examinations	124
Table 13	Student evaluation of teaching	125
Table 14	Institutional vs. Student Responsibility	126
Table 15	High Quality Student Intake	126
Table 16	Value Added	127
Table 17	Definitions of Quality in Higher Education	127
Table 18	Purpose of higher education	130
Table 20	Framework of Stakeholder determinants of quality in higher education (SDQ)	156

#### **ACKNOWLEDGEMENTS**

I am extremely grateful to the many individuals who have assisted me in various ways in the last few years. I wish to particularly thank my supervisor Dr. Justine Mercer, who I got to know only during the last one year. However, this period has been the most critical and her advice and guidance, most of which has been online, has been instrumental in completing this thesis. Her encouragement, very constructive and timely feedback on my email queries and drafts essentially allowed me to move forward without any gaps. Thanks are also due to my previous supervisor Dr Hugh Busher, who guided me during the early phase of my research.

It would have been extremely difficult to collect all the data without the assistance of my colleagues Majid El-Shattery, Kasi Vishwanath, Soufiane Blidi and Mohammed Hassan who gave up a great deal of their personal time in helping me distribute and collect the questionnaires. Special thanks are also due to Shamsa Al-Musalmi for so very willingly helping me with the tedious job of data entry, Belema Edet for the final proof reading and Amira Bashar for painstakingly translating the questionnaires. My thanks to my friends and colleagues Dr. Maha Kobeil and Nabila Al-Macki their unwavering confidence in me which gave me the motivation and encouragement I needed in what has been a very lonely journey at times. I would also like to thank Michael Beaumont of the University of Manchester for so willingly acting as a sounding board for my arguments during the process of developing this thesis.

Most of all, I thank my husband and my children for rearranging their lives so that I could complete my work in time. Their forbearance and understanding was what essentially gave me the time and space to complete this.

#### **CHAPTER ONE**

#### **INTRODUCTION**

The last few decades have witnessed the integration of the world economy as a result of multiple factors, including the deregulation of world capital markets, market liberalisation and most importantly, the explosion of worldwide telecommunications. This has in turn created an intensively competitive and dynamic environment which creates often conflicting pressures on organisations to rapidly adapt and respond to changing scenarios (McGregor, 2002). However, while competition has become more intense, resources have become scarcer. Organisations that employ public funds such as educational institutions are no exception, as they face increasing pressures to demonstrate sufficient value in return for resources employed (Pounder and Coleman, 2002).

With increasing levels of awareness and sophistication of local and international communities, higher education providers (HEPs) have to accommodate and satisfy increasing demands from stakeholders. Furthermore, these institutions that were once protected from free market dynamics are now forced to become much more competitive, in order to attract and retain students and to meet performance demands of the global marketplace. For instance, increasing expectations from stakeholders has resulted in increasing calls for radical improvement in graduate business school education (Grey, 2004). Facing escalating costs, resource constraints and increasing levels of competition, HEPs as a response, borrow strategies from the business sector that are greatly influenced by factors such as quality and price in order to survive (Michael, 1997). Consequently, the issue of service quality in higher education (HE) has risen to the top of the research agenda with the main focus on service satisfaction from the perspective of the student (Wright and O'Neill, 2002). Srikanthan and Dalrymple (2007) point to the elaborate measures initiated by many governments to regulate standards and improve quality, as an indication of the desperate need to bring about fundamental improvements in the management of quality in HE.

However, despite the vast amount of literature on the topic of quality, there is very little agreement on the fundamental precepts that underpin quality in HE. Most studies on quality focus on students' levels of satisfaction within particular institutions while the views of

other stakeholders have not been adequately examined. It is apparently taken for granted that as the core issues in managing quality have already been identified by experienced educators and relevant bodies such as accreditation agencies, it is not necessary to validate these assumptions by exploring the views of those who are directly involved or affected by the process of HE.

#### 1.1 RATIONALE AND CONTEXT OF THE STUDY

## 1.1.1 Conceptual Context

Although a relatively new entrant in management science, the management of quality has gained dominance in organisational practice and forms the basis for evaluating organisational excellence (Srikanthan and Dalrymple, 2003). Experts opine that the most powerful trend currently influencing organisational strategy is service quality (Abdullah, 2006). In the modern context in HE, quality management has implications for a whole range of issues including student numbers, curriculum development, student and staff recruitment and retention, teaching and learning, student performance, resource management and faculty productivity. Nevertheless, there is increasing concern that quality assurance requirements are diverting too much of academics' attention from their primary purposes of teaching, research and community service, particularly at a time when pressure on resources is growing (Yorke, 1999).

The implementation of quality driven strategies has an inherent problem in that quality is an elusive construct which is difficult to define and measure accurately (Sahney et al. 2006). Similar to the abstract nature of most services it is difficult for HEPs to differentiate their offerings, thus making it difficult for consumers to evaluate different provisions until the process is completed or acquired (Hill, 1995). In most countries, HE has evolved from a niche-service to one that caters to a mass market, marked by increasing student numbers and diverse providers which has necessitated more appropriate measures of quality (O'Neill and Palmer, 2004). The range of abilities in the class room is now very extensive and students are more varied in their socioeconomic backgrounds, age and experience. HEPs have the difficult task of proving that they can educate larger numbers of students from diverse backgrounds effectively and at lower unit costs while maintaining their integrity and academic standards.

HE has a range of stakeholders of which students form only one part, albeit a primary group (Chua, 2004). The list of stakeholders can easily be extended to include the government, employers, society, professional bodies and accreditation agencies. The complexity increases at higher levels of study, as students themselves become a part of the academic community as in the case of postgraduate students engaged in research and teaching. Telford and Masson (2005) stress that a better understanding of the quality values of other stakeholders particularly staff who design and deliver the programmes are also extremely vital, because, if nothing else, they have an impact on student participation. Through their interaction with students as teachers and facilitators, an institution's academic community will probably influence students' achievements and experiences within the institution to the greatest extent. For an HEP, the employers of its graduates are also key stakeholders who will ultimately benefit from the knowledge and skills of the individuals they employ. Wright and O'Neill (2002) point out that the role of HE as a vital antecedent to career success has been largely driven by employers' increasing demand for a university education as a necessary prerequisite to employment and many, therefore, consider employers as the real customers in HE.

Senge (2000) emphasises that any model for management can succeed only if it is based on the shared values of all stakeholders, particularly the main participants. However, the management of quality covers a broad area and perceptions of quality differ from person to person (Madu, 1998). While some may consider quality in HE to be associated mainly with the quality of teaching and learning (the process), others may consider the quality of the student intake (input) or the knowledge and skills attained by the graduates (output) to be the main criteria (Chua, 2004; Cheng and Tam, 1997). There is a lack of general consensus on how best to manage quality in HE and, there is as yet, no definitive model to evaluate quality within HE (Becket and Brookes, 2005). Research on educational quality also shows a lack of agreement on the dimensions of quality (Abdullah, 2006) and furthermore, although the literature acknowledges that service quality is a multidimensional phenomenon, there is a lack of research on general service quality that reveals the multiple perspectives and reflects the inherent interactive process of all services (Svensson, 2003). Srikanthan and Dalrymple (2007), therefore, call for a fresh approach to developing models for managing quality in HE, as current models are very narrowly focused and based on adapting industry models to HE.

In order to manage quality effectively, an essential first step would involve identifying key quality values of the primary stakeholders, as without some degree of consensus on major quality values, quality management will not be effective in the longer-term. It is entirely possible that perceptions of quality will vary from one group of stakeholder to another. If this is indeed the case, issues on which there are major differences would require further dialogue and discussion. However, as far as the knowledge of the researcher extends, the literature with very few exceptions, has failed to adequately ascertain what staff, students and employers, view as fundamental to quality in HE. Most studies on quality in HE focus on students' expectations and levels of satisfaction within particular institutions, while a few studies explore staff perception on the effectiveness of institutional management and quality. Even among these studies there is very little evidence whether such satisfaction surveys actually measure those aspects or dimensions which are indeed considered relevant by the respondents. One of the few exceptions to this is Telford and Mason (2005) who sought to determine the extent to which students and staff shared the same values within a single institution. Some data is also provided by Hewitt and Clayton (1999) on the expectations of major stakeholders involving a single course of study.

This study, therefore, addresses a gap in the literature by proposing a framework for managing quality that is based on the quality values of students, staff and employers as the core stakeholders who are directly involved in the educational process or are directly affected by the outcome of HE. With the complexities inherent in managing quality in HE and with the increasing pressures on HEPs, this study identifies those quality dimensions on which HEPs must concentrate more efforts on, rather than spreading themselves thinly all around. More importantly, it is argued that identifying the quality values of these stakeholders is critical, as they underpin the motivations, expectations and resultant behaviour of the key participants. A quality framework that addresses the management of key quality values can help develop shared values and quality consciousness within institutions, as opposed to current quality assurance regimes that are imposed by external regulatory bodies.

## 1.1.2 Physical Context

The study has been conducted in Muscat in the Sultanate of Oman which is located in the south-east region of the Arabian Peninsula. The country has a relatively short history in HE

with the establishment of the first university in 1986, similar to other countries in the Gulf region. However, in a relatively short time span the number of local HEPs has grown rapidly from 8 in 1996 to 36 in 2006, the majority of which are privately funded and managed. This rapid growth has been achieved through the government strategy of encouraging the private sector to set up local institutions having academic affiliations with established international universities. The HE sector is governed by the Ministry of Higher Education (MoHE) which regulates all public and private HEPs in the country. In accordance with the regulations of the MoHE, most HEPs offer only undergraduate programmes.

The very short history of HE in Oman and the rapid growth in the number of private HEPs during this period has created a unique environment in terms of academic management and institutional recognition. Particular features are the relatively large number of private sector HEPs and the prevalence of different models of HE in the country, as partner universities are based mainly in the United Kingdom, U.S.A, Australia and Jordan. This creates problems in adopting a common system for institutional or programme accreditation and quality assurance. The local HEPs have to conform to the requirements of their affiliate partners and overseas quality assurance agencies, meet local student, community and industry needs and expectations, as well as conform to the regulations of the MoHE. Given the different models of HE offered by each, the ensuing differences in curriculum and possibly standards and the lack of a common platform for assessing institutions, it is difficult for students and the external community to assess which institutions best meets their expectations and requirements. Currently the MoHE and the Oman Accreditation Council is in the process of establishing a unified system of accreditation and quality assurance based on best practices in Australia and U.K (Carroll, 2006).

In this context and given the lack of research on HE in Oman, any study on the management of quality, particularly those pertaining to private sector HEPs would provide important insights into local issues in HE. This study by seeking to determine the expectations and quality values of key stakeholder groups, which is an area that has not been explored fully even internationally, provides vital background information on relevant quality issues. Hence, its contribution to knowledge is significant and two fold: it seeks to provide an integrated view of quality values in HE from a combined stakeholder perspective which is

not as yet available in the literature and secondly it explores this in a physical context which has been entirely unexplored in terms of research on quality in HE.

#### 1.2 RESEARCH AIMS AND OBJECTIVES

This study aims to identify congruencies and differences in quality values between the three groups of key stakeholders in HE i.e. students, academics and employers, so as to propose a viable framework for managing quality that would address the major expectations of the three groups. The study is located in Muscat, Oman and focuses on a range of undergraduate programmes in Business Studies and Information Technology which are the two most commonly offered subject disciplines in Oman

The key research objectives are:

- 1. To identify criteria that are considered important to managing quality in higher education by academics, students and employers in Oman;
- 2. To determine any differences or congruencies in the values and attributes/criteria of quality among the three stakeholder groups;
- 3. To propose a framework for managing quality based on the stakeholder determinants of quality.

In order to achieve the above objectives, the following research questions have been formulated:

- 1) Which criteria are viewed as specifically relevant to quality in HE by academics, students and employers in Oman?
- 2) What differences or congruencies exist in the quality values and expectations of the three stakeholder groups and what are the implications for HEPs?
- 3) What framework for managing quality can be proposed based on the stakeholder determinants of quality?

## 1.3 STRUCTURE OF THE THESIS

Following the introduction, chapter 2 discusses the existing literature on quality in HE. It examines the various issues that comprise the complex, multifaceted concept of quality and its application to the HE sector.

Chapter 3 discusses the research approach that underlines this study and presents the methodology that addresses the research questions that have been raised. It also addresses various issues regarding the validity and rigour of the study.

Chapter 4 presents and discusses the findings from the in-depth semi-structured interviews conducted with a sample of each stakeholder group. It identifies the quality values of the three stakeholder groups and their views on what is specifically relevant to the quality of HE.

Chapter 5 presents and discusses the findings from the survey conducted with a larger sample from each stakeholder group and examines the differences and congruencies in quality values and perceptions between the groups.

Chapter 6 sets out the proposed framework of quality based on the stakeholder determinants of HE as identified in Chapters 4 and 5.

The final chapter concludes with the major findings and the contribution to knowledge made by this dissertation. It also outlines the limitations of this study and the scope for further research.

#### **CHAPTER TWO**

## A REVIEW OF THE LITERATURE ON QUALITY IN HIGHER EDUCATION

This chapter discusses the existing literature on the concept of quality in HE and the variables that influence this multifaceted issue. The first section examines the different perspectives and definitions of quality. Section two identifies the various stakeholders of HE and section three explores the relevance of existing models of quality management to HE. Section four discusses the literature on technical and functional quality and examines the various input, process and output variables in HE. Next, the findings from the few studies that have been conducted on quality values are discussed. The chapter ends with a summary.

## 2.1 DEFINING QUALITY IN HIGHER EDUCATION

While quality is recognised as a fundamental issue in management theory and practice, there is considerable variation in how it is perceived by different stakeholders and the ensuing implications for organisational performance (Saad and Siha, 2000). Education is a service and services are described as activities or processes that are differentiated from physical goods by virtue of four essential characteristics, namely intangibility, perishability, heterogeneity and inseparability of production and consumption (Hill, 1995). A particular feature of services is that they are behavioural rather than physical entities. This is particularly pertinent to HE which is often described as a process of transformation involving the analytical and critical development of the student (Harvey and Green, 1993). Furthermore, HE involves a highly subjective and intangible product which is the outcome of complex multifaceted service delivery, post-purchase experience and an accumulation of tangible and intangible offerings (Wright and O'Neill, 2002), all of which involve students, faculty, employers and the community, thus making it extremely difficult to evaluate. The quality of the multi-dimensional educational experience is influenced by a myriad of factors and variables, at the end of which the student achieves a profile of knowledge and skills which the world in general will perceive as the quality of the educational provision (Yorke, 1999).

Although an HEP may be defined by the quality of its educational provision, the measurement and evaluation of quality is subject to many different interpretations and problems. A fundamental issue in quality management is the lack of consensus in defining quality, although most people seem to believe they can intuitively recognise quality when they come across it. The ambiguity arises as people perceive quality differently, making it an elusive concept to define (Sahney et al. 2006). Sallis (1993) suggests that quality is a dynamic idea which exerts an emotional and moral influence which makes it difficult to link it to any one particular meaning. Furthermore, the emphasis in the literature on quality is predominantly product-oriented, while service quality has received considerably less attention (Gupta and Chen, 1999). This is unfortunate as service quality in particular is a multifaceted construct and there is a lack of consensus on the various facets of service quality and their interrelationships (Hung et al., 2003).

The main difference between product and service quality is the fact that unlike products, customers do not evaluate service solely on the outcome but consider the process of offering the service as fundamental (Zeithaml et al., 1990) which also makes it more difficult for the customer to evaluate service quality. Edvardsson (2005) highlights the influence of emotions on customers' perceptions of quality and the importance of knowing the positive and negative drivers of customer emotions as customers may respond in various ways. Initially, quality management in the services sector was seen as improving internal processes without considering the impact or interrelationships between the processes and the ultimate customers (Brigham, 1993). Eventually, the focus shifted to the consumer and now, most definitions of service quality are customer focused (Galloway, 1998), so that if consumer expectations are met, service quality is considered satisfactory (Hill, 1995). Perceptions of service quality are, therefore, based on the difference between consumer expectations about the performance of the service and their assessment of the actual experience of the service. The perceived importance of critical service elements to customers and their subsequent satisfaction on these service elements are the criteria for evaluating service quality (Hung et al., 2003). Parasuraman et al., (1994) identify five dimensions or service elements for evaluating general service quality, namely tangibles, reliability, responsiveness, assurance and empathy. Failure to meet customer expectations on any of these dimensions can result in a satisfaction gap.

In order to have clearly defined systems for quality management, it is important to have a clear statement of what exactly is meant by quality (Doherty, 1997). A main contention in defining quality is whether quality is a desired state, as put forth traditionally or whether it is a process, a view which considers the dynamic nature of the industry, market needs and stakeholder expectations (Saad and Siha, 2000). There are various definitions of quality, reflecting different approaches to quality management and taking into account different aspects and perspectives of quality. For example, quality has been associated with offering distinctive or special products or services from a user-oriented perspective. Accordingly, the ISO 8042 defines quality as the totality of features and characteristics of a product or service that helps satisfy particular needs (Yorke, 1999). Although this view of product or service quality focuses on particular needs of users, conflicts may arise in prioritising different needs and wants and in evaluating how they are met. Other related definitions of quality also include: fitness for purpose, conforming to requirements or specifications and achieving excellence (Sahney et al. 2004). Later approaches relate quality as a concept where whole organisations develop the capacity to continually learn and implement customer wants (Harvey and Green, 1993). The emphasis is on quality as a total organisation-wide effort whereby quality should be a way of life which influences the attitude and behaviour of everyone. Quality is thus perceived to be a state of mind and not confined to mere processes or procedures.

When applied to the context of HE, industry based quality concepts present significant limitations and as with other services, are inconclusive (Cheng & Tam, 1997). There is a long-standing debate about the appropriateness of re-defining industrial/business concepts to make them relevant to HE which is perceived as a public good (Harvey and Green, 1993; Campell and Rozsnyani, 2002). There is also a tendency to criticise the emergence of market-led approaches to quality in HE which is held responsible for the increasing emphasis on consumer orientation (Gibbs and Iacovidou, 2004). Perspectives on quality in HE have evolved over the years, ranging from experience to techniques and styles and to process and have been associated with the following definitions (Campell and Rozsnayi, 2002; Watty, 2005):

- being exceptional or distinctive (excellence);
- achieving consistency particularly in process;
- being fit for purpose (conformity to specified objectives or standards);

- being accountable, effective and efficient (providing value for money) and
- being transformative, wherein education is considered an ongoing process of transformation including the empowerment and enhancement of all involved.

From the range of definitions available, it is evident that formulating a single, comprehensive definition is problematic (Sahney et al. 2004). Srikanthan and Dalrymple (2003) note that definitions of quality are stakeholder relative. They mapped the different definitions with the differing priorities and perspectives of each stakeholder group and suggest that the consistency, conformity definitions may be associated with academics and administrators, the value for money and excellence definition would be more relevant to students, sponsors and funding bodies, while the fitness for purpose definition would be of relevance to employers.

Most of the definitions have been criticised for different limitations. The 'consistency in process' definition is criticised on the grounds that it is an insufficient although necessary goal of quality management, leading to sterile and bureaucratic processes stifling creativity and innovation (Doherty, 1997). The efficiency definition considers that when evaluating the process of achieving desired outcomes, one must also consider the extent to which the desired outcomes are achieved (effectiveness) and the efficiency of converting the input into output. However, in HE measuring efficiency and effectiveness can be difficult as many of the important functional aspects are intangible and can vary considerably from one group of students to another and from one tutor to another. The 'fitness for purpose: conformity to predetermined objectives or standards' definition of quality is used extensively in business and has been quite popular in HE as well (Lomas, 2002). Watty (2005) finds that the fitness for purpose definition of quality is the prevailing view of quality amongst accounting academics in Australia. The definition has a strong commercial orientation and assumes that if the product fulfils the purpose for which it is intended its quality is assured. The starting point, therefore, is the purpose for which the product is intended, but for which the definition does not provide any guidance and assumes that any specification would be complete without missing gaps. In reality, where the product or service is complex such as HE, defining the purpose is no simple matter and any assumptions can weaken the product or outcome. This approach to quality is useful if the objectives, standards, specifications and indicators used for judging quality as well as evaluating whether the prescribed objectives

have been attained are clear and accepted by all involved constituencies (Cheng and Tam, 1997). Another view of fitness for purpose is that it can accommodate all other views of quality, where for example, the purpose may be identified as excellence, value for money or transformation (Watty, 2005).

Interestingly, very few of the definitions are really focused on the student, who is arguably the main customer in HE. The 'transformative' definition by Harvey and Knight (1996) which focuses on enhancing the capabilities of participants and ultimately empowering them is one exception. Becket and Brookes (2005) interpret transformation as the critical ability to assess and develop knowledge and observe this is more important to internal stakeholders. A distinctive character of HE is that it is closely aligned with the concept of the learning society, which requires societies to transform themselves in order to prevent decline (Yorke, 1999). The ability to transform learners by enhancing their ability to think for themselves is seen as the highest level of achievement to which HE can aspire (Srikanthan and Dalrymple, 2007). Transformation involves cognitive transcendence and engagement with the meaning of the subject which in turn requires institutional transformation for learning, teaching for transformation and assessment for transformation. In fact, Harvey and Knight (1996) contend that the other characteristics of quality including excellence/high standards, fitness for purpose and efficiency and effectiveness are simply part of the view of quality as transformation. This definition integrates very well with the generally accepted tenet of service quality i.e. customers are active participants of the service delivery process, because although there is considerable debate about students as customers, their transformation does require a very active and joint participation between students and HEPs (Hill, 1995; Williams, 1993).

Another definition which focuses on students is provided by the Quality Assurance Agency for Higher Education (2004, p.1) which defines academic quality as "how well the learning opportunities provided to students enables them to achieve their award". This includes ensuring the appropriateness and effectiveness of teaching, overall support structures, assessments and learning opportunities provided to the students. One criticism of this definition is that it is too general to be readily implemented (Eagle and Brennan, 2007). Cheng's (1995) definition of educational quality in Cheng and Tam (1997) is more comprehensive although still generic and encompasses the whole process of education as well as stakeholders; it states:

the character of the set of elements in the input, process, and output of the education system that provides services that completely satisfy both internal and external strategic constituencies by meeting their explicit and implicit expectations (p.23).

Closely linked to the issue of educational quality is the issue of standards which is another term that is essentially subjective and can have different interpretations (Doherty, 1997). Yorke (1999) makes the distinction that while quality is the totality of all the aspects that influence the students' experience, academic standards refer to the set of expectations about the students' programme of study. Doherty (1997) refers to assessment or output standards i.e. the nature and levels of student attainment required. Lomas and Tomlinson (2000) emphasise that standards are measures of outcome that provides for clear and unambiguous judgments about whether the outcomes are satisfactory. The standards set for a programme of study is inevitably linked to the outcomes and ensure a certain level of knowledge and skills from graduates of that programme. A key characteristic of standards is that they are never static but Morley and Aynsley (2007) note that standards imply standardisation or homogenisation with tacit and explicit understandings of what constitutes desirable graduate qualifications and characteristics. Together with the increasing focus on student satisfaction and massification of HE, there has been increasing assertions of falling academic standards and grade inflation (Clayson and Haley, 2005; Lomas and Tomlinson, 2000). The evidence also indicates that while students consider HE primarily as a route to a career, they are indifferent to whether high standards are maintained (Rolfe, 2002). Instead, it is being increasingly claimed that they now tend to shop around for the easiest courses with the highest grades (Carlson and Fleisher, 2002). However, others such as Marsh and Roche (2000) refute these assertions as they find that students in fact do not rate lecturers who give them lighter work loads positively. They find that although there is a positive correlation between student evaluation of teaching and the grades obtained by students, this is mostly explained by the fact that student perceive they have learned more when they obtain good grades.

Closely linked to the concepts of quality and standards is the purpose of HE. Similar to the problem of defining quality, Doherty (1997) contends that it is impossible to arrive at a single identifiable purpose for any form of education as the needs of the different stakeholders although overlapping in many respects are also different. Generally, the

objective of education may be considered to be the acquisition of knowledge and skills for both intrinsic and instrumental purposes (Eagle and Brennan, 2007). Heyneman (2006) emphasises the role of private and public HE in enhancing societal cohesion and in ensuring that their graduates are able to live up to the expectations of the labour market and provide value to the community in general. He suggests that the more a university exhibits good behaviour and professional standards, the more likely will its students contribute to social capital that is tolerant, understands diversity and is willing to work towards a common good. Wals and Jickling (2002) contend that a university's role is to develop dynamic qualities in students that allow them to be critical and to act with a high degree of autonomy and determination at least in their professional lives. They argue that educators must seek more diversity of thought in order to ensure sustainability and meaningful learning that will enable students to cope with poorly defined situations and conflicting or diverging values and interests. Eagle and Brennan (2007, p.49) suggests that the goal of HE is to develop graduates with the "ability to think critically and laterally, to solve problems creatively, to adapt to change, and to understand the social dynamics of the organisations in which they will work". If the objective of HE is to enable students to engage in effective actions in increasing uncertainty, quality systems have to identify those features which develop this characteristic in all of its programmes (Srikanthan and Dalrymple, 2002). For the effective management of quality of education, the objective of providing that education would have to be clearly understood, even if there are a range of objectives to fulfil.

The fact that there does not seem to be any agreed definition of quality is taken as indicative of the problems inherent in deciding appropriate measures of assessing quality. However, Garvin (1988) writes that reliance on a single definition for quality can be a source of conflict and can result in communication problems. Indeed, it may be a futile exercise to seek a single best definition of quality as it is not a 'unitary concept' but must be defined in terms of 'qualities' (Green 1994, p. 17). Although quality will always be subject to varying interpretations, most of the definitions have various points of similarities between them (Cheng and Tam, 1997). Certainly, the complexity and multi-faceted concept of quality in HE may not be best described by a single, definition and cannot be easily assessed by only one indicator. It would be productive to determine the views of different stakeholders when defining and evaluating quality so as to allow potentially different but legitimate views to be expressed. Meeting such expectations may, nevertheless, be particularly challenging as each

stakeholder group may have contradictory priorities and expectations based on their different modes of interaction with the institution (Mahapatra and Khan, 2007).

#### 2.2 STAKEHOLDERS IN HIGHER EDUCATION

Stakeholders in HE include students, teaching as well as non-teaching staff, employers, government and other funding agencies, accreditation bodies and the general community, each with their own criteria and perspectives (Telford and Masson, 2005). Amongst these, the more significant stakeholders are those who either have an effect on the process or outcome of the service or are directly affected by it. The primary internal stakeholders, therefore, would invariably be those being educated (students) and the educators (academic and management staff) as they participate in or are responsible for and influence the process of education. Furthermore, employers of graduates are external stakeholders who are ultimately and directly affected by the outcome of HE and, therefore, must be considered as a key stakeholder (Hewitt and Clayton, 1999).

The focus on these three main stakeholder groups in this study is not to understate the importance of other external groups such as the government, families of students and the society who have legitimate interests in HE. As Eagle and Brennan (2007, p. 48) argue, even if both students and employers considered the role of HE was to support the economy by preparing graduates for jobs, this would not be the only legitimate purpose of HE. They write that:

the community which contributes to higher education through general taxation, may reasonably suppose that it is the purpose of higher education to produce well-rounded citizens who are sensitive to the needs of vulnerable groups and who may be prepared to sacrifice some self-interest for the common good.

Government agencies and funding bodies are also extremely relevant as they may often have a direct or indirect effect on funding, licensing and approvals; nevertheless their main role being more regulatory in nature are not in the same category as students, staff and employers.

References to students in the quality assurance literature range from that of customer, consumer, partner, participant and stakeholder and it is now widely acknowledged that

students as stakeholders have an increasingly powerful influence in the process and outcomes of HE (Johnson and Deem, 2003). Definitions of quality in HE consider students as two distinct analogies: one as that of a customer who buys a service in expectation of career benefits and the other as that of raw material that will be transformed by the process of HE into an individual with added skills (Eriksen, 1995). In the latter view, the primary input is the student who is subjected to a transformation process (the value adding process of HE) which in turn produces an output (the student after exposure to a value-added service). A distinctive feature of many services is that customers are also partial employees as they are intensively involved in the production and delivery aspects of the service (Hill, 1995) and this is exemplified by students as processors of information (Williams, 1993) and co-producers (Hill, 1995) particularly in the case of post-graduate students.

The conceptualisation of quality as the difference between students' expectations and their perception of their actual experience is considered by many as a valid and reliable indicator of educational quality (O'Neill and Palmer, 2004). However, while service quality and customer orientation are the focus of the service industry, there are many who question whether students should be considered as customers and indeed whether this can be an appropriate approach (Gibbs and Iacovidou, 2004). Johnson and Deem (2003) find that the concept of the student-consumer is largely resisted by academic managers. They observe that any attempt to present students as customers requires careful articulation of students' needs, which then becomes the terms of reference by which the customer concept is created, defined and asserted. This requires redefining and clarifying students' identity, rights and status in relation to the academic/professional, which results in tensions between management and academic staff.

There are fears that in an environment where student expectations are the focus, learning, curriculum and programme quality would suffer. Barnet (1997) argues that HE is now a large-scale service industry, increasingly embracing the concept of customer care despite the opposition and ongoing debate on the mismatch between the customer-centered approaches and traditional academic values. Eagle and Brennan (2007) point out that while students may consider gaining an advantage in terms of their career as a key indicator of quality, they may not really consider high academic standards as representative of high quality or as essential for career advancement. While emphasising that HEPs must ensure that the implicit and explicit needs of students and other stakeholder groups are met, they

point out that there is some currency in the "notion that students are simply in the HE system to acquire a qualification and that any education picked up along the way is incidental to this primary aim" (p.44). The resistance to the student-customer concept may result from the concern that it legitimises all student demands that HEPs will then have to satisfy. Education cannot be treated as a mere transaction involving the payment of money for a service rendered, even if one ignores all other complexities and considers only the fact that HEPs are required to regulate standards for their awards which involves not rewarding those students who fail to meet these standards.

Arguably, as students bear the larger proportion of the costs of HE in many countries, this gives them the right to be treated as customers (Eagle and Brennan, 2007). This then raises the concern that students as fee paying customers will take less responsibility for their own learning and will place the responsibility for their failure or poor performance on the HEP (Clayson and Haley, 2005) or more precisely their tutors, an expectation which would also have to be managed effectively by HEPs. Nevertheless, there is evidence of a direct link between student satisfaction and retention which suggests that an institution must focus more attention on student satisfaction if it wishes to reduce drop out rates (Wright & O'Neill, 2002). As Basave (1998) argues, if an educational system does not take into account the diversity and the aspirations of those for whom it is designed, it goes against intellectual and cultural values. Thus, discerning and managing student expectations have become an important aspect of managing the quality of the student experience. In fact, Wright & O'Neill, (2002) recommend that students should be involved in all stages of the HE service design process so that their needs can be properly ascertained and provided for. Furthermore, although students' needs may remain the same, the relative priority of the needs may change over time which requires the delivery process to reflect this change. It is also important to recognise that students are no more homogeneous than any other group of stakeholders and there will be significant differences between them. Undoubtedly, students will have different objectives and sets of expectations from HE, a fact which is under recognised by academics and administrators (Eagle and Brennan, 2007).

However, while students as primary customers have the right to obtain best quality education, the fact is that students may not really be in a position to evaluate or comprehend what constitutes a 'good' course or learning experience in terms of content and outcomes in the longer term, in relation to their immediate experience of it. Education may have the

distinction of being the only service where it is difficult for the customer to assess the quality and relevance of the service, as students may not fully comprehend the relevance of a course until later years of study (Dickson et al., 1995). For instance, new students fresh out of secondary school may confine their objective to acquiring a qualification for employment without realising the importance of developing the skills required for personal and professional development. Therefore, Amaral (2006) stresses that students may be considered only as immature consumers who are not in a position to rationally evaluate data on future benefits accruing from a programme of study.

Clearly, the notion that the customer is always right cannot be taken literally in HE as this can corrode the acquisition of necessary knowledge and skills and, therefore, harm the interests of students themselves. It is also necessary that students understand their role in the learning process and the role of academic and administrative staff in facilitating learning opportunities (Lammers et al., 2005). Academics may accept that students are in some sense customers, provided that this is interpreted with a degree of sophistication, as there would be few arguments against an approach to educational delivery in which the learner is the central focus. Hence, while the concept of students being a simple customer purchasing a service offers little relevance to HE, students may be thought of as professional customers of an intangible service having uncertain outcomes and involving a lengthy interactive process which requires their active participation.

Others consider the student as the product while the employer is the real customer of the product (Bailey & Bennett, 1996). The value placed on HE as a vital antecedent to career success has been largely driven by the increasing demand for university qualifications by employers (Wright and O'Neill, 2002). But there are others (e.g. Gibbs and Iacovidou, 2004) who are extremely critical of attempts to include employers and employable skills in the realm of HE. They consider this as unnecessary and even dangerous as it can threaten the academic independence and credibility of HE, which should be grounded only in critical thinking, tolerance and self-development, and not in the needs of others such as employers. This again reflects the traditional view that education is intrinsically different from other services and, therefore, metaphors related to the market place are harmful to the educational process. Macfarlane and Lomas (1995) highlight the growing tendency for large employers to negotiate with HEPs to provide programmes which are tailored to meet their particular needs and demands. They argue that in the pursuit of the financial rewards that such

partnerships offer, concerns about the quality of students' educational experience and fundamental academic values, including academic freedom and the development of students' critical abilities and autonomy are overlooked.

Nevertheless, there can be no doubt that employers can greatly influence the future of HEPs and their graduates through their recruitment policies. Jameson and Holden (2000) emphasise that being a graduate will have little social significance unless employers have an understanding of how graduates can contribute to the workplace. Morley and Aynsley (2007) find that employers in the UK can undermine equity and widening participation initiatives in HE by restricting their recruitment to elite HEPs. They attribute some of the reason for the increasing significance given to league tables to the demand for elite educational credentials by employers. There is also evidence of employers expressing increasing dissatisfaction with the skills displayed by HE graduates (Johnson and Spicer, 2006). Thomas (2007) highlights an article in the Economist (2006) which indicates dissatisfaction on the part of employers about the quality of recruits and the time taken to find suitable job candidates. Rhodes and Shiel (2007) observe that in order to address the current knowledge and skill deficit in the UK, establishing links between HEPs and employers is given high priority by the government. As a result of the growing frustration with the quality and content of tertiary education, large organisations now depend on their in-house training facilities (termed corporate universities) to provide for their strategic human resource development needs (Holland and Pyman, 2006). In the current economic climate, it is becomingly increasingly more apparent that what constitutes a graduate job has broadened and that it is probably too ambitious for fresh graduates to aspire to get directly into permanent, professional level employment (Jameson and Holden, 2000). This in turn has a significant impact on what employers seek from HE and HEPs cannot afford to ignore their expectations as probably the most important reason for students seeking HE is enhanced career prospects.

All quality management models stress the commitment and motivation of staff. The role of academic staff as a key stakeholder having a direct influence on the overall process and outcome of HE cannot be over emphasised. In addition to the appropriate blend of factors such as curriculum and classroom infrastructure, the enthusiasm, expertise and teaching style of instructors are vital to learning, as they determine to a great extent the outcome and the overall experiences of students. It is well established that the enthusiasm and motivation

of lecturers translates to high levels of student motivation and learning (Hill et al, 2003). Anderson (2000) emphasises the role of lecturer and student interaction and the passion and enthusiasm conveyed by the lecturer in enhancing students' engagement with the subject. High levels of staff motivation also correlate positively with professional satisfaction and the overall quality of services offered (Konidari and Abernot, 2006). However, Lammers and Murphy (2002) in Hill et al. (2003) find that while lecturers had a role in giving information, they do not necessarily stimulate thought, change attitudes or develop behavioural skills that are necessary for complex interactions essential in HE. Hence, in order to be effective, educators must use their judgement, rationality and decision making abilities rather than rely on routine (Hill et al. 1996).

A shared awareness of common goals allows an organisation to work collectively rather than as multiple separate units and, thus, fosters trust among participants (Srikanthan and Dalrymple, 2007). Such a collective consciousness emerges when different people share awareness of the same issue from a variety of perspectives and are conscious of others' view points and also allows the organisation to be flexible and dynamic, which is a necessary condition in an increasingly dynamic environment. It would be logical that an appropriate model for managing quality in HE considers the perspectives and quality values of students, academic staff and employers as three of the main stakeholder groups.

## 2.3 RELEVANCE OF EXISTING QUALITY MANAGEMENT MODELS TO HIGHER EDUCATION

The concepts and practices of quality management in HE have only been implemented for just over a decade (Srikanthan and Dalrymple, 2003) and there is still no agreement on a common model for quality in HE (Srikanthan and Dalrymple, 2007). The literature advocates that the management of quality in HE should be handled differently from that of other services (Mizikaci, 2006). While most quality management models originate from and focus on manufacturing and general services, it is generally held that HEPs cannot be compared to business units and any concept derived from commercial enterprises is inappropriate for HE. The central argument revolves around differences in terms of the public sector orientation of most HEPs and the commercial orientation of business organisations (Cuthbert, 1996a). Since HEPs are public sector organisations (traditionally), they are generally considered to have a moral dimension to their service which is to provide a benefit to the society that is not rooted in economic benefits (Drucker, 1994; Roffe 1998).

More recently, Eagle and Brennan (2007) highlight the view that while elementary education has a substantial public good component, HE is to a very large extent a private good with the benefits accruing mainly to students through enhanced earnings. Nevertheless, they do note that HEPs cannot function solely on the premise that HE is a purely private good, as graduates have a vital impact on the wider community and the economy.

Other constraints in applying generic quality management models to HE include, the difficulty of considering students as customers who must be satisfied (Chua, 2004); the interactive, multidimensional, varied and lengthy process of education which is fundamentally different from a commercial process (Tribus, 1996) and too subtle for meaningful measurement (Harvey, 1995); and the difficulty and inappropriateness of reducing variation in educational processes and outcomes (Srikanthan and Dalrymple, 2007). Specific challenges also include problems arising from over-regulation and control of academic freedom. Saunders & Walker (1991) argue that the greatest challenge of applying quality management models to HEPs is the difficulty of identifying an appropriate management structure that would not restrict the diversity, innovation and creativity of academic institutions.

However, such arguments do not necessarily provide insurmountable reasons for the unsuitability of industry based models to HE, considering that HEPs are defacto business entities although perceived as being established to serve the public. HEPs, despite or because of their moral and social obligations are no exception to the increasing requirements for efficiency and accountability. While most business organisations understand the need for incremental improvement of activities to eliminate waste and create more value, Emiliani (2005) points out that HEPs continue on processes that consume resources but do not create value. There is also enough evidence that HEPs have begun to successfully and constructively apply ideas taken from the business world including manufacturing and service industries (Doherty, 1997). Deem and Brehony (2005) note that the concept of managerialism is now widely prevalent in UK universities and public sector organisations and includes business models such as the efficiency model, decentralisation and the learning organisation. They observe that each model represents management responses to changes in the macro and micro environment of HE and has resulted in various changes including external accountability, wide-spread use of performance indicators,

league tables, target-setting, benchmarking and performance management. The rise of managerialism in higher education is often criticised and Santiago & Carvalho (2003) argue that managerialsim is usually justified by the two-fold argument that HEPs are not capable of renewing themselves at the same pace as environmental changes and that traditional collegial decision-making bodies tend to perpetuate the interests of academics which creates irrationalities and inefficiencies. Others such as Trow (1994, p. 11) advocate a role for 'soft' managerialism which still views HE as autonomous and governed by its own norms but with a more effective and rationalised management. However, Kanji & Malek, (1999) contend that academics consistently and without adequate rationale, oppose what they see as the introduction of new management techniques as well as any external interference on the institution. Moreover, ultimately educational institutions just like other modern day organisations are caught in the cross currents of change and have to be innovative in order to meet the challenges of the new information age (Senge, 2000). Although it is without doubt important to understand the specific complexities involved in managing quality in HE, the presence of such complexities should not cloud the fact that many management concepts and strategies essentially underpin the long-term survival of an organisation and hence can be useful to any HEP with a mission to fulfil.

There are numerous industry based models of quality management such as Total Quality Management (TQM), ISO 9000, and the European Foundation for Quality Management (EFQM) excellence model, all of which have been applied to HE with at best, inconclusive results (Sahney et. al., 2006). Some universities have developed TQM based strategies by introducing the concept of quality circles involving academic and non-academic staff in order to build a culture of continuous improvement (Lomas, 2004), while others have applied the EFQM model to achieve institutional excellence (Pupius, 2001). The EFQM excellence model is a non-prescriptive framework which includes a number of fundamental elements such as results orientation, customer focus, process management, people management, continuous learning, innovation, partnerships and public responsibility. The entire process is driven by self-assessment, which should lead to improvement activities aimed at the ultimate goal of excellent results, which are defined as those which show positive trends, sustained good performance or meet appropriate targets (Jackson, 2000).

Doherty (1997) notes that the term TQM is used generally as a carpet bag that refers to more than one particular approach and includes those which may emphasise different

aspects of quality improvement. TQM is a combination of quality control and quality assurance and improvement (Hoyle, 1994). All TQM approaches espouse the philosophy of continuous improvement, which is achieved by relying on a motivated and cooperative workforce who are individually and in groups responsible for achieving excellence in their work by contributing to improvements (Wiklund et al., 2003). The development of a continuous quality culture finds approval with many academics (Doherty, 1997). However, despite its theoretical promise, in practice, TQM benefits are not easily achievable and attempts to implement it are often unsuccessful (Rad, 2005). Aly and Akpovi (2001) note that while many American universities have applied TQM initiatives in an effort to remain competitive in the face of rising costs and increasing public demands for better quality, many have not been able to sustain it. Cruickshank (2003) observes that the application of TQM principles to HE was intended to make HEPs more relevant and responsive to the needs of employers and others including funding agencies. However, the application of TQM in universities is heavily criticised mainly on the grounds that reformation is claimed by focusing only on non-academic support functions such as student admissions, administration and management of funds (Sohail et al., 2003). It is also argued that while models which have their roots in manufacturing can be easily applied to mass services requiring comparatively less contact with customers, it may be more difficult to do so with more complex professional services such as education (Silvestro, 2001). More specifically, the application of TQM in HE is considered inappropriate as it is based on the principle of customer satisfaction, while HEPs are not in the business of delighting students who are their primary customers (Chua, 2004). A practical way of looking at the issue of customers in HE is to treat it as a wider concept involving all relevant stakeholders and ensuring adequate recognition and management of student expectations although this does not necessarily entail meeting all such expectations.

More significantly, attempting to implement industry based quality management across all operations of an HEP is considered by many to be futile, as the teaching and learning function is too complex. A pertinent weakness of industry models is the undue focus on measurement, whereas the subtle process of teaching and learning does not lend itself to meaningful measurement (Srikanthan and Dalrymple, 2007). Yet another criticism is that the focus of such approaches is on a homogenous, zero-defects approach and reducing costs through reducing errors rather than enhancing quality which becomes only a subordinate function. It is a fact that product control, minimising variation and avoidance of error which

are critical aspects of industry based quality management models, cannot be applied literally to HE with its varied products and processes. Nevertheless, what should be considered is whether the application of such models in HE necessarily implies homogeneity and a lack of originality in outcomes, or whether they can help develop a sense of ownership and commitment among employees towards identifying and solving problems before they affect outcomes.

Whether they are derived from industry or elsewhere, practical ideas for managing and improving quality always merits serious analysis (Roffe, 1998). On closer examination, the major criticisms do not really indicate a complete lack of applicability of these models to HE. Yorke (1999) cites the example of ISO 9000, when he argues that although there were expressions of interest in such models in the early 1990s, very little has actually been done to explore how they can be useful to HE. The literature does reveal that TQM approaches offers potential benefits for the core functions in HE including the teaching and learning process, curriculum development and research activities (Vazzana et al., 1997). Eriksen (1995) contends that TQM is especially relevant to HE as it is based on the simple but powerful fundamental belief that it is better all round to ensure that things are done right the first time rather than adopt after-the-fact controls by which time it is too late to rectify problems. As Brigham (1993) recommends, models such as TQM could have major benefits to academic institutions, if HE is not considered as totally unique and isolated from other industries. Indeed, the most persuasive feature of TQM is its role in building a sense of identity and commitment within the organisation where staff realise that the success of an individual contributes to the success of the whole organisation (Williams, 1993). Furthermore, the concept of continuous quality improvement is based on the principle that only those involved in carrying out a process are fully capable of measuring its characteristics (Doherty, 1997). Such an approach has strong parallels with Senge's concept of the learning organisation, which is increasingly being suggested as a positive strategy for implementing growth and change in HE (Konidari and Abernot, 2006). Wu (1996) defined a learning organisation as one that cultivates life-long learning in its members, encourages personal potentiality, accommodates changes and develops continuously (as cited in Hsieh, 2005). In order to develop a learning organisation, members must have a strong willingness to deepen and increase the course of their learning (Hsieh, 2005) and in a HE setting requires collaborative learning opportunities that can benefit students and employees. Collaboration between staff and students, effective communication and cooperation between managers and those being managed as well as between internal service providers and external scrutineers are vital. However, much of the literature on universities as learning organisations overlooks the critical role of students' participation while others point out that there may be competing views of what constitutes important learning in a university (Gouthro et al., 2006).

Lomas (2004) emphasises that any quality initiative is unlikely to lead to success unless the selected strategy is appropriate for the organisation and its organisational structure and communication channels allow quality initiative(s) to be disseminated effectively. Most importantly the organisational culture must be conducive in that it facilitates the discussion, assimilation and implementation of new ideas. According to Colling and Harvey (1996), TQM approaches require HEPs to adopt key principles for procedure and team approaches which are not endemic across institutions. They note that academic course teams rarely behave as teams and often individual autonomy is prized above all else, often leading to inefficient and ineffective management of the process and outcomes of teaching and learning. Ironically, Aly and Akpovi (2001) find that the main difficulty in applying TQM in American universities is the resistance to change by academics and administrators, followed by the lack of leadership. Interestingly Sitkin et al. (1994) observe that the TQM philosophy which is based on the concept of employee empowerment faces opposition from academics who, as the most empowered of employees find it difficult to reconcile empowerment and control as a means of achieving organisational objectives.

Changing such a scenario and embedding a continuous quality culture and a sense of collective responsibility requires strong commitment from management. However, the lack of vision and top management commitment has often been cited among the main reasons for failure in implementing quality control mechanisms in organisations (Saad and Siha, 2000). Despite the increased importance of the role of academic managers, very few of them actually have the experience and skills required for such positions and this coupled with the turbulent and continuous expansion that HEPs find themselves in, has lead to falling levels in morale of lecturers and academic managers (Holmes and McElwee, 1995). Significantly, while managers in other fields, both public and private are more likely to have received extensive training, this is not generally the case with academic managers whose legitimisation is often based on their academic status (Deem and Brehony, 2005).

Srikanthan and Dalrymple (2007) observe that models for managing and enhancing different aspects of educational quality have either looked at improving the teaching and learning function or at providing more conducive management methodologies. They argue that such models which look at the teaching and learning function alone are flawed in their approach as both the pedagogical and service aspects of HE should be addressed. They propose a "holistic" model for quality management which addresses the service areas as well as the core areas of teaching and learning and has at its core the transformation of learners, proactive collaboration at the learning interface and commitment at all levels, supported by management (p.266). The core elements of their Quality Management in Education (QME) model include transformation of both learners and the institution, synergistic collaboration at the learning interface and significant commitment by all individuals. Therefore, the model is appropriately designed for HE. However, although Srikanthan and Dalrymple (2007) argue that the shared vision of all stakeholders is very important for any quality model to succeed, there is little evidence that the QME model encompasses different perspectives other than the educational dimensions already present in the literature.

The differing views of quality have also resulted in different evaluation measures and a range of performance indicators (PIs) (Cullen et al. 2003). The prevalence of PIs is common in HE but according to Soutar and McNeil (1996) PIs are more measures of activity rather than true measures of the quality of educational service. The accreditation system in the United States has identified a broad range of PIs including access, productivity and efficiency, student learning, degree completion, and economic returns (Schray, 2006). The Higher Education Funding Council for England (HEFCE) has provided PIs which measures access to HE, non-continuation rates, completion rates and research outputs. According to Pursglove and Simpson (2007), PIs are lag indicators which at best reflect outcomes of past activity without indicating the specific causes of any deterioration or what action must be taken by HEPs to achieve desired outcomes. Moreover, the collection of data required for generating PIs has resulted in more administrators, documentation and increased levels of bureaucracy, while stifling innovation and creativity (Bruneau and Savage, 2002).

League tables and rankings also provide a basis of comparison of quality between HEPs for external stakeholders. University rankings generate a lot of interest and while the media feel the rankings have added positive value, academia, on balance, feel the opposite (Policano,

2007). Academics tend to regard such tables as based on statistical indicators which attempt to reduce complex human interaction and experiences into mere numbers (Doherty, 1997). Doherty points out that covert league tables in HE were based on the assumption that the main aim of HE was to provide entry to a research career which is not a viable assumption in the current context of mass HE. Policano (2007) argues that with the proliferation of rankings which is essentially media driven in order to boost circulation of various publications, the value of rankings have diminished, although the overall impact remains significant. The most important role of rankings is the information provided to prospective applicants about HEPs in general; however they do not necessarily provide an accurate indication of the quality of specific programmes nor does it indicate that HEPs not included in the rankings provide poor quality. Yorke (1997) writes that as overall rankings do not represent specific conditions in particular academic units, students will be better served by providing data on what each unit is trying to achieve, and the extent to which it is succeeding. One effect of rankings has been to influence applicants to place more emphasis on variables like "perceived prestige rather than on variables like the quality and relevance of the curriculum" (Policano, 2007, p.43). Rather than rankings, Policano recommends that HEPs should be rated, for example on a 1-5 star basis so that students have an idea of key institutional factors but would also seek more detailed information about the appropriateness of the programmes offered by the institutions within a certain rating scale.

In contrast to the management of quality in industry, HE has largely favoured external monitoring of quality through accreditation and quality audits with the objective of monitoring the institutions' performance against stated goals and key performance indicators. Telford and Masson (2005) observe that the emergence of the external quality assurance regime based on externally set standards in the UK, is a result of the increasingly competitive landscape and is quite different from the internal evaluation processes prevailing within universities until a few decades ago. In his study on the relevance of quality to institutional performance assessment, Pounder (1999) highlights the difficulty of producing a valid and reliable effectiveness scale for quality as a basis for the comparative assessment of institutional performance. His attempt to develop a unidimensional scale for quality led to a series of behavioural examples which reflected only the specific views of quality held by the providers of the examples, which according to him highlights the ambiguity inherent in the concept of quality. Pounder contends that the challenge for HE is to locate a concept or concepts more appropriate than quality as a basis for comparative

analysis of institutional performance and suggests that other more specific performance variables including information management, planning-goal setting, productivity-efficiency and cohesion, offer a firmer base for a comparative analysis of institutional performance. However, Pounder's differentiation between quality and the 'other' variables is unclear as the concept of managing quality can encompass all the variables he suggests. Moreover, the lack of a unidimensional scale for quality that facilitates comparative analysis of institutional performance should not preclude institutions from evaluating the quality of their provision, if only for their own longer-term success.

Although there is as yet no definitive model for managing or evaluating quality in HE, Becket and Brookes (2005) have identified certain key constituents of an analytical quality framework in order to assess current quality management practice. These constituents include:

- the degree to which inputs, processes and outputs in HE are assessed;
- the degree to which different stakeholder perspectives are considered;
- the extent to which different quality dimensions are considered;
- the balance of quality assurance versus quality enhancement practices and processes.

There are currently two aspects to managing quality in HE: quality assurance and quality enhancement (Lomas, 2004). Quality assurance deals with the monitoring, evaluative and conformance aspects of managing quality that is best aligned with the fitness for purpose definition of quality (West-Burnham and Davies, 1994). Quality assurance has many critics essentially because of its emphasis on detailed documentation and paper trails for audit purposes, and the hard managerialist approach that is required to ensure that these requirements are met (Hargreaves, 1998). It is also criticised on the ground that it fails to give sufficient emphasis to teaching and learning. Furthermore, the transformative role of HE is achieved only if there is sufficient focus on transformational change throughout the institution with the objective of improving quality and adding value, an approach which is categorised as quality enhancement (Middllehurst, 1997). Quality enhancement with it transformative character and longer-term outlook is considered a better approach to help embed a quality culture within institutions (Lomas, 2004). Williams (2002) contends that quality assurance and quality enhancement should not be considered as two distinct functions but as two integral parts to quality management. HEPs are increasingly required to

have quality assurance procedures that are both rigorous and transparent to external stakeholders while also continually enhancing the quality of provision by embedding quality enhancement initiatives (Becket and Brookes, 2005). The emphasis should be on quality enhancement with accountability being a consequence and not vice-versa or two separate aspects (Srikanthan and Dalrymple, 2007); by combining both quality assurance and enhancement the emphasis should on embedding a quality culture within the institution (Becket and Brookes, 2005).

An effective model of quality would be one that allows HEPs to enhance the quality of their provision, their interactions with and experiences of key stakeholders and achieve their longer term objectives. However, current quality control practices have been seen largely as a means of control on the part of outside funding bodies and place undue emphasis on documentation and evidence for external bodies and the resulting disruption to normal activity threatens the enthusiasm and commitment of staff (Colling and Harvey, 1996; Westerheijden, 2000). The ineffectiveness of the control approach is highlighted when one considers the fundamental principles of quality management, which relates service quality to processes and ultimately to morale and management (Srikanthan and Dalrymple, 2003). Educational institutions are typically complex organisations embedded within even more complex communities, where knowledge is fragmented into specialised areas and educators are engaged in a highly individual activity of teaching (Senge, 2000). Therefore, sufficient leverage is required to make changes in such a complex system. This requires staff and students (as co-participants) to consider quality management as a normal, integral and continuous function of all activities, rather than at particular times where undue emphasis is placed on bureaucracy, special documentation and other requirements.

## 2.4 FUNCTIONAL QUALITY VERSUS TECHNICAL QUALITY

The literature on service quality distinguishes between functional quality, i.e. the way in which the service has been delivered, and technical quality i.e. the outcome or the product of the service (Lewis, 1991). Hill (1995) describes technical quality in terms of the tangibles, knowledge, solutions, etc. provided during the service, while functional quality refers to how the service is provided including the interpersonal behaviours of the service staff during the process of the service. Customers may be unable to discern the technical quality of the service they receive due to their lack of expertise, and, therefore, are more apt

to judge service quality based on the way the service has been delivered (Mangold and Babakus, 1991). Anderson (1995) advocates that organisational resources should be directed towards improving functional or process quality thereby improving customer perceptions rather than the outcome or technical quality. Furthermore, measuring the characteristics or features of the process will provide leading indicators that will help identify areas for improvement, while measuring outcomes will provide at best lagging indicators which are too late to act upon (Tribus, 1994).

Based on the service quality literature, the focus of managing quality in HE can be directed towards three broad areas: the quality of inputs to the educational process, the quality of the process of education, and the quality of outputs from the process. Given the complexities, time period and the level and depth of interaction involved in HE, the process of education or functional quality could be considered the most critical aspect of quality management. Accordingly, Yorke (1999) argues that the main issues relating to educational quality are process-related, essentially whether the educational process serves as a satisfactory bridge between entrants and the programme's intended outcomes. This requires adequate demonstrable evidence as to whether it is the educational process rather than the inherent ability of the students which has contributed to achieving outcomes. Eriksen (1995) contends that if conformity to predetermined standards is important, the closer the output is to the standard, the higher is the operational or process quality. However, as Cuthbert (1996b) argues, the real outcome of HE is more than just certification which is what can be measured easily, while the deeper benefits may be obvious only some years afterwards and, therefore, cannot be easily measured. Trow (1996, p.52) rightly points out that

Education is a process pretending to be an outcome. That is what makes all measures of educational outcomes spurious. Our impact on our students can never be fully known; it emerges over their whole lifetimes and takes various forms at different points in their lives.

Therefore, there are difficulties in evaluating the quality of both the process and outcome in HE. The ongoing transformative nature of HE does not lend itself to a simple, discernible product or outcome, as the impact of the transformation may be felt years after the experience (Harvey, 1995). Moreover, the intangibility of the educational process and the lack of physical evidence present difficulties in analysing and evaluating perceptions of

process quality (Mahapatra and Khan, 2007) and student perceptions of process quality during the actual process itself may not be valid indicators.

The variables that comprise each of the input, process and output dimensions in HE can be developed as follows (Sahney et al., 2006):

- *Input*: student intake and characteristics, programme and curriculum, experience and qualifications of teaching staff and support staff, physical infrastructure and resources including library and teaching and learning facilities.
- Process: The teaching and learning methods and environment, design (class sizes, schedules) research activities, assessment and evaluation activities, extracurricular activities.
- Output: academic achievement in terms of marks/degrees awarded, graduation, dropouts, acquisition of transferable skills and employment.

A relevant issue in managing quality would be to identify the appropriate balance between the input, process and outcomes dimensions. Accreditation agencies in the United States have conducted workshops and forums on how to balance the emphasis on input, process and outcomes in accreditation decisions, but implementation has been inconsistent (Schray, 2006). There are further complexities involved because although the desired outcome should dictate the process to be followed, in reality the actual process that is followed determines the achieved outcome (Saad and Siha, 2000). Acceptable levels of quality will not be attained if courses are designed inefficiently or delivered inefficiently or assessed inefficiently (Eriksen, 1995). Students' input provides the raw material that is transformed through the process of HE and, therefore, the quality of the input provided will determine the quality of the process and the output. Students themselves may have important influences on each other (Owlia and Aspinwal, 1996). In a process intensive service such as HE, the degree of close personal interaction between students and between students and lecturers will make the management of functional or process quality even more problematic. Hence, although managing functional or process quality may be the most important aspect of ensuring technical or outcome quality, it may be difficult to assess and evaluate this process effectively and independently of the inputs and achieved outcomes. It is important to realise the interdependent character of input, process and output. Furthermore, the degree of importance that is attached by stakeholders to the input, process and output aspects of HE

may differ. For instance, when judging quality, students may tend to attach more importance to what was provided to them, while overlooking intangible aspects of the process such as the extent to which they were stimulated to think for themselves or to take responsibility for their own learning which are of keen interest to academics (Yorke, 1999). For employers, the attributes and qualities of graduates (output) may take precedence over the totality of the learning experience that is critical to academics and students. Hence, a first step would be to determine the input, process and output aspects that stakeholders value more as this would help to determine the appropriate balance between the three dimensions. These dimensions of HE are discussed further in the following sections.

### **2.4.1** Input

In HE students are considered the main input undergoing transformation, although the student plays three roles: that of input, consumer and processor (Williams, 1993). In any educational system the learner is at the centre and must be the major focus of the educational process (Muller and Funnell, 1992). According to Eriksen (1995, p.15) the primary input in HE is:

the student (before exposure to a value-added service) who is subjected to a transformation (the application of a value-added service) which in turn produces an output (the student after exposure to a value-added service).

Therefore, the quality of student intake is considered by many to be a necessary condition for institutional success (Cheng and Tam, 1997). Admissions criteria are believed to play a critical role in ensuring the quality of academic programmes as they ensure that students admitted onto programmes have a higher probability of succeeding in comparison to students who do not fulfill the admission criteria (Lawrence and Pharr, 2003). They argue that weak students contribute less to the learning of their peers and can cause lecturers to lower expectations. Pursglove and Simpson (2007) find a positive correlation between entry requirements of universities and the class of degree awarded, while there is a negative correlation between entry requirements and the degree of participation by students from non-traditional backgrounds (i.e. from generally poorer areas where it is unusual for school leavers to enter university). They find that entrants with lower achievements at entry level were more likely to drop out and if they did proceed onto graduate, achieved lower classes of degrees. Hence, admissions standards have a significant quality control role through its

impact on the quality of the student intake which in turn, may significantly influence the quality of the overall educational outcomes. As a result, the quality of student intake affects the institution's rating, thus fuelling a fierce competition for students with high marks (Srikanthan and Dalrymple, 2007).

Others, however, regard the quality of student intake as immaterial, as it is easier for HEPs to demonstrate better outcomes with higher quality intake. Instead the degree to which the institution has aided the enhancement of students' knowledge and skills or the value added, irrespective of the quality of the intake, should be considered the critical factor. This view considers that the quality of the educational process rather than the inherent ability of the students, should contribute to the excellence of outcomes by providing a satisfactory bridge between entrants to the programme and its intended outcomes (Yorke, 1999). Measuring the value added in terms of relating the entry and exit qualifications can be problematic (Yorke, 1997) and entails measuring the additional progress students have made over and above what they would 'normally' be expected to achieve given their prior attainment (Gorard, 2006). This evaluation is carried out in schools in the UK and is considered a potentially fairer way to measure the impact of educational provision as it takes into account the abilities of students when they arrive at the school by measuring progress made rather than simply raw outcomes (Welsh Assembly Government: Consultation Paper, 2007). Lawrence and Pharr (2003) argue that with the comprehensive changes that have been made to the traditional curriculum of most programmes over the last decade, the validity of admissions criteria has to be re-examined. Hence, irrespective of the quality of intake what would be more relevant to evaluating quality is whether the curriculum, teaching, assessment, support and guidance are the most appropriate with respect to the student profile in order to achieve defined objectives.

Other key inputs include the curriculum that greatly influences the process and outcome of HE. In fact, the curriculum can encourage or discourage the development of subject and practical knowledge, the development of core transferable skills, the choice of teaching and learning methods and assessment strategies (Nabi and Bagley, 1998). The increasing diversity in students' backgrounds and range of abilities in the classroom has brought pressure on academics to design, develop and deliver curriculum that is accessible and which acknowledges such diversity (Stefani, 2005). Thomas (2007) remarks on the impact that economic and demographic changes have had on business schools, including the need

for more flexible curricula, greater emphasis on lifelong learning and the changing characteristics and skills required from graduates. Over the last decade many business schools have radically redesigned their curricula, in particular by emphasising integration across functional areas and developing effective team work and communication skills (Lawrence and Pharr, 2003). Middlehurst (2001) contends that the content and curriculum must be fit for the purpose for which it is designed and must offer value for money, either for the direct customer (student) or the indirect purchaser (state or employer). A critical issue is whether curriculum developers are able to determine whether the curriculum they design is able to meet the ever changing needs of both students and potential employers (Anon, 2006).

In terms of discipline-specific knowledge, Baruch and Leeming (1996) find that in the early stages of a graduate's working career, more specialized or focused knowledge may be needed, whereas later, more generalization is necessary. Given the importance of curricula, there are arguments for standardising or offering part-common curriculum across disciplines, but according to Yorke (1999) this would discourage academic growth and creativity and may be detrimental in the long term. The importance of practical skills in understanding theoretical concepts is emphasised by Paloniemi (2006) who stresses that students must be aware of how to adapt their formal education in practical work situations. The challenge for HEPs lies in developing curricula that provides adequate scope for students to enhance complex interactive skills involving a critical mind-set. Research evidence clearly suggests that curriculum can influence tutors to focus on the subject matter rather than on the development of critical thinking (Kember, 1997). Developing students for employment requires a balance between professional and subject/discipline based knowledge (Bowden and Marton, 1998). Stefani (2005) argues that if a key role of HE is to support the development of critical thinking and reflection, then curriculum should be built around the needs and aspirations of the learner, where learners can learn to challenge and be challenged. The context in which the content is learned is vital and curriculum is instrumental in providing this balance between content and context. When designing curricula, sufficient opportunities must be provided to develop key skills by ensuring that students repeatedly come across similar advice and assessment in a range of courses. However, Dillon and Hodgkinson (2000) find that some students find this repetition irritating and demotivating and, therefore, recommend that curriculum designers could use courses as a context for skills development while seeing the overall programme as the context for skills assessment.

The fitness for purpose of learning resources and the extent to which they are effectively used by the HEP will greatly impact on the overall learning experience and as a result may influence perceptions and the outcomes achieved. The inputs available to the learner that form the overall support systems are also critical aspects of quality management (Middlehurst, 2001). He notes that technological advances have extended the range of media that can be used in the delivery and support of HE and this can alter the requirements that underpin qualifications. Student expectations of curriculum and other input criteria may also vary; for instance, because of their busy life style and needs, mature working students expect HEPs to offer a variety of courses, campus and delivery options that are easily accessible and user friendly (Wright and O'Neill, 2002). Chadwick (1995) observes that with the pressures of the economic environment, students look for more flexibility in course structures enabling them to opt in or out of modules and even institutions. The increasing popularity of online degree programmes indicates that a large proportion of students, particularly part-time working professionals, are giving more importance to flexible modes of delivery over traditional classroom style HE (Emiliani, 2005).

Without any doubt, teaching faculty is a key input in the HE process as they influence to a large extent the delivery of the curriculum and most importantly, the overall learning experience of students. However, as with other aspects of the HE process there is difficulty in defining the ideal profile of an effective lecturer. Although effectiveness in the classroom comes in many forms, Gilbert et al. (1993) writes that the most significant characteristics of effective lecturers include a rich understanding of the subject, ability to provide constructive feedback and realistic evaluation, insightful planning, organising, effective communication with students, strong interpersonal skills, and taking responsibility for the quality of aesthetics and tangibles. There is considerable support in the literature that the quality of the student experience is most greatly influenced by the expertise, teaching styles, commitment and enthusiasm of the teaching staff (Hill et al, 2003). Cook (1997) also finds that students consider academic staff factors as having the most impact on their success and among such factors the approachability of academic staff is a key factor. Therefore, in terms of inputs, the effectiveness of HE would depend on the appropriate blend of teaching and learning

resources and facilities, curriculum factors, as well academic staff factors including approachability, enthusiasm and teaching styles.

### 2.4.2 Process

Cheng et al. (1994) identifies three categories of services: *pure* services in which the customer must be present or is the most critical aspect of the service, *mixed* services in which there is both face-to-face as well as back-office contact with the customer and *quasi-manufacturing* services where there may be no real face-to-face contact with the customer. They categorise education as a pure service, considering the degree of interaction involved between students and the education provider and particularly because learning does not materialise without some participation by the student. In such contexts, customer satisfaction is most often determined by the quality of personal contacts. Therefore, much of the research on the service quality aspects of HE deals with the process aspects, for instance, on the effectiveness of course delivery mechanisms, the quality of courses and teaching and learning (Oldfield and Baron, 2000). The educational process is normally divided into two distinct areas: the administrative and academic support areas (library, enrolment, etc.) and the teaching and learning function (Srikanthan and Dalrymple, 2007).

A key determinant of the quality of the educational process is whether it is appropriate for the entrants to the programme to achieve the intended learning outcomes (Yorke, 1999). There is then the assumption that an excellent educational process would be followed by excellent outcomes; however this need not necessarily be the case. The debate on stronger intake leading to excellent outcomes versus an educational process which provides opportunities for a weaker intake to achieve at least threshold outcomes has to be considered. Is the quality of the educational process better in the first case because the outcomes are better? In the latter case, can efficiency of the educational process be accurately evaluated i.e. how much of value must be added and can achievement of threshold learning outcomes by weaker students be considered as indicative of good quality? Gorard (2006) notes that UK schools have been dealing with the concept of 'value-added' for many years and currently there are many measures in use. Fischer Family trust data, for example, predicts what grades students should get at different ages based upon their prior achievement and on the postcode of their homes and their schools. However, there is nothing comparable for HEPs.

Obviously, excellent teaching is not always followed by good learning and good student performance is not necessarily an indicator of good teaching; however, there is ample evidence that learning happens as a consequence of good teaching (Yorke, 1999). Researchers point out the need for rewarding and recognising good quality teaching in an environment where there is inadequate emphasis placed on the teaching function (Lomas, 2004). However, as Cuthbert (1996) points out, the relationship between student learning and teaching is neither simple nor direct, and the approach to learning adopted by the student has a significant influence on the teaching and learning process and outcomes. Other factors that must be considered include personal interaction between student and staff in the form of pastoral support and sensitivity to student needs (Hill, 1995). The complexity increases as service quality can be heterogeneous and the quality of interaction can also differ within the same institution.

While HE was traditionally more concerned with the transmission of knowledge, the explosion of information in today's knowledge economy requires tutors to develop a different skill set that involves seeking, analysing and evaluating information (Stefani, 2005). To enhance students' capacity for critical thinking, tutors must be able to offer clear guidance about what is required and provide feedback in order to improve the ability for critical reflection throughout the courses they teach (Harvey & Knight, 1996). Collaborative, transformative and critical learning that values and encourages diversity is crucial in developing critical skills (Hill et. al. 2003). In this respect, while quality of student intake may not necessarily be a major focus of quality management as discussed earlier, the quality of input provided by students during their learning experience is no doubt critical to the quality of the process and outcomes. Student learning depends to a very large extent on the student's own approach to learning, thereby rendering any simple assumptions about the relationship between teaching and learning meaningless (Cuthbert, 1996-a). Students can be considered as co-producers of their own learning (Hennig-Thrau et al., 2001) and, therefore, unless they participate to the fullest possible extent, the learning outcomes or objectives may not be met satisfactorily. The transformative aspect of HE wherein students' analytical and critical skills are developed will be realised only by the joint participation of students and academic staff in achieving the programme objectives (Harvey and Knight, 1996). The critical factors internal to the student that influence the transformative process of HE are motivation, maturity and talent or capability (Eriksen, 1995). However, undergraduate students, in particular, may not be prepared for a learning environment which involves self-directed participation and may consider themselves as merely inputs or raw material in the educational process to be transformed without putting in the effort required for that transformation to take place effectively. The difficulty for the lecturer is in convincing the student of the need to put in the necessary effort required in order to benefit effectively and completely from their experience (Hewitt and Clayton, 1999). Mattick and Knight's (2007) findings show that students find the process of self-directed learning and participation daunting, as they are uncertain about the appropriate amount of individual study, what to cover and how its success could be evaluated.

It follows that managing student participation, expectations, motivations, values and aspirations so that they are actively involved in and effectively contributing to the learning process are an extremely critical aspect of quality management. The service sector equivalent of this is called 'customer organisational socialization' whereby customers are supported in acquiring the service skills and knowledge required to interact effectively with the organisation's employees in order to achieve the planned outcomes (Telford and Masson, 2005, p.108). Most consumers will have a set of expectations when they enter a service encounter but while their expectations will be well-defined if they are familiar with the service, expectations may be ill-defined in unfamiliar service contexts (Hill, 1995). Other factors which can influence student expectations include word-of-mouth communications, students' personal needs and their past experience of education. Undergraduate students in particular do not have any comparative frame of reference with regards to expectations in HE other than that of their schooling system and hence unrealistic expectations may negatively influence their perceptions of quality (Hill, 1995). Therefore, while undergraduate students may expect close relationships with tutors, postgraduate students may base their expectations on their previous undergraduate experiences. There is also the possibility that student expectations and needs may change over the period of time that the student is in the institution. There is also some evidence that student perceptions and expectations of quality are influenced by their cultural orientation (Tan and Kek, 2004) but there is room for more investigation. In a study on UK university lecturers Rolfe (2002) finds that students tend to adopt passive learning approaches and expect all information to be provided to them as a result of their secondary school experiences and time and societal constraints. Comparatively, part-time or mature age students tend to view education just as any other commercial activity and will, therefore, have the same expectations from HEPs, such as convenience, high quality and low costs and service (Haworth and Conrad, 1997).

In one of the very few studies to be conducted on learning behaviours of students (of nursing) in the Arabian Gulf region, Bridger (2007), writes that students tend to be passive learners who exhibit poor learning tendencies including learning by rote, inability to integrate concepts and ideas within a subject area, not taking any initiative to resolve issues prior to asking for help, and inability to use prior learning. Similarly, Goodliffe (2004) writes that the majority of Omani students enter higher education with a background of teacher-centred instruction and rote learning. Therefore, these students may expect tutor-centred approaches from their schooling experience to continue on into HE, and attempts by HEPs to move to student-centred teaching and assessment methods may not be appreciated.

Student learning is influenced by three contextual factors namely, course organisation and resources, teaching and learning activities and assessment (Struyvern et al., 2002). How the student uses the context depends on four student-centred factors: students' selfmanagement, students' motivation and needs, students' understanding and students' need for support. Telford and Masson (2005) find that student values and expectations and thereby positive perceptions of quality are associated with their ability to contribute and participate, role clarity, positive perceptions of the organisational climate in which the service takes place and the extent of student satisfaction. This then brings us to the issue of how HEPs can manage student perceptions and expectations over time, which is a fundamental requirement of managing service quality as per the literature. Student expectations over the lengthy process of HE will definitely impact on their motivations and behaviours and ignoring such expectations and needs may not only negatively impact on students' perceptions of quality but may also detract them from contributing positively to the learning process. Managing such expectations may entail managing superficialities that may not actually have a major bearing on the outcome of HE. Hill (1995) advocates using existing students on school visits and open days in order to shape the expectations of prospective undergraduates and make them as realistic as possible. Students should also be encouraged to reflect on their past learning experiences so that they can build on positive approaches and unlearn negative ones (Hill et al, 1996). Nevertheless, it must be noted that the literature on this very important aspect of managing educational quality is limited.

Educationalists also advocate a deep approach to learning which is fostered by appropriate teaching and assessment methods that promote active engagement with tasks and provide opportunities for independent studying. Once again, the ability of the lecturer to

demonstrate his/her personal commitment and enthusiasm for the subject and thereby stimulate students' interest and commitment is vital in shaping deep approaches to learning (Pennington and O'Neil, 1994). Good teaching integrates three aspects of competence: practice, disciplinary knowledge and generic skills through structured, explicit goals, curriculum balance between content and understanding, and a range of assessment methods (Srikanthan and Dalrymple, 2007). In order to achieve this, the academic role should be student centred and will include lecturing, facilitating, empathising, coaching and designing customised learning experiences which stimulate learning. Unfortunately, the intense pressure to achieve qualifications for a competitive job market can result in students trying to avoid the more complex demands of autonomous or critical learning; hence, they are tempted to focus on learning for assessments or surface rather than deep learning (Chadwick, 1995).

Additional factors that add to the complexity of the educational process include the differing perspectives of staff and the need for synergistic involvement and collaboration among the programme team. Narasimhan (1997) highlights that the complex process of teaching and learning has been complicated further by a number of factors including:

- increase in class sizes due to expansion in student numbers;
- mixed ability students in the same classes due to the diversity of student intake;
- increasing demands of students who are better informed and have thus become more aware of their rights;
- pressure from regulatory authorities;
- emphasis on innovation in teaching and learning and student-centred approaches.

The integration between research and teaching, the two core activities of universities also has many critics. While many studies indicate that research can enhance the relevancy and currency of the researcher's teaching (Andresen, 2000) others such as Rowland et al. (1998) reject the premise that there is an automatic synergistic relationship between teaching and research. The pro-research groups point out that research can create learning communities where there is a collaborative relationship between lecturers and students (Lomas, 2004). While teaching has always been important in HE, research has generally been held in higher regard (Stefani, 2005). Johnson and Deem (2003) note that in the UK, the emphasis on research activity and income through the Research Assessment Exercise forces academics to

focus on research rather than teaching. They observe that while creativity in research is rewarded, excellence and innovation in teaching are largely ignored. Elton (1998) argues that the emphasis on research at the expense of teaching is not healthy, as in order to improve quality HEPs must concentrate more on improving their core activity of teaching. Fortunately, the balance between the privileged status of discipline-based research and facilitation of student learning is now being redressed by an increasing pressure on academics to provide effective teaching and learning environments, leading to a rediscovery of the scholarship of teaching (Stefani, 2005). Deem and Brehony (2005) also note that a feature of 'new managerialism' in UK universities is the increasing pressure on academic staff to carry out both research and teaching to a high standard. In Oman, most HEPs focus on teaching rather than research particularly in the early years of their establishment. Recently, however, there has been increased emphasis by the MoHE on research activities with the establishment of the National Research Council and all HEPs are now required to demonstrate evidence of research and scholarly activity.

The re-emphasis on the teaching function can also be attributed in part to the problems associated with the massification of HE and the corresponding implications in terms of resources and student demographics. The focus on efficiency and effectiveness together with the decline of economic resources and student numbers force academics to be innovative in teaching, learning and recruitment strategies, often without the core skills required to do so. There is the danger that a superficial understanding of popular definitions of the scholarship of teaching will pass for a real understanding of what is involved at different levels of teaching and assessment (Stefani, 2005). She recommends taking a further proactive step where all staff engaged in supporting student learning reflect on and interrogate the terms of reference for the scholarship of teaching by applying the principles to their own classroom teaching. Srikanthan and Dalrymple (2007) argue that universities should move from the rituals of teaching to commitment to learning, and this requires a fundamental paradigm shift in the notion of teaching as a routine and subsidiary task to a key performance indicator. This as Farrugia (1996) recommends requires the continuous professional development of teaching staff leading to the acquisition of key professional traits that help staff to adapt to changes such that their prestige and material rewards depend more on the quality of teaching and guidance they provide rather than solely on research.

The professional development of staff in terms of appropriate curriculum development, innovative teaching methods, student motivation and the development of core skills is critical if academic staff are to effectively fulfil their role in an increasingly demanding workplace. Obviously, the overall quality of the process in such an inter-personal and interactive environment such as HE will depend, as Lomas (2004) puts it, on the level of commitment throughout the organisation to quality and its continuous improvement. Transformative leadership which stimulates staff to question and alter basic assumptions and behaviours while encouraging innovation and change can help develop an organisational culture that can sustain the quest for constant improvements (Fullan, 2001). Furthermore, the management's commitment to quality must be visible, permanent and present at all levels and must translate into clear values that are disseminated throughout the institution (Calvo-Mora et al. 2006). Hence, leadership and management are very vital aspects of providing the right direction and initiative for establishing and sustaining the required commitment within the institution.

### **2.4.3** Output

Notwithstanding the argument for focusing on functional or process quality, there has been a shift in the focus of quality management in HE from process to outcome, the fundamental concern being that quality should be demonstrable. Emery et al. (2001) argue for an output oriented approach where the focus is on developing students' capabilities, knowledge and skills in accordance with the expectations of the industry. The most important evidence of quality in HE is considered to be performance, especially student achievement of learning outcomes that is represented by a broad range of performance indicators including access, productivity and efficiency, student learning, degree completion, and economic returns from HE (Schray, 2006). HE, particularly in the United Kingdom has been involved in an outcome based approach to quality whereby success is measured and evaluated in terms of the extent to which the predefined learning outcomes have been achieved (Lomas, 2004). The touchstone of quality for the Higher Education Funding Council of England is the output of the system in terms of what graduates can do at the end of the progamme (Hewitt and Clayton, 1999). In the United States as well, there has been a fundamental shift in the focus of quality assurance and accreditation agencies towards student learning outcomes (Welsh and Dey, 2002). Schray (2006), however, notes that current accreditation procedures

have a long way to go in assuring standards in student learning outcomes, as attention has always been focused on resources and process standards.

The emphasis on outcomes or output essentially arises from the necessity of having an objective measure for assessing the quality of provision (i.e. the inputs and the process). Williams (1997) contends that it is also closely aligned with increased levels of managerialism resulting from demands for more efficiency and accountability, as outcomes can be more easily measured against established targets or standards. If student perception of the process of education is not the best reliable indicator of quality, then arguably the outcome of that education would be a more appropriate and objective measure of the process. In this respect, the role of learning assessment is to measure the value added by the instructional process towards developing the specific learning competencies that are identified by the institution (Amin and Amin, 2003).

However, there are doubts whether the complex nature of learning can be easily reduced to a set of competencies and learning outcomes and for some, attempts to do so imply an unwelcome shift towards utilitarianism (Barnett, 1994). Considering the complex multidimensional nature of HE, Hewitt and Clayton (1999) question the notion that the output of HE is a more appropriate measure of quality than the process. They find that academic staff consider the quality and integrity of the learning environment (process) which requires students to be proactive partners in the overall experience rather than the outcome to be the most critical factor. In industry, the quality of output is measured by the extent to which it conforms to some predetermined standard: the closer the output is to the predetermined standard, the higher is the operationally defined quality (Eriksen, 1995). In HE, the composition of the desired portfolio of outcomes, attributes, skills and qualities is debatable and the question of which outcomes are more important than others must be considered. Furthermore, some outputs or outcomes are more easily measured than others (Lomas, 2004). The number of first class degrees or distinctions would be the easiest to measure but a high proportion of 'firsts' need not necessarily indicate a high level of quality and there can be justified variation according to the subject (Yorke, 1997). The influence and outcome of HE should (if it is effective) continue long after the formal programme of study has been completed and hence, it is doubtful whether HE can be reduced to a simple, measurable end-product.

Graduates' success in gaining employment or access to higher qualifications/ research is an important measure of the quality of educational provision. HE has a key role to play in developing skills for lifelong learning so that graduates can be more effective in their work place, can process knowledge effectively as well as apply skills and knowledge in different contexts (Yorke, 1999). The most critical function of HE is considered to be the development of skills that enable learners to learn or find out for themselves rather than imparting subject knowledge (Bourner, 1998). For this to happen, the focus of curriculum and teaching and learning must be on helping students to learn how to find out for themselves and assessments in turn should be more about testing the ability of the student to find out. There is no doubt that preparing graduates for entry to the world of work requires much more than the passive reproduction of subject knowledge (Eagle and Brennan, 2007).

In a labour market which is marked by uncertainties, rapid change and competitiveness and where companies are downsizing and delayering in order to remain competitive (Jameson and Holden, 2000), graduates' preparedness to adapt to the world of work may be their key distinguishing factor. Harvey and Green (1994) report five broad areas of graduate attributes which are important to employers including knowledge, intellectual ability, the ability to work in modern organisations, interpersonal skills and communication. While subject specific knowledge is one aspect, preparing students for employment is not as straightforward as one may consider and a key difficulty is identifying criteria relating to transferable skills that are particularly relevant for employment (Yorke, 1999). Employer satisfaction surveys across various fields give low ratings for academic subject knowledge while transferable skills e.g. problem solving, team interaction, communication etc., are very highly prized as these are skills learnt in one domain which can be transferred into other domains (Srikanthan and Dalrymple, 2004). Such attributes or qualities are generic in nature rather than specific, a fact which underscores the changing nature of organisations. However, Morley and Aynsley (2007) do find that employers' perception about what constituted quality in HE differed across sectors with scientific and technical employers placing greater emphasis on graduates' subject knowledge.

Employers observe that technical and organisational changes have resulted in increased expectations from new recruits, making qualities such as creativity, willingness to learn, team skills and group leadership abilities more important. A survey on global business capabilities identified the skills of flexibility, cultural sensitivity and integrity as the most

preferred list of managerial attributes from a list of 40 capabilities (Thomas, 2007). He finds that recruiters of business school graduates look for candidates who possess complex interactive skills and higher level evaluative abilities including the ability to link things together and frame complex problems. Employers of business graduates, alumni and even students are also increasingly demanding behavioral and societal skills as well as critical skills (Hawawini, 2005). Behavioral skills include entrepreneurial and leadership qualities, the ability to work with others, to communicate effectively and to demonstrate multicultural awareness. Societal values include the ability to make ethical decisions which take into account corporate social responsibility. However, developing behavioural and social values in their students may not be possible for HEPs as students may already have developed certain behavioural values by the time they are of college going age. Peronne & Vickers (2003) emphasise the impact of the transition that graduates have to make from tertiary education to the workplace in the current competitive economic climate. They note that this transition most often produces feelings of anxiety, stress, depression and of low self worth, which are not widely appreciated by employers and HEPs nor are they anticipated by students themselves. In order to better prepare graduates to adapt to the workplace HEPs and academics would need to put in greater effort to enhance student awareness of the importance of core transferable skills and personal and behavioural development.

Research on student learning indicates that assessment is integral to student learning and that it essentially drives the curriculum (Stefani, 2005). Assessment of student learning is considered to be the most significant factor affecting transformation (Srikanthan and Dalrymple, 2007). However, it is often found that tutors encourage a surface approach to learning by assessing those aspects which require memorisation of facts rather than focusing on how students use, evaluate and interpret information (Struyven et al. 2002). Stefani (2005) warns that assessment is often considered as a separate entity from teaching and learning and is usually considered after the course content has been decided on. She recommends an integrated view of the scholarship of teaching, learning and assessment where it is recognised that all three are complementary and directly related.

Feedback on assessments which provides students with closure and constructive ideas for improvement is also an integral part of an effective assessment strategy. Assessment methods which are perceived to be inappropriate by students also render the learning experience less satisfactory (Struyven et al., 2002). In their study on assessments in

relation to industry needs, McHardy and Allen (2000) find that although students understand the relevance of non-traditional innovative methods of assessment (e.g. roleplays) they are not comfortable with such methods and prefer prescriptive methods of learning and assessment. Gibbs and Simpson (2005) observe that quality assurance agencies and HEPs focus on assessments in terms of what they measure rather than in terms of how they support worthwhile learning. They contend that standards are improved when assessments improve student learning rather than simply measuring limited learning. They also note that the quality of student learning has been shown to be higher in assignmentbased courses than exam-based courses. Struyven et al. (2002) find that multiple-choice formats or questions that emphasise detailed factual answers encourage a surface approach, while open, essay- type questions encourage a deep approach to learning. They find that students with good learning skills and low test anxiety rates favour essay type exams, while students with poor learning skills and low test anxiety favour the multiple choice format. Most importantly, they emphasise that students seek information and form opinions about what is expected by the teacher and then tailor study strategies that fit the task. Mattick and Knight (2007) find that anxiety associated with poor performance can drive students to focus on assessments in a way that is incompatible with high-quality learning and without reconsidering their learning approaches. Hence, if the assessment strategy is not well designed or comprehensive it may become comparatively easier for students to provide the teacher with whatever s/he expects without really going through a transformative process or enhancing their capabilities.

### 2.5 STUDIES ON QUALITY VALUES

Most studies on quality in HE seek to determine the extent of student satisfaction with the quality of teaching and learning, facilities and other related aspects of the HE experience (Joseph and Joseph, 1997) while Hill et al. (2003) note that there is very little empirical research into student perceptions of quality in HE. In line with studies measuring service quality in other sectors, the majority of the more detailed quantitative studies on HE quality seek to explore the differences between students' pre-purchase expectations and their perceptions of actual service performance (Wright and O'Neill, 2002). Such models like the SERVQUAL are based on the 'disconfirmation paradigm' (O'Neill, 2003, p.310) as they seek to explore the relationship between students' pre and post-service experience, based on the premise that satisfaction results will indicate how well the actual service performance

matches expectations. The five service dimensions that are included in the SERVQUAL include tangibles, reliability, responsiveness, assurance and empathy (Parasuraman et al., 1988). An important criticism of the SERVQUAL is that customer expectations may not actually exist or be clear enough in the respondents' minds to serve as effective benchmarks against which their perception of the actual quality of the service can be measured (Iacobucci et al., 1994). A modification of the SERVQUAL is the SERVPERF which measures only the actual perception of customers about service quality and does not compare it to pre service expectations.

The use of these disconfirmation models in HE has been criticised, particularly as they have been constructed for defining customer values and expectations in the general service sector and, hence, overlook a number of key areas specific to HE (Telfrod and Masson, 2005). Subsequently the HEdPERF was developed which is a more comprehensive, performance based scale of quality constructs within HE (Abdullah, 2006). However, although all the service quality measurement models are supposed to be very comprehensive, there is very little evidence that the service elements that are evaluated by these models including HEdPERF, measure what students really consider as relevant. Accordingly, Wright and O'Neill (2002) recommend that HEPs undertake extensive research in order to identify those factors deemed most important by students in their evaluations of service experiences, so that they can then be used to target specific improvements. Cuthbert (1996a) observes that numerous studies have also been conducted on student learning using well-validated instruments such as the classroom environment scale (CES), the individualized classroom environment questionnaire (ICEQ) and the student experiences questionnaire (CSEQ). However, after evaluating these instruments, he concludes that there is considerable diversity in the range of constructs used and that none of these instruments would provide an appropriate tool for evaluation as part of a quality assurance system. Nevertheless, such studies can aid HEPs in monitoring student perceptions of the teaching and learning process, which is an important step in managing their expectations and needs.

Hewitt and Clayton (1999) investigated the quality perceptions of students and staff on the BSc Optometry programme in Aston University, UK, using semi-structured interviews. Their findings reveal that while students share the staff's view of the paramount importance of learning resources, they are less inclined to accept responsibility for making proper use of these resources. The emphasis for students is on being taught rather than learning pro-

actively and accordingly six of the students' top ten requirements relate to direct support from staff. Academic staff consider the process and the quality and integrity of the learning environment rather than the outcomes as being most important. Hewitt and Clayton (1999) nevertheless suggests that the differences are reconcilable so as to support a quality model that focuses on education as an interactive process and allows multiplicity of purpose with the possibility of individual actors simultaneously playing multiple roles.

Using focus groups Hill et al. (2003) examined perceptions of quality in HE among students on nursing, management and education programmes. They found that the majority of the students' comments related to the 'quality of the lecturer' and 'student engagement with learning' as influenced by the expertise of lecturers who could facilitate debate and discussion and were supportive. Other areas that were identified included among others the student support unit and the need for shared experiences with other motivated students. They note that students felt the need for a positive atmosphere that go beyond a well-structured lesson and pleasant learning environment where the quality of the interaction between student/student and student/faculty determines the quality of the learning experience

Lagrosen et al. (2004) examined the dimensions that constitute quality in HE, from the student perspective and compared these with the dimensions of quality that have been developed in general service quality research. They developed a 32 statement questionnaire after carrying out 29 in-depth interviews with business students from Austria, Sweden and the UK, following which the questionnaire was delivered to students at two universities in Austria and Sweden. They find the interpretation of quality as excellence best matches student's view of quality, while specific quality dimensions include among others, library resources, information and responsiveness, corporate collaboration, courses offered, teaching practices and campus facilities. They find reasonable correspondence with the general service dimensions, but also find several differences as general service elements such as access, courtesy, security, attitudes and behaviour and service recovery not considered relevant by students. They conclude that as a single stakeholder perspective provides only a limited view, such studies must be complemented with other perspectives.

Chua (2004) conducted a survey in order to investigate how quality is perceived by different stakeholder groups including students, parents, faculty members and employers. Her

categorisation of quality characteristics was based on the IPO framework i.e. Input, Process, and Output criteria. 'Input' criteria included in the study were entry requirements and students selection, 'Process' considered the overall teaching and learning process and 'Output' included employability and academic standing; however an adequate rationale for the inclusion of only these particular criteria under the IPO framework was not provided. The questionnaire was based on the SERVQUAL dimensions of Parasuraman et al. (1988), therefore, rendering the study more of an extension of other studies that investigates student perceptions of quality. The main difference offered by Chua's study is that it is one of the few studies that explore quality attributes of HE from various stakeholder perspectives. Her findings indicate that both students and employers perceive the process and output to be the most important categories relating to quality, while the faculty's perspective of quality is wider in view than the others, and indicates that the focus should be on all aspects of their activities (i.e. input, process and output). Chua emphasises that her findings support the view that different groups of customers have different perspectives of quality and, therefore, recommends an integrated quality model that addresses these different perspectives.

Lomas (2004) investigated the views of a sample of senior managers and academics on the most influential factors in effectively embedding quality in an HEP. Using semi-structured in-depth interviews, he finds that respondents' consider the need for a quality culture, training for newly appointed lecturers and continuing professional development and peer review as the most important. Respondents also highlighted the importance of transformational leadership in implementing effective change management strategies.

Morley and Aynsley (2007) explored employers' needs for information on HE quality and standards in the UK. Their interviews revealed that employer recruitment practices seem to reinforce the notion of a graduate elite that could undermine equity and widening participation initiatives. While employers reported in their initial interviews, that they placed least emphasis on information about quality and standards and most emphasis on graduates' interpersonal and communication skills, the majority cited the importance of institutional reputation and the league tables in decision making. The factors that influenced institutional reputation included real or imagined league tables, personal, regional and professional networks, performance of past graduates and prejudice against new universities. They also find that employers equated high standards with rigorous entry criteria.

The one study which focuses on whether or not academics and students share the same quality values was conducted by Telford and Masson (2005), who note that it is very important to understand the quality values of students and academics as they have an impact on various aspects of student participation in the educational process including clarity of the students' role in educational delivery, their motivation and their ability to contribute effectively to the educational experience. Considering the specific nature, duration, level of interaction and complexities involved in HE, the values and expectations of academic staff will undoubtedly play a key role in moulding student participation and motivation. On this premise, Telford and Masson's (2005) investigated the effect of congruence in views on quality values between students and staff on student satisfaction on a single learning programme in Napier University. They used focus groups in the initial stage to generate data for the construction of the survey questionnaire. They found that although lack of congruence in views between staff and students does not necessarily lead to student dissatisfaction, a shared understanding of values is important in order to manage the quality of provision. It was also noted that the majority of issues on which staff and students shared the same values and yet students were dissatisfied, were those associated with lack of resources or factors external to the actual learning experience e.g. how the value of the University's degree awards are perceived by the wider community. The key values of students and academics were found to be those "associated with what the courses are designed to achieve, the manner in which they are delivered and supported, and the commitment required of the different participants" (p.115). Students were found to be primarily interested in vocational courses that will help their career and they considered the commitment of academic staff more important than their actual experience in the classroom. In contrast, academic staff rated commitment of staff as most important, followed by commitment of students as well as the vocational impact of the course. However as the researchers note, these results are not surprising as this study was confined to Napier University which is vocationally oriented and was focused on the specific discipline of optometry which is more technically oriented. Hence, the applicability of the findings is generally limited to that particular programme. Interestingly students did not consider their own commitment to learning as important even though, it is fundamental to the learning process as discussed earlier.

The above studies provide some evidence that different stakeholders may understand the concept of quality with regards to HE in different ways. The literature also lends support to

the premise that any model for managing quality would be ineffective unless it is based on an understanding of how to address the expectations of key stakeholders. However, in order to meet the needs of key groups HEPs may have to focus on different aspects of the education system, an approach which may not really be effective unless there is more understanding of how these expectations and quality values differ.

# 2.6 CONCLUSION

The literature establishes the inherent difficulty in describing and evaluating quality dimensions in HE despite the growing demands for greater accountability and increasing There is clearly a lack of conceptual models of quality stakeholder expectations. management that can be applied effectively to HE. There is also evidence that external assessments of quality through accreditation and quality audits are considered to be far from satisfactory and are seen as bureaucratic and deflecting important time and resources away from the core activities of HEPs. Such approaches will not be effective in contributing to developing and embedding a sustainable culture of quality enhancement since this requires empowerment and mutual recognition, not external control. Furthermore, most studies on quality in HE focus only on students' expectations and levels of satisfaction with a particular institution. While student expectations as direct customers and co-participants are vital, it is surprising that the views of academics as a main participant in the process of HE and that of employers as a key external stakeholder are not sufficiently explored in the literature. It must also be noted that virtually no studies on quality in HE have been conducted in the Middle-Eastern or Arab countries and, therefore, the applicability of the available literature to HEPs in the region may be limited.

While quality in HE is a multifaceted concept with a host of different variables that underpin the input, process and output dimensions, there are no models which seek to manage differences in perceptions among stakeholders regarding the key variables. If HEPs are to manage the quality of their provision effectively in the long term, it is important that a quality culture is embedded within the institution and quality is accepted as a normal academic function. Such an approach to quality is possible only if the key values and expectations of those who are intensively involved in the HE process (i.e. academic staff and students) are identified and addressed. As an external point of reference, employers' quality values and expectations of HE are vital in providing external validity to the outcome

of HE. Although such values and expectations may be in conflict with each other, it is argued that the success of any model for quality management will rest fundamentally on how it acknowledges and manages such differences and builds on the similarities.

#### **CHAPTER THREE**

# RESEARCH METHODOLOGY

This chapter discusses the research methods that address the specific research questions that have been raised. The methodological issues regarding the rigour of this study are also examined.

.

### 3.1 RESEARCH QUESTIONS

The primary objective of this study is to develop a framework for managing quality in HE based on the key quality values of three of the main stakeholder groups. The importance of developing a shared vision, commitment and collective understanding of key participants for the successful implementation of a quality culture is stressed in the literature and, hence, any model for managing quality can only succeed if it addresses the values of key stakeholders. By taking into consideration the similarities and differences in the quality values and expectations of the stakeholder groups, the proposed framework of quality management can ensure that similarities are used to maximum benefit while differences are addressed and managed effectively.

This study focused on three groups of stakeholders in HE whose views are considered fundamental to the objectives of HE and include:

- Internal stakeholders primarily responsible for the quality and delivery of programmes, namely academic staff.
- External stakeholders: employers of graduates from HE.
- Students as co-participants, co-producers and customers of HE.

Based on the research objectives, the following research questions were formulated:

- 1. Which criteria are viewed as specifically relevant to quality in HE by academics, students and employers in Oman?
- 2. What differences or congruencies exist in the quality values and expectations of the three stakeholder groups and what are the implications for HEPs?

3. What framework for managing quality can be proposed based on the stakeholder determinants of quality?

In addition, differences in quality values between academics on the basis of the subject discipline i.e. Business Studies and IT, as well as between students were explored.

#### 3.2 RESEARCH DESIGN

The research design included three phases:

- 1. A review of the literature on the key quality models, criteria, expectations and values in HE:
- 2. A series of interviews with a sample of all three key stakeholder groups that provided the data on stakeholder expectations and quality values to be included in the next stage;
- 3. A survey involving the three stakeholders using a questionnaire developed on the basis of the criteria identified from the first two phases. The survey sought to determine the differences and congruencies in quality values among the wider population.

Most studies on quality in HE are essentially quantitative in nature, aimed at measuring levels of satisfaction and service quality. The majority of studies on service quality including those on HE, are carried out through surveys using questionnaires aimed at measuring customer satisfaction, levels of service quality, quality dimensions, etc. (Madu, 1998). Studies on various other aspects in HE frequently use quantitative surveys or a combination of interviews, focus groups and quantitative surveys. Many studies employ the above methods within a case-study approach focused on particular institutions, for example Chua (2004), Hewitt and Clayton (1999) and Telford and Masson (2005).

This study uses a two-stage research design utilising both quantitative and qualitative methods which is considered ideal for exploring the complexities and contrasting values of the different stakeholders' groups on a multi-dimensional issue such as the management of quality. The ontological underpinning is based on the belief that the three stakeholder groups may not share common perspectives on quality in HE and that in all probability even members of the same group may have differing priorities and expectations. It was therefore concluded that the study should use the interpretative paradigm and qualitative methods in

the first instance to gain an in-depth and comprehensive understanding of stakeholder perspectives. In the second stage, this understanding was further validated by exploring the extent to which similar values are shared by the larger population. This was done through a quantitative survey using a closed, structured questionnaire, which is normally associated with the positivist paradigm.

A multi-faceted methodology that combines the use of both quantitative and qualitative methods can greatly enhance the quality of knowledge created, allowing corroboration through triangulation and richer and elaborate analysis. When findings from different sources or via different methods can be corroborated, confidence is enhanced; even if the findings are in conflict, the enhanced knowledge enables the researcher to interpret the results accordingly (Johnson & Onwuegbuzie, 2004). By combining the most appropriate methods, inferences can then be drawn and conclusions reached which otherwise may not have been possible. Reichardt & Rallis (1994) terms such an approach as the pragmatist position, which calls for using the most appropriate methodological approach for the particular research problem. The researcher, however, must be very clear about how the combined methods interact with each other with reference to the phenomenon under study, as he/she may have to consider possibly contradictory results (LeBlanc, 1995). The disadvantages of such a combination include mainly practical constraints such as time and the much wider scope of the study (Yin, 1994). Saunders et al. (2003) observe that research is qualified as inductive when the researcher develops theory from the data collected and Hyde (2000) describes inductive reasoning as a theory building process which starts with observations of specific instances thereby seeking to establish generalisations about the phenomenon under investigation. Therefore, this study is inductive as the qualitative methods informed the quantitative methods, and on the basis of the details obtained from both, the researcher developed the framework for managing quality.

The study sought to first identify key quality criteria as they emerged from the study and to explore the reasons for their importance through interviews with the three groups of stakeholders. Using inductive logic, the categories, variables and explanations that help explain the different stakeholders' views of quality in HE emerge from the study rather than being identified by the researcher beforehand. Any differences and similarities in views among a larger population were explored through a quantitative survey. The findings from

both the qualitative and quantitative data then formed the basis for a framework for managing quality.

#### 3.3 RESEARCH METHODS

## 3.3.1 Qualitative data: Semi-Structured interviews

An important step prior to the construction of a research instrument is to establish the theoretical foundation or the conceptual framework so that all concepts, attributes etc., that may be part of the subject under study are identified. Constructs, variables and expected relationships are identified based on the theory which guides the further investigation of such relationships. Careful construction of theory, therefore, becomes an essential first step in the actual design of an instrument and ensures an acceptable level of content validity. This can be carried out in various ways depending on the paradigmatic underpinning adopted by the researcher.

An extensive literature review, according to Madu (1998), can ensure that all the major issues/variables are included in a study. Positivist researchers claim the advantages of objectivity and neutrality in values by systematically and objectively reviewing the theoretical base and detaching themselves from the subjects under study (De Rada, 2005). However, qualitative researchers argue that in reality this detachment can ultimately result in researcher bias as the researchers themselves determine the direction of the enquiry, thus greatly influencing the questions/issues included in the instrument (Amartunga et al, 2002). While positivists claim to be detached and value-neutral and hence objective in their approach, the researchers and not the subjects under study determine the content or questions to be asked in the interview or on the survey. Instead, qualitative data with its potential for revealing complexity is often stated to be the best strategy for discovering and exploring new areas and for developing hypotheses (Amartunga et al, 2002). There is thus a strong rationale for combining the theoretical review with the use of qualitative methods so that the research subjects can provide the researcher with the relevant dimensions to be further explored by the study.

The review of literature helped identify the various dimensions and criteria of service quality considered critical to quality management in HE. However, as highlighted in earlier

chapters there is limited research evidence on whether the generally used criteria are indeed important to all stakeholders. Hence, although the literature review provided the fundamental theoretical background, the variables identified were further validated by indepth semi-structured interviews with a sample from each of the three stakeholder groups.

The intention at this stage was to elicit a range of issues that the members of the three stakeholder groups felt were especially relevant to quality in HE, and the contexts and reasons why such criteria become particularly influential. The interviews consisted of openended conversational type questions designed to allow sufficient opportunity for respondents to provide the researcher with their own views, thereby ensuring that they were not influenced by a predetermined list of factors. The descriptive analysis thus generated informed the survey which followed, and reduced the degree of researcher's bias when identifying relevant criteria, which is a major limitation of many previous studies on quality management. In doing so, the criticism that large sample studies using pre-structured questionnaires only include the researchers' own decisions as to what is important and do not allow respondents to voice their opinions on what is to be included, was overcome.

Focus groups were not considered appropriate for this study, as it was felt that Omani students particularly given their social and cultural upbringing may feel inhibited in a larger group setting and may not accurately report what is important to them. Morgan (1997) defines a focus group as a research technique that collects data through group interaction on a topic determined by the researcher. As Stokes and Bergin (2006) observe, the process of group dynamics which is responsible for many of the advantages of focus groups can be a double-edged sword, as it may inhibit participants in a group situation. For instance, social pressures may cause over-claiming or can influence some participants to publicly agree with others' views while privately disagreeing. They find that such group pressures can obscure the identification of a range of beliefs, attitudes and motivations thus producing a consensus view which lacks validity.

In comparison, individual depth interviews although semi-structured to an extent, allowed the researcher sufficient opportunity to get respondents from the three stakeholder groups to express their feelings on what is important to them in HE, more freely and in detail and to explore why this was so. Stokes and Bergin (2006) also find that focus groups were unable to match the depth and detail relating to issues that individual interviews were able to

provide. However, individual depth interviews have been criticised as being more difficult to interpret because of the sequential nature of data gathering which may cover up any consensus view (Greenbaum, 1998). Since the objective at this stage of the study was not to obtain a consensus view, this was not considered a limitation.

## 3.3.2 Qualitative Data: Sample

A total of 35 participants from the three stakeholder groups were interviewed from the HEPs and largest employers located in Muscat. The sample included 15 students, 10 academic staff and 10 employers and was considered sufficient to generate the descriptive analysis required to identify relevant quality criteria for each group. Strauss and Corbin (1990) suggest that decisions regarding the number of interviews and observations depend upon access, resources, research objectives and the available time and original decisions regarding sample size may also be modified as the theory evolves. The quality and range of information gathered, rather than the number of interviews was considered more important, and after 35 in-depth interviews it was felt that enough data which gave an adequately extensive reflection of the views of the three groups had been collected. The sample included students and lecturers as well as 3 Heads of Faculty/department so as to include the views of academics with leadership responsibilities. The 10 employers selected were from among the larger organisations in terms of staff, out of which, 6 were from private companies and 4 from government organisations and included senior managers, human resource and training managers. Reflecting the existing trends, the student sample included 13 Omanis and 2 expatriates, 9 academic staff were expatriates and 1 was Omani, while 6 employers were Omani and 4 were expatriates.

The focus of the study was further narrowed down to the most common undergraduate disciplines available in Oman which are Business Studies and Information Technology. Business studies programmes included the various business subject disciplines such as accounting, marketing, human resource, finance etc while IT programmes included all computing related undergraduate programmes. 14 of the interviews with students and academic staff were conducted during the three-day Gulf Higher Education Exhibition held in Muscat from the 16<sup>th</sup> to 19<sup>th</sup> of April, 2007. HE exhibitions in Oman are represented by institutions from all over Oman as well as international institutions and attract a large number of people. The exhibition provided a valuable opportunity to interview a wider

sample of staff and students than would otherwise have been possible. Second or final year students and academic staff from the two subject disciplines were selected from the stalls of the different HEPs in Oman that offer these undergraduate programmes. After being given an overview of the research and the purpose of the interview, the respondents were asked whether they would like to participate in the interviews. Care was taken to ensure that the interviews were conducted during quieter periods of the exhibition and ample time was given to each interview. The remaining interviews with students and staff were conducted later at 3 private sector HE institutions in Muscat. The interviews with employers were conducted at their offices after prior appointments were made.

# 3.3.3 Qualitative Data: Data Collection and Analysis

All interviews with the respondents were conducted in English, which was not an issue as they were either teaching, studying or working in environments where English was the main medium of communication or instruction.

The interviews started with a general introduction to reinforce the purpose of the interview. During the interviews, the key criteria, expectations and issues that significantly influenced the respondents' view of quality in HE were elucidated. The interviews were semistructured in that certain primary questions were identified initially, but the conversations were conducted in an open-ended manner designed to promote a free flow of discussion so as to access the perspectives of the interviewee in depth. The interviews, as recommended by Ratcliffe (2002), were not just used as a means of data gathering but involved active interactions aimed at arriving at contextually-based findings by examining the 'why' as well as the 'what'. Such an emergent approach is typical of qualitative research, as it seeks to observe and interpret meanings in context and, therefore, it is neither possible nor appropriate to finalise questions before data collection has begun (Hoepfl, 1997). To an extent convergent interviewing was adopted so as to allow the researcher to refine the questions after each interview. In convergent interviewing, the researcher asks questions about issues raised in previous interviews, to find agreements or disagreements between the interviewees with explanations for any disagreements (Rao and Perry, 2003). Hence, a few probe questions identified in each interview were developed for subsequent interviews so that agreements and disagreements among the interviewees and the reasons were examined.

The probe questions were introduced only towards the end of the interviews so as not to lead the interviewees. Some of the questions were intended to clarify understanding and perceptions of each group regarding the different criteria and variables and the reasons behind such perceptions.

The researcher was an active participant in the interviews by seeking clarifications or asking for further amplification; however care was taken to be non-directive and to not use leading questions. The respondents were encouraged to be reflective and thus helped to shape the actual research from the very beginning. The main themes put forward by the interviewees were explored and the semi-structured nature of the interviews provided latitude in further exploring the range of issues identified by the interviewees, which allowed their own perspectives and reasons to emerge. Questions such as "Can you give an example?" and "Can you elaborate?" were used in the course of the interviews. The rich qualitative data from the interviews helped to gain deeper insights into the input, process and output dimensions of quality in HE and also helped to further refine the components of quality identified from the literature. Another objective of the interviews was to help identify commonly used terminology and particularly complex issues and differences in interpretation and perception among the three sample groups. Potential areas of misunderstanding and misperception due to different cultural backgrounds and experiences were explored. The general understanding of all groups in terms of the expressions and terminology was closely monitored and helped identify the areas/terms which had to be dealt with carefully in the questionnaire so as to avoid ambiguity.

Given the complexities of social phenomenon, it is indeed dangerous to assume that the meaning of such phenomenon is unambiguous. What has to be carefully considered is whether or not the majority of individuals would ascribe the same meanings to those topics as intended by the researcher. Four interviews were first conducted as a pilot study with two students and one with each of the other groups which were not included in the sample and allowed the researcher to refine and develop a relevant line of questioning.

Following Yin's (1994) recommendations, an interview protocol was used to ensure consistency in analysing the interview data and improve reliability. The protocol consisted of the interview schedule (Appendix 2) which recorded details of the respondents and several loosely structured interview questions that served as a means for gently probing for

information around the objectives of the study. The interview data was written down in the interview schedule provided with spaces for notes. Consistent abbreviations of certain typical words were used to reduce the time taken. As soon as an interview was completed, the transcription was checked to ensure that all points had been completed coherently while details were still fresh in the researcher's memory. The schedule helped to structure the information i.e. from neutral demographical facts to more detailed and subjective information and provided a sound foundation for thematic analysis, which was considered appropriate as it is more exploratory than content analysis (Miles and Huberman, 1984).

Creswell (1994) advocates that there is no single correct method of qualitative data analysis. He suggests that researchers first identify themes and categories, a process called decontextualisation, through data reduction and interpretation. The researcher can then interpret the data to bring out the larger picture, which is called recontextualisation. As a first step, the cases in this study were divided on the basis of the stakeholder groups. The transcriptions were then reviewed several times so that data could be categorised according to basic themes that surfaced from each stakeholder group. Labels were assigned to each theme, without making connections among them in the first instance.

Four main themes or dimensions initially surfaced from the data: teaching and learning, curriculum, resources and outcomes. Next, the data was organised as sub-themes (referred later to as quality criteria) under these main themes. The sub-themes were categorised using the criterion of best fit. However, due to the complexity and interrelatedness of HE there were many variables that could be categorized under more than one theme. For instance, subject knowledge could be categorized under curriculum, teaching and learning and assessments. Here, the responses to probe questions helped to clarify which aspects were of relevance to the respondent. Cases which did not seem to belong anywhere in the first instance were placed separately. This category was again carefully scrutinised at the end and several of these variables were then placed under the first four main themes. Most of the variables that did not fit under the main themes identified so far, dealt mainly with broader generic institutional issues and were placed under a new theme labeled 'institutional factors'. The last theme to be identified from the data was 'admission criteria' under which two sub-themes, student aptitude and commitment, was placed. All the variables that were identified were then analysed further so as to locate key patterns, similarities and differences particularly within and between each stakeholder group.

A content analysis of the data also focused on differentiating between actual experiences, expectations, values, and practices which could be used for later stages of the study. This process also allowed the information to be compared and contrasted with the secondary data from the literature review. The organisation and analysis of the information enabled the confirmation of known general criteria (deductive) as well as provided evidence from which to infer new criteria (inductive). The key issues thus revealed by each group were then organised for integration into the survey questionnaire.

### 3.3.4 Quantitative Data: Survey questionnaire

The second research question seeks to explore the differences or similarities in the perceptions of the three stakeholder groups about key quality criteria in HE. This was accomplished through a quantitative survey including a wider sample of the population using structured questionnaires. The intention was not to find what percentage of respondents in the population responds in a particular way so as to establish a law like regularity which is often a main criticism of the use of quantitative methods in the study of social phenomenon (Maggs-Rapport, 2001). Instead, the objective was to find whether the findings from the interviews are likely to be true of the larger population and, hence, retains the essentially interpretative nature of the study. A case study methodology was not considered appropriate as the objective was to identify similarities and differences between the stakeholder groups which are not biased by institutional factors. Focusing on a particular institution would provide the views of only that institution's stakeholders and not the wider population, and hence any proposed framework of quality management based on these views would have limited validity.

The weakness of a highly structured survey method involving a large sample is that although it may provide the required breadth to the study, it may not provide the necessary depth and understanding of all the issues involved. Although the emphasis in large scale surveys is generally on quantitative data, the possibility of designing the questionnaire to collect only quantitative data or a mix of both qualitative and quantitative data was considered. The limitation of using closed-ended or fixed response items is that they may oversimplify the complexity of some opinions and limit the choices to the pre-determined

frame of reference. Further, closed-ended questions tend to over exaggerate the satisfaction or positive reactions of the respondents (Batchelor et al., 1994). Nevertheless, they are generally easier for respondents to answer and have fewer missing data than open-ended questions (Fowler, 2002). Furthermore, as the issues involved in this particular study are multifaceted and complex, it was felt that it may be over ambitious to ask a large sample of respondents to construct their own individual reflections. Although descriptive data could have been obtained from a larger sample, the analysis of the data and its interpretation would have been problematic given the complexity of the phenomenon under study. Data from open-ended questions also would not provide the necessary evaluative information nor provide adequate consistency in responses which is necessary for identifying differences and similarities in values between the three groups.

The main advantage provided by using questionnaires in the second stage of this study, was the standardisation and uniformity in the data-gathering process which was considered important in order to be able to interpret and contrast the findings between the different groups. The danger of limiting the depth and breadth of the study was overcome by the exploratory interviews in the first stage thus providing depth to the survey instrument as well as helping to understand and interpret the overall results of the survey. Therefore, a detailed questionnaire consisting of fixed-response items based on the various criteria that underpin each quality dimension, all of which were identified from the interviews was considered to be ideal and in line with the objectives of the study. The challenge was in creating a standardised questionnaire that did not oversimplify the complexities and interdependencies of the concept of quality and was at the same time respondent friendly.

The questionnaire was designed to be self-administered and was structured so that all aspects of the questionnaire were clearly specified and undisguised. i.e. the purpose of each question was made clear to the respondent. The draft questionnaire was piloted with 15 staff, 6 employers and 30 students who were not part of the final survey. The student and employer questionnaires were prepared in both English and Arabic taking into account the local context, as many students and employers in Oman may be more comfortable with Arabic. Hence, any tendency to misinterpret any of the terms or statements due to language problems was minimised. The pilot study provided valuable feedback on whether the respondents will ascribe the same meanings to those topics as intended by the researcher. The respondents were asked to comment on any perceived ambiguities, omissions or errors

concerning the draft questionnaire. Based on the feedback received some changes were made, for instance, three statements were rephrased so that the meaning of the sentence and their context was more clear. The revised questionnaire (Appendix 3) was then submitted to two senior academics for feedback before being administered for the full-scale survey.

A total of 64 items divided into 8 sections were compiled for inclusion in the final version of the questionnaire. As Nunnally and Bernstein (1994) note it is very unlikely that single-items can measure any complex theoretical concept and, therefore, as a number of related items were included under each section the likelihood for getting consistent answers and reducing random answers was improved.

The criteria from the qualitative data were grouped under the following six dimensions, each with a range of pertinent statements that asked respondents to rate the importance they attach to various criteria on a five-point Likert scale anchored at: least important (1) to most important (5).

- 1. Admission criteria
- 2. Institutional factors
- 3. Curriculum and content
- 4. Resources
- 5. Teaching and learning
- 6. Outcomes and assessment

Sections 7 included a few general statements from the literature with the purpose of cross-verifying and further analysing stakeholder responses on issues identified from the interviews. Sections 8 was aimed at determining the extent of the respondents' agreement with various definitions of quality in HE and the purpose of HE as identified from the literature on a five point scale ranging from (1) strongly agree to strongly disagree (5).

A three on the five point scale was considered as the middle ground of opinion, representing neutrality. An introductory section requesting for limited organisational and personal demographics was also included. This section differed slightly for each stakeholder group and furnished the researcher with the respondents' gender, organisation, level of study (students), programme of study/ teaching (students and academics), designation and

working experience (academics and employers) while the core part of the questionnaire was identical.

## 3.3.5 Quantitative Data: Sample

The survey focused on private sector HEPs located in Muscat which is the capital of Oman. Muscat offers the advantage of having the largest number of HEPs, students and employers in comparison to all other cities/regions in Oman. Moreover, the majority of HEPs in Oman offering undergraduate programmes for at least 5 years are located in Muscat. It is considered necessary to consider the minimum limit of 5 years for HEPs in order to ensure that the institutions are relatively mature and have had sufficient exposure to the local environment in terms of institutional policies, student needs as well as market trends. The focus on private HEPs is based on the importance given by the MoHE on developing the HE sector in Oman through the strategy of increased private sector participation. Out of the total 36 HEPs in Oman as on December 2006 offering undergraduate degrees, 22 are private HEPs while 14 are government institutions (Guidebook for Universities and Colleges, MoHE, 2006). The government HEPs includes the one university and regional institutions such as the technical and educational colleges having centralised programmes and management. Very few of the 14 government owned HEPs offer an extensive range of academic courses as most are designated for specific disciplines namely education and health and vocational diplomas.

Although the private sector HEPs are affiliated to international universities, they are strictly regulated by the MoHE both in terms of the number and the range of programmes they offer. Most of these institutions offer undergraduate degree programmes in at least three broad disciplines such as business, IT, engineering, arts etc. Very few of the HEPs in Oman offer post graduate programmes.

Out of the total 36 HEPs, 27 have been in existence for at least 5 years, out of which 13 are in the private sector. Furthermore, 12 of the 13 private HEPs established before the 5 year period are located in Muscat. Out of the 7 HEPs offering programmes in Business Studies and Information Technology questionnaires were distributed to 5 institutions which agreed to take part in the study.

- 1. College A: (offering both IT and Business programmes)
- 2. College B: (offering only Business programmes)
- 3. College C: (offering both IT and Business programmes)
- 4. College D: (offering only IT programmes)
- 5. College E: (offering both IT and Business programmes)

In addition to the students and academic staff of the above HEPs, 30 of the largest employers of graduates in the two disciplines in Muscat were included in the survey. The employers belonged to various sectors including banking, oil and gas, government ministries, telecommunications and other services.

Sampling technique for the survey was two-tiered and non-random: first the institutions were deliberately selected and then individuals within the institution were selected on the following basis:

- **Students:** All students on the final year of degree programmes in Business and IT disciplines in the 5 institutions were included in the study. A total of 290 questionnaires were distributed based on the student numbers reported by the institutions and 204 completed questionnaires were returned
- **Academics:** All academic staff teaching on the undergraduate programmes in the two disciplines in the selected institutions were included. 95 questionnaires were distributed and 74 completed questionnaires were returned.
- **Employers:** The study focused on senior line managers and human resource managers of the 20 largest companies in Oman in terms of staff, representing a cross section of both government and private sector organisations in Oman. 60 questionnaires were distributed to the 20 organisations, out of which 48 were returned.

# 3.3.6 Quantitative Data: Data Collection and Analysis

The possibility of email and mail survey was not considered a viable option in Oman particularly for gaining access to students who are resident in different areas in Oman and may not have access to an efficient email or mailing system. Letters and e-mails were first sent to all institutions for permission to distribute the questionnaires. The questionnaire was handed personally to contact persons in each HEP who in turn distributed it to the

appropriate students and staff within their respective institutions. The completed

questionnaires were collected back by the researcher within a week. The questionnaires for

employers were personally handed over to the human resource department in each

organisation by the researcher, and collected back within 1-2 days. In some cases the

employer questionnaires were faxed back to the researcher.

In a quantitative survey a large proportion of non-responses will reduce the sample size

thereby considerably increasing sampling error. Substantial non-response can also result in

bias if the non-respondents are not a random sub-set of the entire sample, in which case the

results will not be representative of the entire population (De Rada, 2005). A total of 445

questionnaires were distributed to academics, students and employers. Of these, 323 were

returned and nine discarded due to incomplete responses, thus giving an overall response

rate of 70.6 %. A small number of missing data points were replaced with scale-average

scores.

The response rate for each stakeholder group was as follows:

• Academics: 75.8%

• Students: 66.9%

• Employers: 80%

The data was analysed using Statistical Package for Social Sciences (SPSS) version 14. In

the first step, the demographic profile of the respondents in terms of stakeholder group,

gender, subject area, designation and organisation were analysed. Further analysis was

carried out as outlined below:

3.3.6.1 Consistency estimates

In order to determine reliability, the survey data was statistically analysed to establish

internal consistency of the instrument, by estimating how consistently individuals respond

to the items within a scale. Cronbach's Alpha is a measure of internal consistency and

measures how well a set of items measure a single construct. If the inter-correlations are

high, then this indicates that the items are indeed measuring the same underlying construct.

The minimum standard for these measures in the social sciences is 0.71 (Bryman & Cramer,

74

1995). Cronbach's alpha across all 64 items was found to be 0.945 which meets the standard for internal reliability. The internal consistency of the measurement scales within the questionnaire was assessed for the 6 quality dimensions which were also found to be above the recommended standards (Table 1).

Table 1: Internal consistency of scales

Dimension/ Scale	Cronbach's alpha
Admission	0.713
Institutional Factors	0.779
Curriculum	0.841
Resources	0.829
Teaching & Learning	0.925
Outcomes & Assessments	0.857

#### 3.3.6.2 Kruskal Wallis Test

Means were calculated for the 42 criteria for each stakeholder group which were then ranked with 1 being the highest rank and 42 the lowest. However, although the means of ordinal data are used by many researchers, it must be noted that means from ordinal data may not be precise or accurate since the intervals between points on the Likert scale do not have a consistent meaning (Jamieson, 2004). Therefore, percentages of responses above and below the mid point of the scale (i.e. 3/average/neutral) were also examined for each criterion.

Given the ordinal level of measurement of Likert scales, non-parametric statistical techniques were employed to further examine the data for patterns and concordance between the three stakeholder groups and between the two subject disciplines. Congruence of quality values or the extent to which students, academic staff and employers considered the various quality criteria to be important was examined using the Kruskal Wallis test for n independent samples. The Kruskal Wallis one-way analysis of variance by ranks is a non-parametric method to determine if n samples are from the same population i.e. if significant differences exist between n groups. Where the probability of the stakeholders sharing the

same value was less than 5 per cent, this was considered an indication of significant difference or lack of congruence (p<.05). The chi-square critical value for 2 degrees of freedom and the 0.05 level of risk is 5.991, below which it can be accepted that there are no differences between the samples (Lind et al. 2005). However, when we reject the hypothesis that the populations are identical using the Kruskal Wallis test, it cannot be stated how they differ. Therefore, the percentages of respondents who rated each statement below and above the midpoint/neutral (3) and the simple means of each criterion helped determine the importance/agreement attributed by each group.

## 3.3.6.3 Mann Whitney Test

In order to examine if any differences existed on the basis of subject areas, the responses of students and academics was analysed using the Mann Whitney test for 2 independent samples. The Mann-Whitney test is a non-parametric method to determine if two samples are from the same population, similar to its parametric counterpart, the two-sample t-test. Where the probability of the respondents in each subject area sharing the same value was less than 5 per cent, this was considered an indication of significant difference or lack of congruence (p<0.05).

# 3.4 VALIDITY, RELIABILITY, GENERALISABILITY AND ETHICAL ISSUES

The concept of validity refers to the fit between the solution provided by the research and the reality (Amartunga et al, 2002). In other words, validity refers to the basic trustworthiness of the whole research process: the instruments, the data collected and ultimately the findings (Bernard, 2000). Validity is concerned with establishing correct measures for the concepts under study (construct validity), the accuracy of the findings (internal validity) and the extent to which the findings can be generalized (external validity) (Miles and Huberman, 1994). There is a cycle here with the validity of the data entirely dependent upon validity of the instruments used for collecting the data. Both being valid, the findings and conclusions must naturally arise from the data. All surveys require the researcher to decide on a range of critical issues concerning sampling, non-response, questionnaire design and administration that influence the ultimate accuracy of the results (Fowler, 1993). The validity of an account or measure depends on the accuracy with which it represents those features of the phenomena that it is intended to describe, explain or

theorise (Hammersley, 1987). Internal validity is concerned with two issues, the accuracy of the measure or instrument and whether it actually measures what it is intended to measure. For this particular study, it was essential that a questionnaire dealing with the multidimensional concept of quality captured all relevant dimensions, complexity and interdependence. Content validity was enhanced by further investigating the preliminary concepts that were initially identified from the theoretical literature through in-depth qualitative exploration. The interview data helped to build the various constructs in the questionnaire, providing a balance between breadth and depth of the relevant issues and also helped to frame the variables and concepts in the right context.

The design of the questionnaire presents one of the most critical and preventable threats to survey validity as poorly designed instruments lead to erroneous conclusions being drawn. Positivist research emphasise reliability wherein the instrument or technique consistently produces the same result when applied repeatedly to the same subject. Reliability however indicates that the results are reproducible but does not in any way indicate that the results are correct. The extent to which the results are correct/true depends on the extent to which the instrument or technique satisfies the requirements of construct validity. It is important that respondents' answers not reflect differences resulting from the design of the questionnaire itself but should indicate actual differences in respondents' views, attitudes, perceptions etc. (Fowler, 2002). A key factor that influences generalisability and reliability of a survey is the wording of the questions (Madu, 1998); given the complexities of social phenomenon, it is dangerous to assume that the meaning of such phenomenon is unambiguous. Díaz-Martínez and Navarro (1997) argue that the transparency and generality postulates of positivism make the assumption that words or linguistic expressions used in a study have unambiguous meanings that denote objective realities and that these meanings are known by all the subjects of that study or at least the overwhelming majority of subjects. Of particular concern in the larger survey in this study was the possibility of differences in understanding of terminology between the different stakeholder groups. As all stakeholder groups with the possible exception of students the majority of whom are Omani nationals come from a diverse multicultural background, this was identified as a potential problem. Furthermore, while academics may have common interpretations of a particular term in education, it is possible that students or employers may not have the same interpretation. Differences in perception may also exist between members of the same stakeholder group. As these issues were particularly explored during the interviews, this helped the researcher to treat these concepts carefully so that the statements framed in the questionnaire were clear as possible to all groups. Triangulation of questionnaire statements was established in the instrument design through two or more carefully worded questions that facilitated relationship constructs of the criteria from different angles.

Vital issues that were given due emphasis by the researcher especially when piloting the questionnaire include clarity and consistency so that questions were clear and conveyed the same meaning as far a possible to respondents, particularly in this case, to each stakeholder group. The use of Arabic and English questionnaires also ensured that the chances of misinterpretation of statements due to language problems was minimised. During the pilot study, the questionnaire (both Arabic and English) was discussed with some of the respondents after they had completed it in order to determine their interpretation of the key terms and statements. As the researcher does not speak Arabic, an Arabic speaking academic who also helped translate the questionnaire was present during this discussion.

Consistency of response or the lack thereof, is also an important factor in determining internal validity. In a self-administered anonymous questionnaire there is no opportunity to question inconsistencies in response from one question to another. Such inconsistencies may arise because of a lack of seriousness on the part of the respondent towards the survey itself. Concerns that key variables or particular dimensions may be prone to measurement error in self-administered questionnaires owing to respondents' haste or lack of understanding can be alleviated to an extent by the repetition of key variables in different ways thus reducing random answers. Hence, multiple items were incorporated in the questionnaire for each dimension and some of the statements in the last two sections of the questionnaire re-emphasised key findings from the interviews.

Sample triangulation was achieved by conducting multiple interviews with each stakeholder group and an extensive survey. Methodological and data triangulation which refers to the use of 'complementary, symmetrical parallel' methods so as to get alternative views (LeBlanc, 1995, p. 6.) was achieved through the use of both interviews and quantitative surveys. Furthermore, the research design provides for sequential triangulation which refers

to the use of both methods in sequence such that the data set from one approach (interviews) feeds into the planning of the next approach (survey). The underlying assumption in such a combination is that the respective weaknesses of the two methods will be counter-balanced by their respective strengths. In using triangulation, however, it is stressed that combining qualitative and quantitative methods is not necessarily to validate the findings from either method, particularly the use of quantitative methods to validate qualitative findings. Such qualitative or quantitative findings should be valid on their own. The objective of a combined methodology does not entail combining methods that yield comparable data but must include alternative methods that provide complementary data.

In order to achieve external validity, surveys are often used so as to arrive at widely applicable statistical generalisations while a smaller qualitative study such as a case study may try to generalise the findings to a particular theory (Riege, 2003). Interpretation of findings in any study has to be based on a total view of the sampling frame. External validity refers to the extent to which findings of a particular study are applicable beyond the immediate sample or outside the specific research setting in which the study was carried out (Reige, 2003). This requires the sample to be representative of the entire population being researched (Stenbacka, 2001). Accordingly, the selected sample for the survey included final year students and academic staff of five out of the seven private HEPs in Muscat which offer the selected subject disciplines, so the findings are applicable to private sector HEPs in Oman that offer the relevant programmes. As the sample of employers included many of the larger private and public sector organisations their responses can be considered applicable generally to employers in Oman. However, as argued by Bernard (2000, p.52) the whole concept of validity is never completely resolved when considering the view that the "truth is never final" and that validity itself depends on the "collective opinion of researchers" and can never be proven absolutely. The breadth of the sample in this study has been a deliberate choice in order to enhance the external validity and generalisability of the findings to Oman in general within similar subject disciplines. Depth of understanding and knowledge of the nuances involved is vital in order to provide a theoretical explanation of a general phenomenon (Hyde, 2000).

Stenbacka (2001) argues that the issue of validity in qualitative research is useless as the purpose in qualitative research never is to measure anything. She contends that validity refers to the extent to which the qualitative researcher has understood the respondents'

reality on the specified area of study. The interaction between the respondent and the interviewer must provide for improved possibilities for collecting 'good' data, where those aspects which are important to the respondent can be pursued in depth (p.551). Indeed, indepth exploration of quality values and criteria that are specially considered relevant to managing quality in HE by each stakeholder group was given careful attention by the researcher during the initial exploratory interviews. The nature of the particular concept under study may also have its own potential issues of concern. For instance, according to Madu (1998) most studies on quality deal with perceptions rather than the actual experience which makes it difficult to translate these perceptions to actual feelings and can potentially further aggravate the issue of misperceptions. In this study, however, as the objective is to identify the views of respondents on what they consider is most important to managing quality rather than determining the respondents' perceptions of their actual experiences within a particular institution, the potential problem of equating perceptions to actual experience does not arise.

Qualitative researchers debate the applicability of the notions of validity and reality to interpretative research as these terms refer to the evaluation of positivist research. Guba and Lincoln (1994) offer that the interpretative tradition must instead aim for 'credibility, transferability, dependability and conformability' (p. 105). Credibility deals with the problem of reality being interpreted differently by different individuals and is parallel to the concept of internal validity. It involves the approval of the research findings by peers or other interviewees i.e. gaining credibility from others. Transferability is similar to external validity and Riege (2003) argues that if research findings are to satisfy the notion of transferability, adequate descriptions must be provided for readers to determine whether the findings are applicable or transferable to their own settings. By its very nature, the small size and inductive approach of qualitative research is criticised for its lack of external validity (Sykes, 1991 cited in Stenbacka, 2001). Creswell (1994) argues that repetitiveness of results or external validity has no place in inductive research as the objective is not to generalise findings but to form a unique interpretation of events. The solution is for qualitative researchers to be most careful, systematic and reflective in making the research process clear for themselves in order to describe it for others who can then judge its applicability to their needs. This requires the researcher to be explicit about the methods used so that the reader obtains a complete picture of the context and processes involved. Dependability is analogous to reliability and considers whether the procedures and techniques used in the inquiry process are consistent and requires compatibility between the research question and the research design. Confirmability is similar to objectivity in positivism and refers to whether the interpretation of the data is logical and unprejudiced i.e. whether the conclusions drawn are appropriate to the data. By interweaving quotes by respondents under each stakeholder group and differentiating the conclusions that can be drawn from the actual findings as supported by the data, careful consideration has been given to provide a clear and accurate description of the context and processes followed.

A qualitative researcher becomes the instrument of data collection and thus results may vary greatly depending on who conducts the research. Strauss and Corbin (1990) emphasises the theoretical sensitivity of the researcher which refers to the researchers' personal trait whereby s/he is aware of subtleties in data, has the capacity to understand and give meaning to data and is able to separate the pertinent from that which isn't. The qualitative researcher requires a great degree of accuracy in reporting events, points of view and interpretations that are attributed to the people who they study (Richards, 2003). Therefore, qualitative researchers must go beyond description in order to arrive at the meanings and concepts that are understood by the participants (Gregory, 2005). Richards (2003) asserts that contrary to the general perception "qualitative inquiry demands rigour, precision, systematicity and careful attention to detail" (p. 6). This becomes even more complicated given the fact that people are often not sure of exact meanings and may provide at best vague, inconsistent interpretations and may frequently change their minds. It would be difficult to converge all such interpretations into one consistent picture (Riege, 2003). The researcher's task then, is to report the indecisions of the subjects involved rather than implying a stability not provided by the subjects.

Researchers are ethically responsible to ensure that their research methods and instruments are of the highest quality and as unobtrusive and inoffensive as possible (Fogelman, 2002). All researchers have the ethical responsibility of reporting the findings of their study as accurately and truthfully as possible and they should satisfy themselves that the techniques proposed are appropriate. They are also morally obliged to safeguard the interests of the subjects of their study or those who are affected by their work. Fogelman (2002) also highlights the concept of informed consent of respondents which is tied in with anonymity and confidentiality. He observes that many surveys which do not require respondents to state their names are termed as anonymous while in actual fact the researcher may have

devised a means for identifying respondents using some sort of numbering system. Researchers, therefore, have the moral obligation to ensure that what they have indicated as the objective and process of the study is indeed what actually happens. The names of all interviewees, respondents and institutions that took part in this study have, therefore, been kept completely confidential. A preamble to the self-administered questionnaires explicitly stated the purpose of the survey and the anonymity and confidentially of respondents' views was stressed. Anonymity was important in encouraging respondents (particularly students and academic staff) to be truthful about their opinions, thus reducing the potential bias that may result when respondents try to conform to what is professionally or socially desirable or expected. Respondents were informed that the objective of the survey and interviews was to determine their views on generic quality management rather than their views on a particular institution, a fact which was clear from the nature of the questions.

#### **CHAPTER FOUR**

# FINDINGS AND DISCUSSION: QUALITATIVE DATA

The focus of the first research question was on identifying criteria considered important to quality in HE by academics, students and employers in Oman. This chapter presents and discusses the findings of the interviews with each stakeholder group which informed the construction of the questionnaire for the next stage. Six main themes or dimensions (i.e. teaching and learning, curriculum, resources, outcomes, admissions policies and institutional criteria) with various sub-themes were identified which are discussed individually in this chapter.

## 4.1 EMERGING THEMES

All interviews started with the question, "what do you think of first when I say the words: quality of higher education?"

# Students' responses essentially included:

- "reputation of the institution",
- "good teaching",
- "teachers' ability to make students understand the subject"
- "faculty who are interesting and enthusiastic",
- "the currentness and content of the programmes,"
- "programme clarity,"
- "well-structured programmes that are relevant for employment", and
- "the extent of knowledge and skills gained".

# Responses of academic staff focused particularly on:

- "the academic standards of programmes",
- "relevancy of the curriculum towards workplace related skills",
- "the content of the curriculum,"
- "the delivery and implementation of the curriculum",
- "the enthusiasm and commitment of teaching staff and students towards the learning process", and

• "the value added to the student in terms of knowledge, skills and personal development".

# Employer responses identified the following:

- "a challenging process which raises the general awareness and overall exposure of graduates",
- "facilitating the development of self-management, commitment and adaptability to work".
- "instilling the desire for continuous personal development"
- "the overall development of the individual in terms of knowledge and skills", and
- "soft skills" as fundamental to quality.

Students identified key indicators ranging from faculty, programme and outcome related factors with the strongest emphasis on faculty related factors and the manner in which they deliver the courses, institutional reputation and relevance of programmes to employment. This emphasis is supported by Telford and Masson (2005) who identify the key values of students to be those associated with what the courses are designed to achieve and the manner in which they are delivered and supported. Academic staff emphasised the academic standards of programmes, curriculum content and the process of delivery, while a few acknowledged the role of faculty and student commitment and the extent of value added to the student in terms of knowledge and skills as synonymous with quality of HE. The responses of academic staff are strongly consistent with Chua (2004) who find that the faculty's perspective of quality is wider in view in that they consider the focus should be on all aspects of HE. The main focus for employers was the HE process which they felt should be challenging enough to enable students to acquire relevant skills and knowledge. They particularly valued general awareness, personal development and core transferable skills as key outcomes of HE which is consistent with Srikanthan and Dalrymple (2004) and Hawawini (2005) who stress that employers value generic and behavioural skills more than the acquisition of subject knowledge.

The data relating to the key quality related criteria were categorised on the basis of the six main dimensions as discussed below:

# 4.1.1 Teaching and Learning

Not surprisingly, the teaching and learning function emerged as a key theme identified by almost all students, many academic staff as well as employers. Students tended to place more responsibility on the teaching and learning process and by default the lecturer for stimulating their interest in the subject and "providing them with adequate knowledge". This is consistent with Cook (1997) who finds that students consider academic staff factors as most critical to their success. A Business Administration student felt that "one of the most important issues in college education is …the way the programme and the subjects are taught should be interesting to motivate students and also should provide good knowledge about important topics and subjects".

Academic staff emphasised the importance of the teaching and learning process in ensuring appropriate delivery of programmes and in stimulating interest and enhancing knowledge, while ensuring that specified academic standards are met. A lecturer of Strategic Management put it thus "The delivery of the programmes, particularly the whole approach to teaching must be appropriate so that ultimately standards are met" while a Head of Department noted "In order to get students involved it is very important that the process of teaching and learning is stimulating and interesting".

Employers also stressed the necessity of ensuring rigour and challenge within the teaching and learning process so as to ensure important outcomes. As one employer emphasised "the process of learning during the students' time in the college/university should challenge and engage the student enough so that students develop themselves and they become mature and responsible. If this happens satisfactorily then everything else will fall into place".

The various criteria or sub-themes that emerged from the data under teaching and learning included:

**4.1.1.a Faculty:** Under this sub-theme, several related factors were identified. Almost all students and many academic staff identified the lecturer as the main actor in the students' learning experience.

One student noted that faculty "represents the institution and greatly influences students' opinion of the college". Five students, four academic staff and one employer emphasised that lecturers must have "impressive/strong communication skills, warm/confident personality, empathy with or understand students' needs" in order to positively influence students' attitude and interest in a subject and in learning. The lecturers' personality and teaching style were considered important in that they can "inspire and motivate students" in general and can "influence students to move in the right direction". In addition, two students stated that lecturers should have "interesting/impressive" personalities in that they inspire "students' confidence in the tutor" and "a pleasant appearance" which in turn influences students' "interest in the subject". When questioned further about the role of tutor's appearance, the same Business student answered that "the way the tutors present themselves affects our views about them, as it tells us if the tutor is really interested in students' opinions". Clearly appearance, personality and tutors' personal efforts were linked together in this student's mind, a factor which is not mentioned in the literature.

While academics felt that their qualifications and experience have a significant impact on their performance and capability, a few students indicated that qualifications are secondary to lecturers' ability to teach. Four business students noted that the ability of teaching staff to make complex concepts accessible and "easier to understand" was more important than their qualifications. However, students also stressed that the tutors' knowledge of the subject area and their ability to make theories relevant through "real life examples" was very important. Students appreciated lecturers who constantly updated their teaching, teaching materials and content by using the latest examples and applications. Students expected lecturers to be stimulating and interesting and to use teaching methods that "break the routine and make us think differently". One statement was that "some lecturers did not take the trouble to make the subject matter interesting" and another student noted "we did not even understand why we had to study some topics or subjects". As students did not understand the relevance of many topics they were not motivated to develop their knowledge or understanding further. This underlines the role of the lecturer as enthusiastic teachers and facilitators who display sufficient empathy with students' learning needs and motivate them to enhance their knowledge as noted by Hill et al. (2003).

Academic staff and employers expected lecturers to be "good role models for students" and regarded their input as vital in shaping student attitudes, making them think and developing

and enhancing behavioural skills. The role of academic staff in "focusing and developing students' attention, enthusiasm and interest" in the subject was emphasised by many academic staff and is consistent with students' response. One lecturer put it thus: "the success of lecturers' lie in whether they can make students enthusiastic about the module" and whether "they provide students with strategic direction in the classroom and outside with regards to all matters related to that module". The expertise and subject knowledge of the lecturer, their qualifications and breadth and depth of knowledge of the subject are key characteristics that were identified as having a significant bearing on the effectiveness and quality of the teaching and learning process. However, similar to students, academic staff also stressed that effective communication and presentation skills were probably the most important characteristics of effective lecturers, as from their experience there are many highly qualified and knowledgeable academics who have unimpressive communication skills and hence are ineffective in the classroom. Two lecturers highlighted the ability of lecturers to plan, organise and present the subject as important.

One employer and a Head of Department noted that faculty members are generally hired based on qualifications and not necessarily industry experience. Such lecturers tended to be very theoretical without using relevant industry examples which was considered important both for IT and Business disciplines. Furthermore, it was felt that some academic staff tended to have weak skills of creativity, presentation and leadership. Strong academic credentials and subject knowledge often do not necessarily guarantee strong personal skills profile and as pointed out by the Head of Department such lecturers tend to be poor at encouraging skills development in students. Academics, particularly those with managerial responsibilities emphasised the lecturers' ability to structure, organise and integrate teaching, assessments and deliver modules in a way that enhances learning clarity and accessibility to students.

Employers repeatedly stressed the role of teaching staff in encouraging and shaping students' personal values and development, rather than only subject understanding. Hence, employer's focus was on the lecturers' role in ensuring outcomes in terms of learners' overall personal development, general awareness and skills; the importance of this outcome was repeatedly highlighted by employers. These findings are consistent with Hawawini (2005) who find that employers of business graduates are increasingly demanding behavioral and societal skills as well as critical skills. As the Managing Director of an IT

company put it: "although HE may not be as critical as the schooling years in laying the foundation for students' development, a university student is still developing himself. The input from professors and lecturers will determine his values towards work, knowledge and the society and also influence his behavioral skills".

Most of the desirable characteristics of lecturers on this wish list and the significance of the lecturers' role in the teaching and learning process find strong support from the literature (e.g. Gilbert et al. 1993; Hill et al, 2003). These researchers identify key characteristics of effective lecturers to include enthusiasm, rich understanding of the subject, insightful planning, organising, strong interpersonal skills, presentation, effective communication and empathy with students. Hence, the faculty is, one, if not the most significant input that influences the quality of HE as viewed by all three stakeholders.

4.1.1.b Teaching methods: The teaching function and use of "interesting" teaching methods was a key criterion for many students, however, their interpretation of 'interesting' was ambiguous and varied. Many students were unable to elaborate when asked for examples of particular teaching methods and most students did not respond very enthusiastically when asked whether they would like to be more involved in the classroom. Students seemed to link interesting teaching methods merely with the lecturers' ability to explain the subject in a way that is not boring. A few (three) business students highlighted the role of student presentations and class discussions and expressed their satisfaction with the more interactive teaching methods used by some lecturers as they felt that it improved "their overall understanding of the topic" and "forced them to think about concepts rather than just listen to the lecture". In contrast, IT students did not seem to have any strong views about teaching methods. They tended to place more importance on handouts and teaching materials and on practical aspects of the course in particular. There was more emphasis on learning by "lecturer demonstration" in the first instance, "followed by practice".

An Accounting student stated that she expected lecturers to use similar or consistent teaching approaches/styles. She did not appreciate wide differences in teaching approaches among staff as she found it difficult to adjust to them. This preference is broadly supported by Pennington and O'Neil (1994) who recommend that students, irrespective of modules, or

programme of study must have a broadly similar experience of teaching and learning which may be achieved by establishing a set of operational principles across different curriculum.

Some of the most critical aspects contributing to high quality education have been identified as relating to particular teaching and learning styles (Hill et al. 2003). The relationship between the use of appropriate teaching methods and deeper approaches to learning is also emphasised by Pennington and O'Neil (1994). However, not all lecturers seemed to be aware of this relationship. The lack of any particular views on the part of IT students regarding teaching methods was consistent with the responses of IT lecturers who did not emphasise or mention the importance of teaching methods as such. What they did emphasise was the importance of making students understand the subject matter and ensuring clarity of explanation. This together with the fact that IT lecturers exhibited a marked lack of awareness of the importance of using different teaching methods on probe questions seemed to imply a tendency to rely on traditional tutor-centred approaches to teaching. In comparison, five business lecturers stressed the role of non-traditional and interactive teaching methods in enhancing students' interest in the subject, depth of understanding and in generating discussion around the topic. However, consistent with the lack of enthusiasm displayed by many students towards a more participatory role in the classroom, these lecturers felt that the majority of students tended to be very conservative in their teaching preferences and are most motivated when concepts were straightforward and focused, rather than complex and application oriented. Moving students away from tutorcentred learning was considered a major problem and lecturers remarked that some students were "extremely negative towards student-centred learning approaches" and even considered that "lecturers were not doing their jobs" or "did not know how to teach if they moved away form the traditional lecturer-centred teaching approaches". The fact that many students were not enthusiastic about non-traditional teaching methods indicates a gap between some of the more innovative Business lecturers and the expectations of the majority of students. Students' reluctance to be more interactive and their resistance to student-centred approaches is consistent with Hewitt and Clayton (1999) who also find that for students, the emphasis is on being taught rather than learning pro-actively. One frequent observation among teaching staff was that students, particularly towards the end-of the semester, wanted lecturers to focus on preparing them for assessments and were not really interested in learning for its own sake.

4.1.1.c. Focus of lecturers on subject knowledge: Without exception, students and many academic staff placed the onus on the lecturer for enhancing students' subject knowledge. Many students also did not see the necessity for wider or independent reading and seemed to place the entire responsibility for ensuring understanding of the subject on lecturers. In comparison, lecturers emphasised their role in drawing attention to fundamental theories and concepts and making complex concepts more accessible but expected more participation from students in taking responsibility for their learning.

When the issue of student responsibility was explored further, seven students indicated that it was their responsibility to prepare themselves for assessments and learn the subject matter. However, as one of them put it, they felt that the responsibility for "making them understand' was fully the lecturers'. Moreover, only one student acknowledged the importance of developing independent learning or thinking skills and the relevance of these skills for performance, both for employment and HE. Academics were very clearly aware of these student expectations but accepted that they were responsible for developing students' knowledge and awareness but only up to a point. They stated that there was considerable resistance from students about accepting responsibility for independent learning even among the best students and expressed certain helplessness in overcoming this. The main issues were centred on the difficulties of motivating and supporting students who were used to traditional teaching and assessment methods in the secondary school system, and the difficulties of learning in a second language. Academic staff seemed to be in dilemma; although many recognised the importance of student responsibility, they did not seem to have any idea of what they could do to enhance it. Some of them placed this responsibility on the management of HEPs, the MoHE and the general community so that the student community is made aware of the importance of participation and learning to learn. This helplessness on the part of academic staff could also be based on the fact that most of them are expatriates while the majority of the student population are Omanis and they, therefore, cannot handle an issue which may have cultural and social implications. In comparison, employers were more emphatic that HEPs and academics should make it "clear to students that unless they acquire effective skills for learning, thinking and doing for themselves, they would not be successful both in HE and outside of it".

The findings indicate a lack of awareness of the concept of students as co-participants in HE among student respondents and a few academic staff. The majority of students did not seem

to realise that HE is not merely about understanding what was taught in the classroom, but that they would have to go beyond this process and start thinking and learning for themselves. These findings are consistent with Hewitt and Clayton (1999) who suggest that students may consider themselves merely as inputs in the educational process to be transformed without putting in the required effort. The challenge for the lecturer is to convince the student to put in the necessary effort required for effective learning. However, the researcher did notice that such awareness could be developed by exploring this issue further with students and by linking the benefits of student participation to personal development and employer expectations.

4.1.1.d. Focus of lecturers on developing skills: There was consensus among all three stakeholder groups that developing and preparing students for the workplace by focusing on skills and subject knowledge was very important. Clearly, theory and subject knowledge for its own sake do not have much value which is strongly consistent with Stefani (2005) who observes that although HE was traditionally more concerned with the transmission of knowledge, today's knowledge economy requires tutors to develop a different skills set that involves seeking, analysing and using information.

While students emphasised the importance of enhancing subject and practical knowledge, most were emphatic that the role of HE is to prepare them for the workplace. However, students' interpretation of what this preparation entailed was less clear. Furthermore, students' interpretation of skills varied; some seemed to equate practical understanding of the subject with the skills appropriate for the workplace while others were not very sure about what employers required. An IT student said, "Most students go to college so that they can get jobs. ......it is important that a graduate is able to do his work properly and this requires practical knowledge of the subject". When questioned further about generic skills required for success, his response was thus, "I think practical understanding of the topics and having the knowledge to complete tasks is what is required from a College degree. If you are able to do this, then you will develop the communication and self management skills on the job through work experience".

Students also did not seem aware of the distinction between skills, knowledge and learning to learn. When the different skills sets were explained to students, the majority felt that soft skills such as communication, presentation, people and team skills, time and self

management were important but most did not seem to place much value on the higher level cognitive skills such as analysis and application. While many business students were aware of the need to develop skills of presentation, communication and dealing effectively with people, none of the students seemed to realise that developing such skills could be a core part of the overall learning process in HE and that the success of this required them to be effective participants.

While academics as a group were the most emphatic about the role of HE in preparing students for the workplace, there was lack of clarity about how these skills could be developed and who was responsible for this process. Academics (mostly business) emphasised the importance of "effective skills of communication, critical thinking, team work and innovation" as critical to the workplace, which was echoed to an extent by business students. They also noted the importance of instilling in students a desire for continuous learning. Similar to some of the students' responses, a few of the IT lecturers seemed to relate practical aspects of the discipline to the workplace and did not seem to be realise the need to develop other skills in their students.

However, employers bemoaned the lack of IT graduates with soft skills. An IT manager stated "IT graduates do not even seem interested in presenting themselves well in job interviews; there seems to be a total lack of awareness of the importance of making a good impression on people or on developing soft skills". Another stated, "we need IT graduates who can understand the problems and issues and can communicate well with other staff and departments". Hence, while IT managers valued soft skills, IT lecturers seemed to be unaware of the importance of enhancing these skills.

The literature indicates that teaching staff have a vital role in developing critical skills by offering guidance about what is required, setting structured, explicit goals, facilitating, coaching and designing customised learning experiences and providing feedback for improvement (Srikanthan and Dalrymple, 2007). However, academics did not seem to be very clear on the lecturers' role in developing generic transferable related skills. Although they stressed the importance of skills, there was less clarity on the actual process of skills development. Their responses seemed to indicate that while their role was to develop subject knowledge, the assessment strategy and the overall learning process would place the onus on students to further develop transferable skills. On being asked whether the students

will be able to successfully acquire such skills on their own just because the assessment strategy demands it, academic staff did respond that "a large majority could not", essentially because of the "weaknesses of the secondary school system". The responses clearly lend support to Lammers and Murphy (2002) who find that lecturers do not necessarily stimulate thought, change attitudes or develop behavioural skills that are essential in HE. The lack of clarity among lecturers regarding their role in skills development could be a main reason for the lack of awareness among students regarding skills. Given the inherent weaknesses in the secondary school education, the role of lecturers in skills development in HE becomes even more critical.

However, employers stressed the role of lecturers and curriculum in developing transferable skills and awareness. They considered the key objective of HE to be the "overall development of students into mature individuals" who have acquired (at least) "the minimum level of skills and abilities to learn and develop" himself/ herself further, in essence, transforming students. Thus, equipping students with "the skills of learning, communication, self-management, thinking for themselves, team and people skills and adaptability" was vital for employers. This is consistent with Srikanthan and Dalrymple (2004) who state that employers placed greater value on generic skills, including the ability to think laterally and general awareness as opposed to discipline-specific skills and knowledge. Although employers did state that subject-specific knowledge was considered very important particularly for more 'technical' jobs such as accounting and IT, which is consistent with Morley and Aynsley (2007) who find that scientific and technical employers placed greater emphasis on graduates' subject knowledge, they nevertheless emphasised that such knowledge lost value if they are not accompanied by generic skills. They stressed that subject knowledge should not be at the expense of relevant skills. These skills closely match what is considered important by academics with the exception of behavioural skills which were only mentioned by employers.

As a training manager in a large multinational company put it: "if a new employee is adaptable and has good skills of thinking and reasoning, then s/he will be able to learn quickly on the job; however with poor skills of adaptability and only subject specific knowledge the person will take a long time to become productive. Even then their contribution will be limited".

4.1.1.e. Contribution of other students: Owlia and Aspinwal (1996) observe that as the quality of students' input is vital to the transformative role of HE, the degree and quality of personal interaction between students will have important influences on each other and can greatly influence the quality of the process and outcomes. However, not many respondents identified this as a major contributor to the overall learning experience which once again highlights the general lack of emphasis on student participation. The three students who did mention the role of other students as an important criterion (one Business Administration and two IT students), were very emphatic that group work, attitude and commitment of other students in the classroom and outside of it were key determinants of the quality of their overall learning experience. The Business Administration student expressed it as "learning together and from each other" and cited examples of specific modules where this had happened, as a result of which his learning experience was richer in comparison to his other classes where students were not that keen. He was one of the few students who placed the blame for this lack of enthusiasm on students themselves, stating that "the attitude of the students, their maturity, and family background and personal values" were key factors that affect students' commitment to studies. He noted that "the more open and committed the students were, the more they had to share with each other and learn from each other". He observed that this enthusiasm also influenced lecturers who would then be motivated to try more interesting methods of teaching. In contrast, two IT students identified the role of competition among students as a key motivator that helped them to learn both with each other and individually.

A probe question explored whether students were influenced by comments by their friends and other students regarding lecturers, modules or generally about an institution. Although some students did not want to commit to this, many did indicate that negative comments from the student population particularly their friends, did tend to influence them which is consistent with Hill (1995). A few, however, stated that they often changed their negative perceptions after going through the experience on their own and they found the reality to be positive.

## 4.1.2 Curriculum

While curriculum was clearly seen as very important, there was a marked difference between the three groups in their perception of what the curriculum should focus on. Most academic staff seemed to feel that the range, relevance and currency of topics covered were the most critical aspects of achieving good standards. In fact, six lecturers identified the content of the programme as probably the single most important input that determines the quality of HE. As a lecturer in management stated "quality in HE is first and foremost about ensuring the right modules and content relevant to the programme". When questioned about how she defined 'right content,' she responded that this would be "content that prepares students adequately for industry/workplace without employers having to invest further in training the student". However, there seems to be inconsistency among academics as to what such content should cover. While some lecturers indicated that academic standards of the programme are determined by the range and depth of topics included, a few (three) preferred a focus on specific topics with substantial depth. As one Accounting lecturer put it, "sometimes the range of topics covered especially at levels 2 and level 3 is at the expense of a more in-depth knowledge of fundamental topics". He noted that "undergraduate HE seem to be focusing on developing managers rather than junior level or middle level staff with the result that the product is not really fit for any of the roles".

Two students emphasised programme clarity, as they preferred to have more structured programmes that included specific core or essential modules relevant to that discipline as opposed to many different options. While IT students emphasised that "knowledge of the latest applications" and advancements in the field were important, students in general were not really concerned about the range of topics covered. They felt that it was sufficient to focus on the most important, fundamental and relevant topics. Surprisingly, however, this view was also echoed by employers and a few of the lecturers as stated above.

Employers felt that curriculum should focus on in-depth knowledge of selective topics identified as fundamental to the subject and the application side of those subjects. Once such knowledge is acquired, employers felt that it would provide the basis on which students can further enhance their knowledge, either through masters programmes or professional courses depending on personal/career aspirations. Employers also stated that while IT and accounting students probably needed to have stronger subject specific knowledge as these subjects are more "technical", even this could be focused rather than spread over a wide range of areas. They, therefore, emphasised the "quality and depth of a few subject related topics" rather than a broad range and that "programmes should have specific learning objectives". While employers agreed that the curriculum should be current

and up to date, particularly in the use of "latest technologies, trends, case studies, applications", they were quite clear that the role of undergraduate education is not to "teach everything". This, they felt, was impossible and not required and the resulting outcome was unsatisfactory when graduates started working. A Marketing Director emphasised: "The focus of the curriculum should be on building a strong fundamental knowledge of subject areas with emphasis on developing flexibility, adaptability and application. Too much subject matter, especially advanced theoretical concepts would result in the student just studying for the examinations and not really appreciating their importance or even understanding them properly".

A significant change highlighted by employers is that nowadays, most applicants seeking their first jobs have a first degree, compared to the past where having a first degree was comparatively rare and, therefore, considered adequate for middle-level managerial positions. The demands on managers are now very significant and require, in addition to educational qualifications, extensive experience as well. Hence, employers stressed "specific knowledge for specific jobs" while the transferable skills acquired by students "should enable them to enhance their on-the- job learning". These findings are consistent with Thomas (2007) and Eagle and Brennan (2007) who note that economic and demographic changes have highlighted the need for more flexible curricula with greater emphasis on lifelong learning rather than reproduction of subject knowledge. They are also supported by Baruch and Lemming (1996) who suggest that business graduates require more specialized or focused knowledge in the early stages of their career while more generalised knowledge became important later on.

Kember (1997) notes that the role of curriculum in influencing teaching staff to focus only on subject knowledge rather than the development of critical skills must not be overlooked. Accordingly, while many Business lecturers highlighted the importance of curriculum focus on the development of skills, IT academics did not. They stressed the practical aspects of the curriculum in association with effectiveness in the workplace, while they ignored the importance of including skills in the curriculum altogether.

# 4.1.3 Learning Resources

While resources were not overtly emphasised by the interviewees, the focus by students and staff was mainly on teaching facilities, up-to-date and adequate library and IT resources. The "range of texts and journals available to students and staff" was seen as critical. IT students emphasised the importance of having access to the latest versions of relevant hardware and software programmes. General campus appearance and facilities for extra curricular activities such as sports facilities, and student forums such as IT clubs and student magazines so as to add variety and fun to campus life were mentioned by 2 students. When asked if they felt that students would participate in such activities, both students and a few staff felt that given the local culture it may take time for wider participation; nevertheless, they stressed that such facilities were necessary to provide all round development and in time participation would improve. A few lecturers also pointed that the staff-student ratio was very important and the tendency to have larger lecture sessions was not seen as very conducive to learning, given the difficulties of learning in a second language. While HE was traditionally concerned with the transmission of knowledge larger lecture groups may have been acceptable, however, as Hill et al (2003) argue, small seminar groups are necessary in order to provide the necessary opportunities for students who have difficulty adjusting to HE.

## 4.1.4 Outcomes

While the quality of the outcome in terms of graduates' preparation for the workplace was identified as a key indicator of the quality of an HEP, there was less clarity on the most appropriate vehicles for such measurement. The sub-themes that emerged under the outcome dimension included assessment methods, student performance in assessments and assessment of subject specific knowledge vs. skills.

4.1.4.a. Assessment methods: Stefani (2005) emphasises that assessment is integral to student learning and that it essentially drives the curriculum. Three Business lecturers felt that it is the assessment "process" which has a more significant influence on the quality of HE rather than outcomes in the form of high grades. This observation is consistent with Hewitt and Clayton (1999) who find that academic staff consider the quality and integrity of the overall learning process which requires students to be proactive partners, rather than the

outcome, to be the most critical factor. Two lecturers emphasised the use of a variety of assessment methods in order to develop different skills and not merely subject knowledge. As one of them stated "...programme teams must incorporate a varied assessment strategy rather than only traditional examinations, if they are really serious about developing skills". Three lecturers felt that a continuous assessment strategy involving quizzes, presentations, short essays and tests would be ideal for developing skills and subject matter and to keep students focused. However, the same staff observed that a continuous assessment strategy would be very demanding for students, given the time demands involved. One lecturer mentioned the role of feedback on assessment performance which she felt had the most constructive role in influencing student learning.

Students did not mention the role of assessments in influencing the quality of their learning. On being questioned, many students felt their assessments were challenging and appropriate to their programmes. A few of them stated that when multiple course-work was involved, the challenge was more of managing their time rather than the assessments themselves. When asked whether such an assessment strategy would develop their time and self-management skills, students responded positively but some still felt compelled to state that given more time and less pressure they would be able to perform better on assessments, which they perceived as important. Students did not have much to say about assessment feedback indicating a lack of awareness of the critical role of feedback in improving learning.

The general consensus was that a variety of assessment methods must be employed across a programme of study to help foster the development of core transferable skills and understanding of the subject. Most academic staff felt that while examinations were appropriate for assessing some modules depending on the nature of the subject matter, exams were the least useful of all assessment methods in developing or assessing skills of creativity, critical thinking, communication and presentation. However, two IT lecturers did state that examinations were the most suitable assessment method for their subjects. Some students particularly Accounting and IT students, indicated that they were happier with examinations than take away assignments, presentations etc. This may imply that they find traditional examinations less challenging in terms of overall requirements in comparison to other forms of assessments which is consistent with the literature that students are more comfortable with conventional forms of assessment. Other than two employers who

mentioned the role of assessments together with the teaching and learning process in providing a challenging process for students, none of the employers mentioned the importance of assessments

**4.1.4.b.** Student performance in assessments: While performing relatively well in assessments was indicated as important, many interviewees felt that high grades did not necessarily indicate a distinctive quality of HE. The general perception was that quality in HE is much more than a set of grades, percentages and degree classifications.

Academics and employers felt that high grades and high pass rates, although indicative of a good level of student performance are not really indicative of good quality provision. Essentially, many interviewees felt that that it would be relatively easy to record high levels of performance in assessments, if HEPs restricted their intake to high achievers but this did not necessarily indicate that the quality offered by the institution was exemplary. Similarly, not achieving high grades may not necessarily imply a lack of quality in the process, particularly if the value added to the student in comparison to the original intake is considerable. Academic staff noted that performance in assessments was influenced by factors such as general student motivation, the support mechanisms available (e.g. workshops on examination techniques), the extent of preparation expended by the lecturer towards assessments in class and students self-management skills. They also pointed out that standards of assessments may not be consistent across institutions and, hence, it may be easier to obtain higher grades in some institutions compared to others. Some felt that "assessments may not really be assessing critical or key outcomes" and, therefore, "it cannot be assumed that high grades would exemplify all the required outcomes". This view supports the existing evidence that students tend to adapt surface learning strategies in accordance with the requirements of the task (Struyven et al., 2002) and getting good grades can merely indicate that the student was successful in meeting task requirements but not necessarily in enhancing subject understanding or skills.

One employer in fact stated that he would be reluctant to employ graduates with very high grades unless it was backed up by a strong personality and well rounded skills that reflect the grades. As he put it: "I would be very suspicious of students who get very high grades. In my experience I have found that such employees lack initiative and creativity, they tend to be good at appearing for examinations but such abilities are not very useful at work".

The researcher did ask employers if their recruitment procedures ignored grades particularly in the case of fresh graduates. Four of the respondents from government ministries and multinationals stated that when the number of applicants was large, grades served a means of screening out applicants.

Students' main concern regarding assessments was that they are prepared and guided adequately, i.e. "teachers should not design assessments which are aimed at testing what we don't know, the aim of the assessment should be to test what we do know." When questioned further, although some students felt that high grades were not necessarily an indicator of quality, performing well in assessments did make them feel good and more motivated to learn. Students also pointed out that overall poor performance of a group is indicative of problems i.e. "If many students fail in the exams or assignments then there is something wrong".

4.1.4.c. Assessing subject specific knowledge versus assessing skills: Consistent with their views on the purpose of HE, academics and employers were emphatic that assessments should focus on both subject specific knowledge and key transferable skills. However, when it came to assessing skills, students were hesitant and uncertain about what this implied. The uncertainty seemed to arise from their lack of understanding and appreciation of what assessing skills would entail. There also seemed to be some apprehension that this would result in assessments which are more difficult than those assessing only subject knowledge. There also seemed to be a lack of appreciation among students that mere passive reproduction of subject knowledge is no longer an adequate preparation for the work place. In contrast, most lecturers and employers were emphatic that assessing transferable skills and subject knowledge is the natural outcome of a teaching and learning process and curriculum focused on transforming students and developing mature individuals.

#### 4.1.5 Admission Policies

Strict admissions criteria and resulting quality of student intake is considered a key determinant of the quality of output in the literature, but this was not identified by any of the respondents. Three sub-themes emerged from the data.

4.1.5.a. Student Aptitude: Two academic staff identified aptitude of students for certain disciplines that were more technical (e.g. computer science and accounting) as a key determinant of students' success and motivation on the course. An IT lecturer put it thus, "often many students join IT and computing programmes as they feel it may be easier and not that language dependant. Such students may not really have the aptitude for logic and programming and hence fare badly".

**4.1.5.b.** Attitude and commitment of students: The literature states that students' role as coparticipants in the process of education is vital. One student identified the importance of students being mature enough to respond "positively and fully" to learning. When questioned further, she clarified that it is not important to restrict admissions only to the best students in terms of performance at secondary school. She felt that student involvement in learning had more to do with "emotional maturity and personal values" rather than actual grades obtained or entry qualifications.

At least six academics expressed strong views on students' attitude and overall interest in the process of learning and in developing their knowledge and skills. They emphasised positive attitude and commitment towards learning and personal development as a key factor that improves the quality of the overall learning process. As one of them put it, rather than focusing on "high entry qualifications, colleges should have some other way of filtering out those students who are really interested in developing themselves and those who are just in because everybody else is doing it or as a result of parental pressure". When asked, "what do you suggest that HEPs do in order to identify the really interested students from others, if not focus on marks obtained at entry level?", she suggested "that one way could be to conduct admission interviews where students have to demonstrate their interest in HE and how they would help themselves learn". It was pointed out by one lecturer that while many part-time working students could really contribute to the learning process as they were in a position to appreciate and apply what they learnt to the workplace, some of them were only interested in obtaining a certificate as they had already secured "good jobs" in the government sector.

A Business studies lecturer noted, "The issue is not whether the students are the most intelligent or have the highest grades. What makes all the difference is when you have students who are really keen and interested in learning and in developing himself/herself".

Academic staff noted that as far as full-time students were concerned, motivation to learn was generally good except for a certain proportion who either just wanted a certificate in order to get a government job or who were there because of their parents. However, most students were used to and thus expected only traditional tutor-centred approaches to learning and assessment. Variables that affected student involvement in the learning process were identified as lack of understanding and interest in the educational process and low levels of self-confidence on the part of learners.

Employers considered attitude of students towards others, and their acceptance of responsibility to be more important than their grades. In fact, some employers felt that students' attitude to work, personal development and responsibility would be a key indicator of students' general attitude to work in future.

4.1.5.c. Range and variety of programmes on offer: Wright and O'Neill (2002) note that as a result of changing lifestyles and societal changes, students expect HEPs to offer a variety of courses with a range of campus and delivery options that are easily accessible and user friendly. Five students and two employers noted that that HEPs should offer a wider range of programmes and courses to choose from, depending on students' interests and preferences. It was felt that this choice was lacking in Oman at present.

4.1.5 d. Strict admissions criteria: Although the quality of student intake is considered by many to be a necessary condition for institutional success (Cheng and Tam, 1997), none of the respondents identified strict rules for admissions and high quality student intake as an important criterion. Some of the academic staff indicated that the value added to the student is what is more important in determining quality of provision of an institution. i.e. a comparison of before and after. As one put it, "Of course it might make our jobs easier as teachers if all students were the cream of the lot. But I would question the notion that quality means you take only the best students because the institution would then have to do very little to add value". This criterion was, however, added to the questionnaire as there is a lot of debate in the literature on the impact of admissions criteria on the process and outcomes of HE.

#### **4.1.6** Institutional Factors:

This was one of the last themes to be identified under which the following sub-themes that were relevant to HEPs in general were included.

4.1.6.a. Institutional Reputation: The reputation of the institution and the resulting prestige associated with studying in a particular institution were considered very important by students. Institutional reputation in terms of employability of graduates was also very important to students. Academics considered institutional reputation to be a key criterion in attracting good students and faculty and in ensuring employability of graduates. Employers admitted that in some instances, recruits were short listed based on their past experience with graduates of particular HEPs or the reputation of the HEP. This practice is consistent with Morley and Aynsley (2007) who find that recruitment decisions by employers in the UK are most often based on institutional reputation. However, this comparison between UK and Omani HEPs is probably not valid considering that UK HEPs have had a significantly longer time to establish their reputations in comparison to HEPs in Oman which have been established only over the last decade. Hence, employers in Oman would not really have sufficient information or time to make accurate decisions based on institutional reputation. Accordingly, one employer did state that as many Omani HEPs were quite new, it was not really fair to graduates to rely too much on institutional reputation.

4.1.6.b. Academic standards: The majority of academics and employers indicated that they expect institutions to be "representative of certain standards" and were very emphatic that standards are maintained. Institutional reputation was considered to be linked to perceived "academic standards that are representative of certain outcomes, i.e. "the abilities of graduates". While many students stressed that the reputation of the institution is a key indicator of quality, only two students actually linked this to academic standards and graduate outcomes. As a student stated: "the college, the course and lecturers must challenge students so that we can improve ourselves. If the standards are not high then students will not be interested in learning and the reputation of the college will not be good. If the level of education (meaning standards) is low, then later on it will be known that the graduates of that college are not good".

However, other students were not clear on what they expected from institutions with good reputations, other than that they were perceived to be superior. In fact, many students were not aware of the association between quality and academic standards while some considered academic standards as the ability of HEPs to ensure that "students perform well". These findings are consistent with Rolfe (2002) who observes that while most students see a degree as a route to better career prospects many are indifferent as to whether or not this involves high academic standards.

4.1.6.c. Links with industry: Two academic staff identified the need for "networking" and "practical cooperation with local and international enterprises". They pointed out the value of establishing strong links with local industry as well as multinational companies so as to enhance student and staff awareness of industry trends and developments. Specific potential benefits as a result of networking were identified as "practical, industry-oriented teaching, relatively higher acceptance and acknowledgement of graduates in the community, exposure to guest speakers from the industry and other key organisations". Other benefits of closer cooperation with industry included "developing case studies, consultancies and opportunities for customised training courses". Staff felt that unless such close links are established, HEPs would function in isolation, with the result that faculty and students would be less aware of the latest industry developments. The ultimate relevancy of IT and Business courses lay in "preparing students for the industry". In order to do this, teaching staff must have appropriate opportunities to update themselves with the latest trends whether in "human resource policies", "accounting and taxation systems", "information systems and use of latest information technologies and their applications" or even "the general usefulness and limitations of theoretical applications in the actual workplace".

4.1.6.d. Extracurricular activities: Although one might expect extracurricular activities to be important to college students generally, it was mentioned by only three students. However, the students who did identify this as an important criterion were very emphatic about the role of such activities in their overall learning experience. They considered them to be opportunities to "develop closer bonds with the institution" which can have a positive influence on their overall development and attitude.

**4.1.6.e.** Attendance requirements: Two students felt that HEPs should enforce strict attendance requirements for its students. Being Omanis themselves, they felt that many Omani students irrespective of age were not mature enough to take the right decision when given a choice and should be forced to attend classes. In this way they would not miss out on classes without which they would not have the skills to be successful on their own.

# 4.2 STAKEHOLDER DETERMINANTS OF QUALITY

The ability of HE to facilitate the successful development of students in terms of a portfolio of "generic transferable skills" was the main recurring theme among employers and academic staff. Students and academic staff were firm in their conviction that the purpose of HE is to produce graduates who are equipped with the "right level and range of skills and knowledge for the workplace". However, although students stressed the importance of preparing for the workplace, they were not really aware of the relationship between a rigorous and challenging learning process and the ultimate portfolio of skills and abilities

There was consensus among academics and employers that although the outcome of HE in terms of knowledge and skills is very important, the process of HE is probably the most vital aspect in developing the longer term abilities of students. Academic staff emphasised standards of the programmes as synonymous with quality while employers emphasised that the HE process and the curriculum must be challenging enough to have an impact on students' overall development and abilities. Students as a group did not exhibit much appreciation of the importance of academic standards but were more concerned that the programmes met their expectations and needs, although many were not very clear about exactly what their expectations included.

Students and academic staff placed the greatest emphasis on the 'role of the lecturer' as one of the most significant input determining the quality of HE. The role of the lecturer in shaping students' classroom experience and thereby the "actual" process of teaching and learning through their teaching methods/styles was greatly valued by students and staff. Employers also emphasised the role of "lectures", lecturers and the institution in providing adequate challenges and self-development opportunities for students during their time at college. An important aspect is the complete absence of references to research and its role in

underpinning teaching. This omission, however, is consistent with the general focus in most HEPs in Oman on the teaching function.

A significant finding was the lack of appreciation by students on their own role in contributing to the teaching and learning process and how this ultimately influences the overall quality of the process and outcomes. While few of the students indicated that the responsibility for learning ultimately rests with themselves, they apparently had not realised or acknowledged this fact until some of the interview questions made them reflect on this. It was quite evident that until this reflection took place many of them tended to place the onus on HEPs and lecturers to "make" them learn and to provide a quality of experience which is not possible in reality unless students themselves are effective contributors to the overall experience. Furthermore, although the role of student participation and motivation in maximising learning was acknowledged as very important by academic staff, they did not seem to be very clear on how student expectations and participation could be managed effectively.

The literature contains numerous references to the demands that the dynamic economic environment places on industry and thereby new graduates. Employers clearly supported this and emphasised that graduates cannot afford to be complacent either about their abilities or their jobs. However, the added demands from graduates are not necessarily in terms of subject knowledge. Employers were most emphatic and focused in their view that the process of HE should be challenging enough to transform students into responsible and reflective individuals. Consistent with the findings of Hewitt and Clayton (1999), employers emphasise graduates' skills profile, their ability to adapt to different requirements and situations and to communicate effectively with people. Graduates' attitude to work and their willingness to learn was considered very vital. Although these views were echoed by academic staff, their responses indicated that they have not really considered their own role in this transformation as opposed to developing subject knowledge.

To answer the first research question, the key quality values expressed by each stakeholder group in the interviews are presented below:

### **Students:**

- Lecturers who are enthusiastic, committed, have impressive personalities, have excellent teaching skills and are able to stimulate interest in the subject;
- Teaching staff with excellent communication and presentation skills;
- Focus of lecturers on developing subject knowledge of students
- Good reputation of the institution in terms of quality of programmes and employability of graduates;
- Curriculum that is current, focused and prepares students for the workplace;
- Practical aspects of subject areas
- Teaching methods that make students think and breaks the monotony of traditional lectures:
- The availability of the latest resources in terms of Information Technology and library;
- Teaching facilities;
- Extra-curricular activities that enhance student participation, improve team work, and create interest towards the college and society;
- Adequate preparation and guidance for assessments;
- Contribution of other students towards the learning process.

## **Academic staff:**

- The academic standards of the programmes;
- Curriculum that covers a range of topics and skills relevant to the workplace;
- Enthusiastic teaching staff with good subject knowledge and impressive teaching skills and who can relate to students' needs;
- Teaching staff with excellent communication and presentation skills;
- Focus of lecturers on developing subject knowledge of students;
- Students as co-producers of learning: students' attitude and efforts towards the learning process;
- Extensive library resources and teaching facilities;
- Assessment strategies that allow the development and evaluation of skills and subject knowledge;
- Constructive feedback to students on assessment performance;
- Links with industry and international collaborations;
- Student aptitude for a programme;

• Support systems available for students.

# **Employers:**

- Acquisition of core transferable and behavioural skills;
- Curriculum that covers a range of topics and skills relevant to the workplace;
- A challenging process of teaching and learning and assessment that emphasises personal development, adaptability and general awareness;
- Focused subject specific knowledge;
- The academic standards of programmes.

# 4.3 CONCLUSION

The dimensions and criteria identified in this chapter were represented in the questionnaire in the form of 41 statements. An additional 23 statements were also included in order to cross verify and further explore perceptions on key values. The quality values identified here are further discussed in combination with the quantitative findings in the next chapter.

#### **CHAPTER FIVE**

# FINDINGS AND DISCUSSION: QUANTITATIVE DATA

The second research question was aimed at identifying any differences or congruencies in the quality values of the three stakeholder groups and the implications for HEPs. Accordingly, this chapter presents the differences and congruencies that were revealed by the larger quantitative survey using the questionnaire developed from the quality criteria identified from the interview data. It also examines stakeholder perceptions about different definitions of quality and objectives of HE. The first part of this chapter presents the findings from the survey and the latter part discusses the findings and the implications for HEPs.

#### 5.1 PROFILE OF RESPONDENTS

The demographic profile of the respondents in terms of stakeholder group, gender, subject area, designation and organisation is presented in tables 2 and 3. The total number of usable questionnaires was 314. The proportion of each stakeholder sample is representative of the actual population, with students forming the largest proportion (n=194, 61.8%) followed by academic staff (n=72, 22.9%) and employers comprising 15.3% (n=48) of the total sample.

Around 62.7% of the respondents were male and 37.3% were female. Gender representation was also proportionately maintained across each stakeholder group. The proportion of female academic staff was slightly higher (51.4%) reflecting the existing scenario in some HEPs in Oman. The proportion of male students at 64.9% was higher than female students (35.1%) which reflect the higher participation by male students in HE.

Also reflecting the existing demographic trends, the majority of students (89.7%) and employer representatives (79.2%) were Omani nationals while only 5.6% of the academic staff were Omanis. The remaining proportion of academic staff includes other Arabs, Asians and a few British and Australian nationals. In terms of subject discipline, 44.3% of the student population were IT students and the remaining 55.7% were Business students. While 45.8% of the academic staff teach on IT programmes, 54.2% were Business Studies lecturers.

**Table 2: Demographic characteristics: Overall** 

	Description	N	%
Stakeholder	student	194	61.8%
	academic staff	72	22.9%
	employer	48	15.3%
	Total	314	
Gender	male	197	62.7%
	female	117	37.3%
	Total	314	

Table 3: Demographic characteristics: By Stakeholder Group

		N		N		N	
		student	%	academics	%	employer	%
Gender	male	126	64.9%	35	48.6%	34	70.8%
	female	68	35.1%	37	51.4%	14	29.2%
	Total	194	100%	72	100%	48	100%
Subject	IT	86	44.3%	33	45.8%		
	Business	108	55.7%	39	54.2%		
College	College A	63	32.4%	26	36.1%		
	College B	40	20.6%	10	13.9%		
	College C	42	21.6%	14	19.4%		
	College D	23	11.8%	11	15.3%		
	College E	26	13.4%	11	15.3%		
Nationality	Omani	174	89.7%	4	5.6%	38	79.2%
	Other Arabs	5	2.6%	24	33.3%	3	6.3%
	Others	15	7.7%	44	61.1%	7	14.6%
Organisation	Private					30	62.5%
	Government					18	37.5%
Designation	HR Mgrs					11	22.9%
	Bus. Func. Mgrs					28	58.3%
	IT Mgrs					9	18.8%

# 5.2 QUALITY DIMENSIONS

Respondents were asked to indicate the degree of importance they attach to different criteria representing the six dimensions of quality, and to indicate the overall importance they attach to each dimension. As the mid-point in the Likert scale indicated average importance (=3), the percentage of responses that rated each criterion >3 (very/most important) and >4 (most important) were determined. Table 4 presents the six dimensions of quality in HE in descending order of means for each stakeholder group.

**Table 4: Quality Dimensions: Descending order of Means** 

Students	>3	>4	Academic	>3	>4	Employers	>3	>4
			Staff					
Teaching & learning	73.7%	42.3%	Curriculum	95.8%	61.1%	Curriculum	95.8%	35.4%
Resources	70.6%	29.9%	Teaching & learning	93.1%	43.1%	Teaching & learning	100.0%	27.1%
Curriculum	66.0%	29.9%	Assessment & Outcomes	94.4%	27.8%	Admissions Policies	87.5%	35.4%
Institutional Factors	61.3%	24.7%	Resources	84.7%	31.9%	Institutional Factors	85.4%	16.7%
Assessment & Outcomes	59.3%	33.5%	Admissions Policies	86.1%	15.3%	Resources	72.9%	18.8%
Admissions Policies	57.2%	21.1%	Institutional Factors	76.4%	22.2%	Assessment & Outcomes	75.0%	14.6%

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

Closely reflecting the interview data, students consider the Teaching and Learning dimension as the most critical aspect of HE while curriculum was given the highest importance by academics and employers followed by Teaching and Learning. However, while Outcomes are next in importance for academics, they are the lowest priority for employers. The importance attributed to the Teaching and Learning and Curriculum dimensions by the quantitative data is closely supported by the earlier qualitative findings. While Resources are ranked second by students, academics and employers rank resources lower, i.e. 4<sup>th</sup> and 6<sup>th</sup> respectively. Students also rank those dimensions that are directly concerned with their performance i.e. Admissions polices (quality of student intake) and Assessment and outcomes as the least important among the six dimensions.

## 5.3 QUALITY CRITERIA

The results of the Kruskal-Wallis Test show that of the 42 criteria, 24 criteria i.e. more than half, had significant differences in relative quality values between the stakeholder groups. These findings as well as the percentage of respondents rating each criterion as very/most important and the criterion rank according to mean value of responses are presented in the following sections:

## **5.3.1** Admission Policies

Two of the four criteria under the Admissions dimension show statistically significant differences between the three groups (Table 5).

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

Table 5: Admission Policies: Kruskal-Wallis

		Criteria	%	%		Chi-	Signific
Criteria	Stakeholder	rank	>4	>3	Mean	Square	ance
Strict	student	41	13.4%	45.4%	148.82	7.552	
Admissions	academic staff	36	16.7%	65.3%	181.40		0.023*
policy	employer	39	12.5%	52.1%	156.71		
Student	student	28	26.3%	61.9%	138.06	26.044	
Attitude	academic staff	16	50.0%	80.6%	185.69		0.000**
	employer	7	56.3%	83.3%	193.78		
Subject	student	33	29.4%	61.3%	152.26	2.129	
aptitude	academic staff	31	15.3%	84.7%	169.10		0.345
	employer	30	22.9%	72.9%	161.28		
Variety of	student	18	30.9%	67.0%	157.78	5.464	
Programmes	academic staff	35	23.6%	55.6%	142.07		0.065
	employer	17	22.9%	91.7%	179.52		
Overall	student		21.1%	57.2%		18.622	
admissions	academic staff		15.3%	86.1%			0.000**
Policies	employer		35.4%	87.5%			

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

**Strict Admissions policy**: The Kruskal-Wallis test shows significant differences between the groups (p=0.023) with 62.5 % of academics rating this criterion as very/most important in comparison to 52.1% of employers and 45.4% of students. The interviews had indicated that most respondents did not consider a strict admissions policy as an important determinant of quality. Similarly, the normal means show that out of the 42 criteria, all three stakeholders rank this criterion very low with academics ranking it at 36 and students giving it the lowest rank of 41.

Attitude and commitment of students: There are very significant differences (p<0.01) in how stakeholder groups perceive students' attitude and commitment to learning. More than 56 % of employers (rank 7) and 50% of academic staff (rank 16) rate student attitude as most important to the quality of HE. In comparison, only 26.3% of students (rank 28) share this view, thus lending support to the interview findings that students do not realise the importance of their role as co-participants in the process of HE.

The responses to the remaining two admissions criteria, i.e. **Students' aptitude** and **Variety of programmes** of study are more congruent with differences among respondents being statistically insignificant. Subject aptitude ranked 30-33 is considered more important that strict admission policies. Consistent with the interviews, providing a variety of programmes of study is considered important by a higher proportion of employers (91.7%) at rank 17 and students at rank 18, compared to academic staff (rank 35).

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

<sup>\*</sup> Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

**Overall Importance of the Admissions Dimension:** The responses show statistically significant differences between the three groups on the overall importance of the Admissions dimension with a higher percentage of employers rating this as very/most important followed by academics and lastly students.

#### **5.3.2** Institutional Factors

There are significant differences in stakeholder values with regard to three out of the five Institutional factors (Table 6).

**Table 6: Institutional Factors** 

		Criteria	%	%		Chi-	Significa
Criteria	Stakeholder	rank	>4	>3	Mean	Square	nce
Attendance	student	27	28.9%	61.9%	148.25	6.081	
	academic staff	25	38.9%	73.6%	175.88		0.048*
	employer	23	33.3%	72.9%	167.33		
High academic	student	24	27.8%	61.9%	143.58	14.252	
standards	academic staff	20	43.1%	83.3%	186.24		0.001**
	employer	18	27.1%	85.4%	170.64		
Extracurricular	student	39	23.7%	54.1%	150.69	6.820	
activities	academic staff	37	12.5%	58.3%	156.14		0.033*
	employer	25	18.8%	81.3%	187.06		
Institutional	student	19	38.1%	62.9%	153.52	4.207	
Reputation	academic staff	33	16.7%	83.3%	152.60		0.122
among students	employer	10	37.5%	87.5%	180.94		
Institutional	student	17	34.0%	64.9%	150.38	4.516	
Reputation	academic staff	27	20.8%	88.9%	162.67		0.105
among employers	employer	12	37.5%	85.4%	178.52		
Overall	student		24.7%	61.3%	150.02	3.978	
Institutional	academic staff		22.2%	76.4%	168.33		0.137
Factors	employer		16.7%	85.4%	171.50		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

**High academic standards of programmes:** Only 61.9% of students consider standards to be important in comparison to more than 83.3% of academics and 85.4% of employers, which is consistent with the interviews findings that programme standards are more important to academics and employers in comparison to students.

**Strict attendance requirements:** There is again a significant lack of congruence between academics and employers on one hand and students on the other as employers (72.9%) and academics (73.6%) consider attendance requirements as very/most important in comparison to 61.9% of students.

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

**Extracurricular activities:** Employers value extracurricular activities significantly higher than students and academic staff, possibly reflecting employers' emphasis on the overall skills development of students.

**Reputation of the institution:** Although the proportional patterns of rating are very similar without any significant differences in the importance attached to this criterion by the three stakeholder groups, employers place a higher value on the overall reputation of the HEP in comparison to students and academics.

Overall Importance of Institutional Factors: Significant differences among the three stakeholder groups indicate that employers (85.4%) consider the Institutional factors dimension more important than both academics (76.4%) and students (61.3%).

#### 5.3.3 Curriculum

The interviews indicated that although curriculum was considered very important particularly by academic staff and employers, there are differences in emphasis by the three groups, especially regarding curriculum content and subject focus. The quantitative data shows statistically significant differences (p<0.05) between the groups on all 5 of the criteria under the Curriculum dimension, indicating that there is a considerable degree of variation in responses (Table 7). The lack of congruence, however, is essentially due to the consistently lower proportion of students who rated the curriculum criteria as important.

Table 7: Curriculum: Kruskal-Wallis Test

		Criteria	%	%		Chi-	Significance
Criteria	Stakeholder	rank	>4	>3	Mean	Square	Significance
Coverage of	student	35	24.2%	56.7%	136.64	34.511	
Curriculum	academic staff	8	55.6%	87.5%	205.64		0.000**
	employer	19	27.1%	81.3%	169.58		
Curriculum	student	37	20.6%	51.5%	143.08	39.970	
subject focus	academic staff	13	48.6%	87.5%	213.81		0.000**
	employer	40	8.3%	54.2%	131.30		
Curriculum	student	8	40.7%	73.7%	147.75	9.840	
practical focus	academic staff	5	56.9%	90.3%	184.22		0.007**
	employer	9	29.2%	100.0%	156.83		
Job relevance	student	5	47.4%	71.1%	146.03	16.411	
of programmes	academic staff	2	66.7%	98.6%	192.21		0.000**
	employer	8	37.5%	93.8%	151.81		
Curriculum	student	10	40.7%	70.6%	139.32	23.964	
skills focus	academic staff	3	62.5%	95.8%	187.48		0.000**
	employer	2	62.5%	93.8%	186.00		
Overall	student		29.9%	66.0%	136.66		
curriculum	academic staff		61.1%	95.8%	203.63	34.292	0.000**
	employer		35.4%	95.8%	172.54		

>3 shows % of respondents who selected 4 (very important) or 5 (most important) >4 shows % of respondents who selected 5 (most important) Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

The coverage and content of the curriculum in terms of range of topics: While curriculum coverage and content has very high importance for academics (87.5%) and employers (81.3%), it has much lower importance for students (56.7%) at rank 35. However, a significantly higher proportion of academic staff (55.6%) rate curriculum coverage as most important, ranking it at 8, compared to only 27.1% employers (rank 19).

The emphasis of the curriculum on subject-specific knowledge: Consistent with the response on curriculum content and coverage, the majority of academics (87.5%) also rate curriculum focus on subject knowledge as very/most important, thereby complementing the emphasis placed by academics in the interviews on this factor. However, this keen focus on subject knowledge is not shared by both students (51.5%) and employers (54.2%), which is also consistent with the interviews. In fact, only 8.3% of employers rate subject focus as most important compared to 48.6% of academic staff and 20.6% of students.

Practical components in the curriculum & links to the industry: The findings show that practical skills and knowledge and relevance to industry are rated among the top criteria, especially by employers and academics. 100% of employers and 90.3% of academics have rated practical components and industry links as very/most important, while the student proportion is significantly lower at 73.7%. However, out of this proportion, while 56.9% of academics and 40.7% of students rate practical focus as most important, only 29.2% of employers give it the highest importance.

Relevance of the programme to the job market: The relevance of the programme in terms of employment is ranked second by academics with 98.6% respondents rating it as very/most important. Student proportion is maintained quite surprisingly, as in the previous criteria at 71.7% (rank 5). While 93.8% of employers rate relevance to the job market above the mid-point, only 37.5% consider it to be most important (rank 8), compared to 66.7% of academics and 47.4% of students. This is again consistent with the interviews as academics and students placed the strongest emphasis on this factor.

The emphasis of curriculum on developing skills for the work place e.g. skills of self management, communication, creativity, team work, analysis, application: A main finding from the interviews was the emphasis given to the development of skills by all stakeholders, particularly employers and academic staff. Similarly, it is found that the focus on skills development in the curriculum is considered very/most important by the majority

of the respondents i.e. 93.8% of employers and 95.8% academics. Employers' responses rank it as the second most important criteria and more important than practical or subject knowledge while academics rate it as the third. Once again, the proportion of students rating the development of skills above the mid-point is maintained as in the previous two criteria at around 70.6% resulting in a significant lack of congruence.

**Overall importance of the Curriculum dimension:** There are significant differences in stakeholder response to the overall importance of the Curriculum dimension, with academics attaching the most importance followed by employers and lastly students.

#### 5.3.4 Resources

There is congruence among the respondents on all the five criteria under the Resource dimension (Table 8).

Table 8: Resources: Kruskal-Wallis Test

		Criteria	%	%		Chi-	Significance
Criteria	Stakeholder	rank	>4	>3	Mean	Square	
Teaching	student	2	52.1%	74.7%	152.59	2.030	
facilities	academic staff	4	50.0%	98.6%	168.51		0.362
	employer	5	47.9%	91.7%	160.83		
Library	student	4	42.8%	77.8%	151.92	3.162	
resources	academic staff	11	48.6%	93.1%	172.51		0.206
	employer	13	39.6%	89.6%	157.52		
Staff student	student	14	38.1%	66.5%	152.57	2.484	
ratio	academic staff	22	36.1%	86.1%	171.26		0.289
	employer	22	39.6%	64.6%	156.77		
Campus layout	student	23	29.9%	61.3%	165.19		
	academic staff	39	2.8%	62.5%	145.29	4.056	0.132
	employer	35	16.7%	52.1%	144.72		
Sports &	student	36	25.3%	56.2%	163.35		
recreation	academic staff	42	6.9%	41.7%	139.56	4.033	0.133
	employer	37	14.6%	56.3%	160.76		
Overall	student		29.9%	70.6%	154.26		
resources	academic staff		31.9%	84.7%	173.62	3.706	0.157
	employer		18.8%	72.9%	146.41		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

The highest importance by the 3 groups is accorded to the following 2 criteria:

Quality of teaching facilities such as classrooms, IT labs and lecture halls and the range and quality of library resources:

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

The number of students in a classroom or the staff: student ratio follows with 86.1% of academic staff and 66.5% of students giving it high importance.

Campus layout and appearance and sports and recreation facilities are given comparatively lower importance by all groups, although it is considered important by the majority of respondents particularly students and employers.

**Overall importance of the Resource dimension**: Stakeholders are congruent in their response to the overall importance of the Resources dimension with the large majority rating the role of resources in influencing the quality of HE as very/most important.

# **5.3.5** Teaching and Learning

There are 16 criteria under the Teaching and learning dimension and responses indicate significant differences between stakeholder perceptions for 10 of these criteria (Table 9).

Table 9: Teaching and Learning: Kruskal-Wallis Test

		Criteria	%	%		Chi-	Significance
Criteria	Stakeholder	rank	>4	>3	Mean	Square	
Accessible	student	1	60.8%	76.3%	146.13		
explanation	academic staff	1	76.4%	91.7%	174.96	11.416	0.003*
	employer	1	75.0%	100.0%	177.25		
Faculty	student	13	38.7%	68.0%	142.33		
communication,	academic staff	6	52.8%	97.2%	184.77	16.461	0.000**
organizing, assessing ability	employer	3	50.0%	91.7%	177.92		0.000
Understanding	student	6	48.5%	71.1%	155.70		
student needs	academic staff	9	50.0%	94.4%	176.88	7.053	0.029*
	employer	31	31.3%	77.1%	135.71		
Faculty ability to	student	29	35.6%	62.4%	139.84		
stimulate thinking	academic staff	10	51.4%	93.1%	187.87	21.661	0.000**
	employer	4	43.8%	97.9%	183.31		
Faculty	student	26	27.3%	63.4%	158.87		
experience	academic staff	40	16.7%	48.6%	137.65	7.634	0.022*
	employer	20	29.2%	77.1%	181.76		
Similarity in	student	42	16.0%	45.4%	167.49		
teaching styles	academic staff	41	9.7%	48.2%	172.60	29.653	0.000**
	employer	42	2.1%	12.5%	94.46		
Student efforts	student	34	19.6%	60.8%	137.35		
	academic staff	12	47.2%	91.7%	205.88	34.934	0.000**
	employer	24	14.6%	89.6%	166.39		
Contribution of	student	32	23.7%	55.7%	137.92		
other students	academic staff	15	41.7%	94.4%	201.59	30.191	0.000**
	employer	21	22.9%	83.3%	170.50		
Interactive	student	40	21.1%	53.6%	139.79		
learning methods	academic staff	21	34.7%	90.3%	201.08	26.888	8 0.000**
	employer	28	14.6%	75.0%	163.70	]	

Additional	student	31	21.6%	61.9%	152.15		
academic support	academic staff	30	22.2%	83.3%	179.99	6.815	0.033*
	employer	38	12.5%	66.7%	145.36		
Teaching	student	3	50.0%	100.0%	152.66		
methods	academic staff	7	52.8%	93.1%	171.26	2.625	0.269
	employer	6	37.5%	75.3%	156.44		
Faculty ability to	student	7	45.9%	70.6%	154.38		
motivate interest	academic staff	14	38.9%	95.8%	169.46	1.888	0.389
in subject	employer	14	29.2%	89.6%	152.17		
Faculty	student	12	36.1%	70.6%	151.98		
qualifications	academic staff	23	33.3%	86.1%	165.94	2.130	0.345
	employer	15	37.5%	81.3%	167.16		
Faculty	student	16	37.1%	68.0%	156.44		
Personality	academic staff	26	27.8%	84.7%	164.88	871	0.647
	employer	27	18.8%	83.3%	150.70		
Preparation for	student	9	36.6%	72.2%	161.98		
assessments	academic staff	29	26.4%	77.8%	157.67	2.741	0.254
	employer	33	22.9%	64.6%	139.14		
Faculty focus on	student	11	35.1%	74.7%	149.11		
workplace skills	academic staff	18	41.7%	88.9%	171.84	5.022	0.081
	employer	16	41.7%	89.6%	169.92		
Overall teaching	student		42.3%	73.7%	152.21		
& learning criteria	academic staff		43.1%	93.1%	171.78	2.840	0 .242
	employer		27.1%	100.0%	157.44		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

Once again, the comparatively lower emphasis by students on many of the criteria is the main reason for the lack of congruence. Among these, the following factors which are directly related to teaching staff show significant differences with p<.05.

Ability of the lecturer to explain topics in a manner accessible to all students: The lecturer's ability to explain complex subject topics in an easily accessible manner so as to facilitate students' understanding is considered the most important of all the 42 criteria by the three groups and is ranked first among all 42 criteria. However, there is more congruence between employers' and academics' responses compared to students. While more than 75% of academic staff and employers have rated this factor as most important (with 100% of employers rating it either very/most important), surprisingly only 60.8% of students consider this criterion as most important.

Ability of the lecturer to communicate, organise, & assess their subject areas: Although considered highly important by all groups, the proportional patterns of rating by employers

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

and academics are very similar showing that they value these core skills of lecturers more than students.

Ability of lecturers to understand individual student needs: Understandably, the empathy that teaching staff have towards students' is very important for students (ranked 6) with 71.1% rating it above the mid point and 48.5% as most important which is consistent with the interviews. Academics also consider this criterion as playing a key role in the quality of teaching and learning with 94.4% rating it above the mid point and 50% as most important (ranked 9). While 77.1% of employers rate this as very/most important, the mean rank is only 31.

Ability of lecturers to make students think, to change attitudes and develop behavioural skills: The responses demonstrate similarity between academics and employers on this criterion. The ability of lecturers to influence students' thinking, attitudes and behaviors is highly prized by employers (97.9%) who rank it as the fourth most important criteria. The interviews had revealed that employers place great value on behavioural skills and positive attitudes, while academics and students had not really emphasised this. The results here show that academics do consider these skills as very important (93.1%, rank 10) but students do not seem to share this high regard with only 62.4% rating it as very/most important (rank 29).

The experience of lecturers in terms of number of years: In comparison to students and employers, academic staff attach less importance to their own experience. Only 48.6% of academic staff rate experience as very/most important compared to 63.4% students and 77.1% of employers.

Teachers teaching on a programme have similar teaching styles/methods: Similarity of teaching styles is given the lowest importance amongst all criteria by all groups. Although there is a significant difference in the proportional pattern of response by the three groups, this factor is considered important by less than 50% of students and academics and only 12.5% of employers.

Teaching and learning criteria for which there is lack of congruence include the following:

The effort put in by students outside the classroom to read and understand the subject material: Similar to other factors which involve student input, students do not attach much importance on their own efforts and initiative to understand the subject before or after lectures, ranking it at 34. Only 19.6% of students consider this as most important while

41.2% rate it as very important. Not surprisingly, the large proportion of academic staff particularly (91.7%) and employers (89.6%) consider this as very/most important.

The contribution of other students to a student's overall learning experience inside & outside the classroom: Only 55.7% of students attach any importance to the influence that other students can have on their own learning, in comparison to 94.4% of academics and 83.3% of employers who recognise the impact that students in general can have on each others' learning experience.

Opportunities provided for interactive student-centred learning experiences in the classroom: Similar to other criteria which require students to put in more effort, interactive teaching methods that should in essence stimulate student interest and motivation, do not find much favour with around half of the students in the sample (ranked 40). Only 53.6% of students rate it above the mid point, in comparison to 90.3% of academic staff and 75% of employers.

Additional academic support systems available to students: Although placed comparatively lower on the list of teaching and learning criteria, additional academic support systems are considered important by more than half of the respondents in each group.

The faculty related teaching and learning criteria on which respondents' are congruent include:

The teaching methods and teaching styles of the lecturers: The importance attributed to this criterion finds considerable congruence among stakeholders and has been rated either as very important or most important by 100% of students. It is ranked as the third most important criteria by students and the sixth and seventh by employers and academics respectively.

**Ability of lecturers to motivate students' interest in the subject:** Motivating student interest is ranked among the top 15 criteria by all stakeholders, without significant differences in responses between the three groups. 70.6% of students, 95.8% of academics and 89.6% of employers acknowledge that this ability of lecturers has a key influence on the quality in HE.

Students, employers and academic staff attach considerable importance to all four of the criteria listed below without any significant differences in their responses. One point of note

is that a smaller proportion of employers consider preparation for assessments as very important compared to other two groups, reflecting the lower importance attached to assessments by employers.

- Qualifications of teaching staff;
- Teachers have an impressive and warm personality;
- Preparing students for assessments;
- Focus of lecturers on developing skills required for the workplace:

Overall importance of the Teaching and learning dimension: Students, academics and employers are congruent in their response with the largest proportion of respondents rating the Teaching and learning dimension as very/most important compared to the other five dimensions.

#### 5.3.6 Outcomes

Table 10 shows that stakeholder responses are congruent without any significant differences for the following three out of the seven criteria under the Outcomes dimension:

Table 10: Outcomes: Kruskal-Wallis Test

		Criteria	Overall	%	Mean	Chi-	Significance
Criteria	Stakeholder	rank	Rank	>3	Rank	Square	_
High Grades	student	25	33.5%	60.3%	165.55	4.425	
	academic staff	38	1.4%	61.1%	145.47		0.109
	employer	34	16.7%	45.8%	143.02		
Assessment	student	21	26.8%	64.9%	153.67	1.238	
methods	academic staff	32	16.7%	83.3%	166.53		0.539
	employer	26	20.8%	72.9%	159.42		
Continuous	student	22	27.8%	62.9%	157.47	.137	
assessments	academic staff	34	18.1%	66.7%	155.18		0.934
	employer	29	20.8%	70.8%	161.10		
Assessment	student	15	27.3%	70.1%	144.44	12.209	
feedback	academic staff	17	40.3%	90.3%	181.56		0.002**
	employer	11	41.7%	81.3%	174.22		
High standards of	student	30	23.7%	55.2%	151.46		
assessments	academic staff	28	22.2%	80.6%	182.22	8.056	0.018*
	employer	36	10.4%	64.6%	144.81		
Assessment	student	20	30.9%	66.0%	150.21	9.352	
focus on skills	academic staff	19	41.7%	84.7%	184.58		0.009**
	employer	32	18.8%	75.0%	146.34		
Assessment	student	38	19.6%	50.5%	149.21	33.644	
focus on subject a	academic staff	24	25.0%	88.9%	205.47		0.000**
knowledge	employer	41	2.1%	41.7%	119.03		

Overall	student	33.5%	59.3%	150.35		
assessment	academic staff	27.8%	94.4%	183.72	8.746	0.013*
criteria	Employer	14.6%	75.0%	147.08		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important) or 5 (most important)

Achieving high grades in assessments: Although considered important, obtaining high grades in assessments is not rated as a very important indicator of quality in HE by all three stakeholder groups in comparison to other criteria. This is also consistent with the interviews, where many of the academic staff and employers stated that although high grades and high pass rates indicate a good level of student performance, they do not necessarily indicate good quality HE provision. Good grades could be the natural outcome of a stricter admissions policy and did not necessarily prove that the quality of the provision was exemplary, clearly providing more support to the value added concept of quality. Although not statistically significant, the proportion of employers who rate grades as very important are comparatively lower (45.8%, rank 34) in comparison to 60.3% of students (rank 25) and 61.1% of academics (rank 38).

- The variety of assessment methods:
- Continuous assessments in a module requiring students to work continuously:

Both the above criteria on assessments have elicited a very similar pattern and proportion of responses from the three groups. More than 62% of respondents in the three groups consider the use of a variety of assessment methods and a continuous assessment strategy as important.

The remaining four assessment criteria show statistically significant differences between the three groups:

The feedback provided to students on assessment performance: Academic staff (90.3%) clearly value the positive impact of providing constructive feedback to students on their assessment performance, followed by employers (81.3%) and lastly by students (70.1%).

Assessments set high standards for student performance: Similar to their views on programmes setting high academic standards, academic staff attach considerable importance to assessment standards. Academic staff rank this criterion at 28, while students rank it at 30 and employers 36. Interestingly and consistent with their views on assessments, although

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important)

Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

employers greatly value high standards of programmes (section 5.3.2), they are comparatively less enthusiastic about high assessment standards with only 64.6% rating it as very/most important.

- The emphasis of assessments on assessing skills for the workplace and
- The emphasis of assessments on subject specific knowledge:

Responses are significantly different with academics placing more importance on both the above criteria in comparison to students and employers. However, when comparing responses to the two criteria, it is found that while students and employers place more importance on assessing skills for the workplace, academics place slightly more importance on subject knowledge. Employers in particular do not attach the highest importance to subject knowledge with only 2.1% rating this as most important which is consistent with their response on curriculum focus on subject knowledge versus skills and the interviews.

**Overall importance of assessment criteria**: Significant differences in responses of the three groups to the overall importance of Outcomes and assessment criteria show that academic staff attach the most importance to this dimension.

#### 5.4 CROSS-VERIFICATION

This section presents stakeholder responses on several general statements on HE. These statements were included for the purpose of cross-verifying and further analysing stakeholder perception on key issues identified in the interviews.

# 5.4.1 Standards versus students expectations

Respondents' perceptions on the importance of high academic standards was explored in the interviews and in section 5.3.2, where it was found that students place the lowest importance on high academic standards compared to employers and academic staff. In order to further explore students' views on programmes meeting their needs and expectations versus the importance of standards, the following statement was included in the questionnaire:

'It is more important for students to cope with their courses easily rather than setting high standards'.

Responses to this statement (Table 11) are consistent with the previous findings on high academic standards in 5.3.2, with a higher proportion of academics and employers

disagreeing with this statement compared to students. Significant differences (p=.000) in the responses of the three groups indicate that 52.8% of academics and 54.2% of employers, disagree with this statement, in comparison to only 21.6% of students. None of the academic staff and employers express strong agreement with this statement while a small proportion (18.6%) of students do. 30.4% of students, 26.4% of academics and 27.1% of employers are neutral on this point.

Table 11: Standards vs. student expectations

		%	%	%	%	Chi-	Signific
	Stakeholder	=5	=4	=3	<3	Square	ance
Coping with	student	18.6%	29.3%	30.4%	21.6%		0.000**
courses	academic staff	.0%	25.0%	22.2%	52.8%	40.908	
	employer	.0%	18.8%	27.1%	54.2%		

<sup>\*\*</sup> Significant at 0.01 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

#### 5.4.2 Examinations

During the interviews, the researcher had noted a general assumption by some respondents on the superiority of examinations over other forms of assessment, thereby negating the importance of incorporating a variety of assessment methods. The importance of a varied assessment strategy was explored in section 5.3.6, the response to which was both congruent and very positive. The following statement was also included to test respondents' views on the appropriateness of examinations.

'Examination is the best method to test students' knowledge and skills'.

**Table 12: Examinations** 

	Stakeholder	% =5	% =4	% =3	% <3	Chi- Square	Signific ance
Suitability of examinations	student	20.6%	27.9%	22.7%	28.9%	20.716	0.000
	academic staff	2.8%	25%	25.0%	47.2%		
	employer	8.3%	10.5%	20.8%	60.4%		

<sup>\*\*</sup> Significant at 0.01 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

There are significant differences in responses with the majority of employers (60.4%) disagreeing that examinations are the best assessment method, while only 47.2% of academics disagree (Table 12). The strongest support for examinations come from students, with 27.9% of them agreeing, 20% strongly agreeing and only 28.9% disagreeing with this

statement, which is consistent with students' preference for traditional teaching and learning methods.

## 5.4.3 Student evaluation of teaching

Student feedback and evaluation of teaching is an important part of quality assurance in many HEPs. In order to determine if there is consensus among the stakeholders on this practice, the following statement was included:

## 'Students have the capacity to evaluate the teaching they receive'.

The findings (Table 13) reveal significant differences in opinion between the group that is the subject of the evaluation and the other two groups. Both students and employers believe that students are capable of evaluating the teaching they receive, with more than 72% of each group expressing agreement, in comparison to only 44.2% of academic staff.

**Table 13: Student evaluation of teaching** 

	Stakeholder	%	%	%	%	Chi-	Significance
		=5	=4	=3	<3	Square	
Student capacity teaching evaluation	Student	26.8%	46.4%	21.6%	5.2%		0.000
	academic staff	4.2%	40.2%	38.9%	16.7%	29.277	
	employer	10.4%	62.5%	27.1%	.0%		

<sup>\*\*</sup> Significant at 0.01 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

## 5.4.4 Institutional versus Student Responsibility

While all HEPs have the responsibility to provide the most appropriate resources and opportunities for student success, students must accept their share of the responsibility as a co-participant in the learning process in order to enhance its effectiveness. However, there seemed to be little awareness of the importance of student responsibility and participation in the learning process during the student interviews. In order to further explore stakeholder perception of student responsibility towards the learning process versus that of the HEP, the following statement was included:

'It is the institution's responsibility rather than the students' to ensure students are successful in their studies'

The evidence indicates that employers and academic staff clearly disagree with the above statement and expect students to take responsibility for their own learning. There is a significant difference of opinion by students, with 47.4% agreeing that it is the institution's rather than their own responsibility to ensure their academic success.

Table 14: Institutional vs. Student Responsibility

	Stakeholder	%	%	%	%	Chi-	Significance
		=5	=4	=3	<3	Square	_
Institutional	student	22.2%	25.2%	28.9%	23.7%		0.000
responsibility	academic staff	4.2%	8.3%	23.6%	63.9%	53.720	
	employer	2.1%	10.4%	29.2%	58.3%		

<sup>\*\*</sup> Significant at 0.01 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

# 5.4.5 Quality of Student Intake

Despite the emphasis in the literature on attracting a high quality student intake in order to maintain standards and quality, none of the respondents in the interviews mentioned the role of a strict admissions policy or the importance of attracting a high calibre student intake. In addition to including strict admission criteria under the admission dimension, the following statement was also included to cross-verify the earlier findings and to further explore any differences in stakeholder perceptions:

'Attracting high quality student intake is a "necessary" condition for ensuring quality in HE'.

**Table 15: High Quality Student Intake** 

	Stakeholder	%	%	%	%	Chi-	Significan
		=5	=4	=3	<3	Square	ce
High quality of student intake	student	12.9%	36.1%	33.0%	30.9%		0.007
	academic staff	12.5%	38.9%	33.3%	15.3%	9.956	
	employer	6.3%	33.3%	14.6%	45.8%		

<sup>\*\*</sup> Significant at 0.01 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

Academics are slightly more inclined to agree that good quality student intake is an important factor in ensuring quality which is consistent with the findings in section 5.3.1. Again, there are significant differences in stakeholder responses on this point (Table 15). While 51.4% of academic staff agree with the statement, 15.3% disagree and 33.3% are neutral. The strongest disagreement with the statement comes from employers, with 45.8%

who do not agree that high quality student intake is necessary for ensuring quality, while 39.6% do consider it an important condition. Student responses are not very strong either way, with 49% of students who do not consider student intake as important, 30.9% who do, while the remaining 33% are neutral.

## 5.4.6 Value Added versus high quality student intake

The following statement further explores stakeholder perception on the importance of high quality student intake versus the value added concept of quality in HE:

'The quality of the student intake is not important; the focus should be on the "value added" to the student.

**Table 16: Value Added** 

	Stakeholder	%	%	%	%	Chi-	Signific
		=5	=4	=3	<3	Square	ance
Value added	student	28.9%	60.3%	27.8%	11.9%		0.039
	academic staff	18.1%	50.0%	25.0%	25.0%	6.508	
	employer	16.7%	58.3%	14.6%	27.1%		

<sup>\*\*</sup> Significant at 0.05 level of significance. <3=disagree, 3=neutral, 4=agree, 5= strongly agree

Once again there are significant differences in responses (Table 16). However, there is a clear indication by all three stakeholder groups that they agree with the value added concept of quality more than a restrictive admissions policy that ensures a high quality student intake. The strongest support for the value added concept is from students followed by employers and lastly, academics.

## **5.4.7** Defining Quality in Higher Education

The last section of the questionnaire asked respondents to indicate the extent of their agreement with five definitions of quality in HE. It was found that the majority of respondents agree to a relatively high degree with most of the definitions. Responses to two out of the five definitions of quality show significant differences between stakeholders (Table 17).

**Table 17: Definitions of Quality in Higher Education** 

			Rank	%	%	Chi-	Signific
	Definition	Stakeholder		>4	>3	Square	ance
Not	Meeting specified	student	5	22.2%	64.9%		0.000**
Congruent	objectives	academic staff	1	44.4%	94.4%	26.392	
		employer	5	20.8%	77.1%		

Not	Efficiency and	student	3	30.4%	71.6%		0.019*
Congruent	effectiveness	academic staff	3	31.9%	81.9%	7.954	
		employer	1	37.5%	97.9%		
Congruent	Exceptional	student	4	27.3%	64.4%		0.192
	education	academic staff	4	18.1%	75.0%	3.300	
		employer	5	33.3%	81.3%		
Congruent	Achieving	student	2	34.5%	76.3%		0.069
	consistency	academic staff	5	15.3%	80.6%	5.353	
		employer	2	31.3%	91.7%		
Congruent	Transformative	student	1	38.1%	71.6%		0.185
	ability	academic staff	2	41 .7%	87.5%	3.373	
		employer	3	22.9%	93.8%		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (agree) or 5 (strongly agree)

# Definition 1: Providing education that meets specified objectives or goals

There are significant differences in responses, although a large proportion of all three stakeholder groups agree that quality in HE is about 'Fitness for Purpose: the extent to which education meets the stated purpose or goals'. The means for this definition show that academics rank this definition the highest, with 94.4% agreeing with this definition. In comparison, students and employers prefer this definition the least.

# Definition 2: The ability of the institution to be efficient and effective

Responses show significant differences for this industry based definition of quality with its focus on "achieving efficiency and effectiveness" which not surprisingly, gets the largest support from employers who rank it highest. 97.9% agree that quality in HE is about institutional efficiency and effectiveness. More significantly, the large proportion of academic staff (81.9%) also agrees with the efficiency definition (ranked 3), which is not consistent with the literature in that academics are not necessarily interested in efficiency or effectiveness. The majority of students (71.6%) although smaller than the other 2 groups also agree with this definition (ranked 3).

There are no significant differences in shareholder responses for the remaining three definitions.

## Definition 3: Providing education that is exceptional and has high standards:

<sup>&</sup>gt;4 shows % of respondents who selected 5 (strongly agree)

<sup>\*</sup>Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

Among the five definitions, the lowest proportion of the three stakeholders agrees with this definition, although this proportion includes more than half the respondents in each group. Employers rank this the last while academics and students rank it the second last.

# Definition 4: Achieving consistency in internal processes and performance:

The consistency definition of quality finds the strongest support from students with 76.3% (rank 2) and 91.7% of employers (rank 2) agreeing with this definition. 80.6% of academics also agree that quality entails achieving consistency in processes and performance but the overall rank by academics for this definition is 5 making it the least preferred definition for academics.

# Definition 5: The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved:

This definition treats transformation as translating into real empowerment of everyone involved in the learning process resulting in a process of continuous learning and enhancement. 87.5% of academics (rank 2), 93.8% of employers (rank 3) and 71.6% of students (rank 1) agree with this definition. 38.1% students strongly agree with the transformative definition of quality making this the most preferred definition by students. In fact, with the high percentage of stakeholder agreement for this definition without any significant differences in responses among stakeholders, the transformative definition is clearly considered very important by all three groups.

## **5.4.8** Purpose of Higher Education

Respondents were also asked to indicate their agreement with the following four objectives/purposes of HE (Table 18).

- Providing education and services that meets the needs and expectations of students
- Developing graduates who meet the expectations of the society
- Developing graduates who meet the requirements of employers.
- Equipping students with the skills to learn and think for themselves

**Table 18: Purpose of higher education** 

				%	%	Chi-	Signific
	Objective	Stakeholder	Rank	>4	>3	Square	ance
	Learning for	student	1	53.1%	79.4%		0.083
Congruent	themselves	academic staff	1	56.9%	97.2%	4.987	
		employer	1	60.4%	100.0%		
	Meeting	student	2	43.8%	78.9%		0.332
Congruent	employer expectations	academic staff	2	40.3%	95.8%	2.203	
		employer	2	54.2%	85.4%		
_	Meeting societal expectations	student	3	35.6%	74.2%		0.126
Congruent		academic staff	3	37.5%	93.1%	4.145	
		employer	3	37.5%	89.6%		
Congruent	Meeting	student	4	32.0%	74.2%		0.199
	student needs	academic staff	4	25.0%	73.6%	3.224	
	expectations	employer	4	18.8%	66.7%		

<sup>&</sup>gt;3 shows % of respondents who selected 4 (agree) or 5 (strongly agree)

Responses show congruence on all four of the objectives with all three groups expressing the strongest support for equipping students with the skills to learn and think for themselves. More than fifty percent of respondents in each group strongly agree with this objective. Students, academics and employers consider the need to develop graduates who meet the requirements of employers as the second most important purpose of HE, followed by meeting societal expectations. Meeting students' needs and expectation finds agreement with the lowest proportion of all stakeholders; nevertheless, this proportion constitutes more than 65% of the respondents of each group. Employers consider meeting students' expectations to be comparatively less important than the other two groups.

#### 5.5 Differences between students based on subject

The results of the Mann Whitney test revealed no significant differences between IT and Business students on any of the criteria on quality or on any of the remaining statements including the definitions/purpose of HE.

# 5.6 Differences between academic staff based on subject

The Mann Whitney test revealed significant differences between IT and Business academic staff on eleven out of the 42 criteria on quality in HE (Table 19: Appendix 1). The differences indicate that Business lecturers place more importance on the following three criteria under the Curriculum dimension in comparison to IT academics:

<sup>&</sup>gt;4 shows % of respondents who selected 5 (strongly agree)

- high academic standards of programmes
- the focus of the curriculum on practical components and the industry
- relevance of the programmes to the job market

Business lecturers also place more importance on the following seven criteria under the Teaching and learning dimension:

- Ability of lecturers to understand individual student needs
- The teaching methods and teaching styles of the lecturers
- Ability of the lecturer to explain topics in a manner accessible to all students
- Opportunities provided for interactive student centred learning experiences in the classroom
- The effort put in by students outside the classroom to read and understand the subject material
- Focus of lecturers on developing skills required for the workplace
- Overall importance of the Teaching and learning dimension

However, IT lecturers place more importance on the overall importance of the Assessment dimension in comparison to Business lecturers. IT lecturers are more in favour of examinations as only 33.3% disagree with the statement that "examination is the best method to test students' knowledge and skills", in comparison to 63.6% of Business lecturers.

Higher proportions of IT lecturers also seem to agree or are neutral on the point of institutional responsibility versus students' responsibility when it comes to students' success. Only 53.8% of IT lecturers disagree that "it is the institution's responsibility rather than the students' to ensure students are successful in their studies" in comparison to 75.8% of Business lecturers.

Among the two more popular definitions of quality in HE, 97.4% of IT lecturers agree that quality is "*Providing education that meets specified objectives or goals*" compared to 90.9% of Business lecturers. However, only 82.1% of IT lecturers agree with the transformative definition of quality in HE compared to 93.9% of Business lecturers.

Regarding the purpose of HE, a larger proportion of IT lecturers (89.7%) consider that HEPs must provide education and services that "meets the needs and expectations of students" compared to 54.5% of Business lecturers. Furthermore, a smaller proportion of IT lecturers in comparison to Business lecturers agree that the "purpose of HE is to equip students with the skills required to think for themselves", which was ranked number one by all stakeholder groups.

#### 5.7 DISCUSSION

The survey results are strongly consistent with stakeholder values as identified in the interviews. The findings revealed strong congruence on many aspects of HE between academic staff and employers. Student responses indicate that a comparatively lower proportion attach the same degree of importance to many of the criteria that are highly rated by academic staff and employers. However, there are also a number of criteria in which there is congruence between all three groups.

# 5.7.1 Defining Quality in Higher Education

Cheng and Tam (1997) and Green (1994) stress the futility of identifying a single best definition of quality particularly in HE and note that quality is best defined in terms of different qualities. Consistent with this view, the study finds considerable support from the three stakeholders for all five of the definitions of quality, thus underlining the multi-dimensional character of quality which cannot be easily assessed by only one indicator. Most of the definitions have similarities between them (Cheng and Tam, 1997) and none of the definitions in principle contradict each other. Hence, it is more practical and constructive to consider the different characteristics of quality including transformation, efficiency and effectiveness, consistency, fitness for purpose and excellence as correlates of quality when evaluating HE provision. Inclusion of all such correlates of quality will ensure that the legitimate views of stakeholders and the different perspectives of quality are addressed.

Academics agree most with the 'fitness for purpose' definition of quality defined in the questionnaire as 'meeting specified objectives' which is based on the business concept of ensuring a pre-defined outcome designed to meet specific objectives. However,

surprisingly, it is least preferred by both students and employers resulting in a lack of congruence between academics and the other 2 groups. This is one of the few definitions that refer to quality in terms of outcomes/objectives as the other definitions refer essentially to the process of HE. Srikanthan and Dalrymple (2003) suggest that this definition is most likely to find favour with employers rather than academics, as it is similar to the business concept of designing and customising products to meet particular needs. However, the findings in this study are more consistent with Watty (2005) and Lomas (2002) who find that the fitness for purpose definition is widely prevalent among academics and HEPs in Australia and the UK. The latter might explain the preference of academic staff for this definition, as three of the HEPs in the survey are affiliated to UK universities and, therefore, follow similar quality assurance systems. It is now common practice to design HE programmes, particularly in business studies around industry input and requirements (Lawrence and Pharr, 2003). There are also a range of HE 'products' (part-time courses, elearning, distance learning, etc.) that seek to address needs of particular market segments and to provide flexibility. Issues that naturally arise from the fitness for purpose definition are the validity of the stated objectives which would depend on their relevance, their acceptance by the broader community and the extent to which stakeholders are satisfied with the outcomes. The relevance of the objectives will depend on the accuracy and comprehensiveness of the market research that informs such objectives.

The idea of achieving efficiency is often misunderstood as implying the need to lower costs and thereby compromising quality (Deming, 1982). The 'efficiency and effectiveness definition' of quality found the strongest support from employers which is not surprising, considering the business emphasis on efficiency and effectiveness. It is surprising that the majority of academics and students (although to a lesser extent) agree with the definition, particularly if one were to consider the general emphasis that HE is essentially different from other services. Some researchers suggest that achieving economic efficiency and offering the best value for money may not necessarily be considered an indicator of quality in HE, particularly by academics. For instance, Cyert (1993) argued that academics believe universities are as efficient as they should be and any further efforts in this direction are unnecessary. Clearly, this is not the case in this study but one would have to consider whether the concern for efficiency and effectiveness shown by academics in this case is because they all work in private HEPs. With the support from all stakeholder groups for this definition, it may be concluded that they interpret efficiency and effectiveness as not

necessarily associated with compromise but leading to a win-win situation for all stakeholders, which is important for long-term survival.

The excellence definition of quality with its emphasis on excellent outcomes and high standards also found considerable support and congruence across all three groups. According to Srikanthan and Dalrymple (2003), the concept of quality as excellence should correspond to the student's view of quality. Nevertheless, it is ranked only fourth out of the five definitions by both students (which is consistent with their lower emphasis on standards) and academics and the last by employers. This indicates that excellence although a desirable outcome of HE is not viewed as the primary indicator of quality by any of the groups.

Although achieving a degree of consistency in all processes and performance is necessary, the consistency definition is criticized for encouraging bureaucracy, and stifling creativity and innovation (Doherty, 1997). Moreover, product control, minimising variation and avoidance of error which are some of the fundamental principles of industry-based quality models, are considered to be undesirable or inapplicable for HE with its varied outcomes and processes. The evidence, nevertheless, indicates that students particularly consider it important that quality management ensures some consistency both in internal processes and performance of HEPs, followed by employers. As students are in the dual position of being the main input and a customer in HE and considering the complexities involved in the processes, it is quite understandable that they desire the comfort of consistency i.e. knowing what to expect. However, although stakeholders' responses are congruent, academics rank this definition the last, indicating that as educators they are more aware of the inherent problems in aiming for consistency in processes and outcomes in the complex process of HE. This does not follow Srikanthan and Dalrymple's (2003) assumption that the consistency definition would be favoured by academics and administrators whose job satisfaction requires that all behavioural norms are met and the core ethos is upheld.

From the various definitions, the capacity of the institution to be transformative and ultimately empower all participants found strong support from all three groups. Becket and Brookes (2005) observe that this definition is particularly important for internal stakeholders and accordingly the strongest support was from students followed by academics as well as employers as external stakeholders. The transformative definition

based on the concept of the learning organisation is one of the newer definitions of quality in HE and is based on the empowerment of all participants by enhancing their critical abilities. It is indeed positive that the three main stakeholders groups in Oman and most importantly students recognise its significance. Srikanthan and Dalrymple (2007) identify transformation of learners by enhancing their ability to think for themselves as the highest level of achievement to which HE can aspire to, which requires innovation and enhancement in all aspects. The emphasis of this definition is on enhancement, not just quality assurance and requires the very active participation of all participants and as Becket and Brookes (2005) note, is most likely to result in a culture of quality being embedded within institutions. However, while transformation and empowerment of participants are very worthwhile goals for HEPs, the other findings from this study indicate that student awareness about their role as co-participants in the transformative process is very poor.

## 5.7.2 Purpose of Higher Education

Similar to the definitions of quality, the majority of respondents agree with all four objectives of HE that were presented without any significant differences between the stakeholders. This leads us to conclude that HE would need to meet a range of objectives centred on meeting the needs of the society, employers and students while most importantly equipping students with the critical skills to think and learn for themselves. This is again consistent with the view that it is impossible to arrive at a single identifiable purpose for any form of education because of differing needs and expectations which may also overlap in many aspects (Doherty, 1997).

Consistent with stakeholder support for the transformative definition of quality, the survey found strong consensus from all stakeholders for equipping students with the ability to think for themselves and thereby empowering them as the ultimate purpose of HE. This objective is considered even more important than preparing students to meet the needs of employers which is consistent with Eagle and Brennan (2007) who stress that the most important goal of HE is to develop graduates with the ability to think critically and laterally. Learner transformation entails using innovative and interactive teaching and assessment methods in order to expose students to poorly defined and varied situations that require them to engage in effective actions in increasing uncertainty. There is strong support from both academic staff and employers for innovative teaching methods but other findings clearly indicate that

such methods may be strongly resisted by students, despite their strong agreement with the transformative objective.

Meeting employers' needs is ranked second and addresses a point of contention in the literature (Gibbs and Iacovidou, 2004) that as employers' needs are secondary to developing critical skills, preparing students for employment is not a valid purpose of HE. The interviews had also revealed that meeting employers' needs is very important, particularly for academic staff and students. The overall findings, therefore, show that although developing critical skills is most important, stakeholders do not believe that this objective is contrary to meeting employers' needs as responses of employers and academic staff both in the interviews and survey emphasise the development of students' personal and critical skills as necessary prerequisites for employment.

All three groups including students, surprisingly, attach the lowest importance to meeting students' needs and expectations among the four definitions, which is a positive indication that students do realise that their expectations may not be the best indicator of quality. Nevertheless, the fact that the majority of respondents in each group do agree that meeting students' needs and expectations is important, indicates acceptance of the concept of the student as a primary customer, partner and stakeholder in HE.

In essence, there should not be contradictions among the different objectives; for instance, equipping graduates with the skills for employment or to meet societal needs does not mean that they cannot think or learn for themselves or should lack critical skills. In fact, these objectives complement each other as do many other aspects of HE. Hence, rather than arguing which objective is more valid as some writers do, it would be more constructive for HEPs to consider how their programmes can develop the overall skills profile that would enable their graduates to meet the demands of the 'real' world, including employers and the society. As Wals and Jickling (2002) note, HEPs would have to provide sustainable and meaningful learning opportunities that will develop dynamic qualities in their students that will enable them to be critical and cope with poorly defined situations in different contexts.

#### 5.7.3 Academic Standards

The issue of academic standards is very closely linked with the concept of academic rigour and quality, as the standards set for a programme of study are a key determinant of the level of performance expected from students. High standards are supposed to be indicative of a high level of quality (Doherty, 1997) and, therefore, three statements in the questionnaire tested the views of the respondents on academic standards: the first on programme standards, the second on assessment standards and the third on the importance of maintaining standards versus students coping with their courses.

Stakeholder responses were not congruent across the three statements as a comparatively lower proportion of students value high standards. A significant proportion of the student sample do not appreciate or possibly understand the importance of maintaining or setting high standards, particularly assessment standards, which is consistent with the interviews. Instead, they place more importance on how easily they can cope with the demands of their study. This is consistent with the literature (Eagle and Brennan, 2007) that students may not really consider high academic standards as essential for career advancement or even representative of high quality. While students consider HE as essential for a career, they are indifferent to whether high standards are maintained (Rolfe, 2002) and as Carlson and Fleisher (2002) note, many now tend to shop around for the easiest courses with the highest grades. The lower importance attached to assessment standards by students may also reflect Mattick and Knight's (2007) observation that students are anxious about assessment performance and as a result do not consider or reflect on the quality of their learning approaches (and by implication standards) and focus only on their performance.

On the other hand, academic staff and employers greatly value programme standards and are firm in their view that academic standards cannot be sacrificed so as to keep students happy or to ensure that they can cope with the demands of the programme. The focus on student satisfaction and the concept of students as customers is often criticised by educators for falling academic standards (Gibbs and Iacovidou, 2004; Barnet, 1997). While this study found that it is important to understand students' needs and expectations and to motivate their interest, i.e. concepts which are associated with the student-customer concept, it also finds that maintaining high academic standards is considered vital in ensuring the quality of

HE by the key internal and external stakeholders. Clearly academic staff and employers do not consider a student oriented approach as inappropriate or incompatible with maintaining high academic standards.

However, although employers value high academic standards of programmes with 85.4% considering it extremely important, they are less enthusiastic about assessment standards with only 64.6% rating it as very/most important. Academic staff are more consistent across the three statements with the most emphasis on assessment standards. The differentiation by employers between assessment standards and programme standards is difficult to explain. Logically, high academic standards are measured and maintained by high assessment standards and accordingly responses of academic staff are consistent on both criteria. One reason for this differentiation could be that some employers feel that the whole process of HE should be of a high standard with appropriate checks and balances which should not be just reflected in the standards of assessments. Furthermore, responses to other relevant criteria show that employers do not really consider assessment performance as a good indicator of the quality of education; therefore their comparatively lower appreciation of high assessment standards may be just reflective of their overall perception of assessments. This is discussed further in the next section on assessments.

## 5.7.4 Quality of Intake and the Value Added View of Quality

Pursglove and Simpson (2007) indicate that admissions standards have a significant quality control role in influencing the quality of the overall educational outcomes as entrants with lower achievements at entry level are more likely to continue to perform poorly in HE. This study found that although academic staff are significantly more enthusiastic about strict admission policies and students the least, the overall ranking for this criterion is among the lowest by all groups. While students' responses may be more or less expected, it is more surprising that employers and academics rate admissions criterion quite low.

The quality of student intake is considered by many to be a necessary condition for institutional success (Cheng and Tam, 1997). However, although this study found that high academic standards of programmes is very important to academic staff and employers, high admission standards i.e. restricting student intake in terms of marks or grades at entry, is not considered a key determinant of quality, particularly by employers. Furthermore, although

stakeholders consider an institution's reputation to be an important indicator of the quality of its provision, apparently the majority do not feel that this reputation should be based on restricting admissions to the best possible student intake. Determining the aptitude of students for the programme of study and their attitude and commitment to HE are considered more important than a restrictive admissions policy in terms of high grades. As suggested by an academic staff in the interviews, conducting admissions interviews aimed at determining students' level of interest and commitment to learning can be a better measure of whether students are really interested in developing themselves through HE, rather than a restrictive admissions policy.

We should consider whether these findings are contrary to Morley and Aynsley (2007) who find that employers in the UK associate high academic standards with strict entry requirements. However, a restrictive admissions policy makes it easier to achieve higher standards even though the institution may have put in comparatively less effort in providing high quality teaching and learning experiences. It is easier for HEPs to demonstrate better outcomes with a higher quality student intake irrespective of the actual quality of the provision or effort by the institution. With a broader range of abilities at entry level, no doubt an institution will have to expend relatively more effort to ensure that their provison is adequate for attaining the desired standards. We should also consider whether the fact that HE in Oman is relatively new, is accountable in part for employers currently not equating high academic standards with strict entry requirements, contrary to the UK. Over time, with more graduates leaving the higher education system, there is the possibility that employers in Oman may form stronger perceptions about the educational provision (including admission policies) by the different HEPs and as a result may equate stricter admissions policies with higher standards.

The findings clearly showed that stakeholders make a distinction between high standards and the quality of the process of achieving such standards. The value added to the students' knowledge and skills profile through the inputs and teaching and learning opportunities provided by the HEP is considered to be a better indicator of quality by all three stakeholders. This view sees quality as a measure of whether the HEP has provided a satisfactory bridge between entrants and the intended outcomes by providing adequate opportunities that sufficiently enhance their knowledge and skills rather than based on the inherent ability of the students (Yorke, 1999). Irrespective of the quality of intake, what

would be more relevant is whether the curriculum, teaching, assessment, support and guidance are the most appropriate with respect to the student profile in order to achieve desired outcomes. This does not imply that the extent of the value added is not important as employers and academics also expect HEPs to maintain rigorous standards.

The value added view of quality corresponds with the admissions policy of most private HE institutions in Oman, where the students' performance in the pre-requisite Foundation year determines their entry to the degree programmes. A pass in the secondary school examinations would generally secure a place on the Foundation programme in most institutions and based on the extent of value added in terms of a satisfactory performance at the end of the foundation year, they can proceed on to the undergraduate programmes. Nevertheless, although the value added view of quality is probably a more accurate reflection of the quality of provision, it entails taking in a broader range of students, some of who may not be equipped to handle the rigour and demands of HE. This requires innovative approaches and greater commitment from both students and teaching staff to walk the extra mile in order to add value to the required level.

## 5.7.5 Core Transferable Skills for the Workplace

A very significant finding of this study is that employers as well as academic staff and students, although to a lesser extent, consider the development of core transferable skills as possibly more important than even subject specific knowledge and practical understanding of the subject. Transferable skills are very highly prized as these are skills which although learned in one area can be transferred into other areas; however, Yorke (1999) observes that a key difficulty is identifying transferable skills that are particularly relevant for employment. The transferable skills expected of graduates of undergraduate programmes as identified in the interviews include interpersonal skills, the ability to communicate effectively, team work skills, adaptability, self-management, creativity, application, problem solving and general awareness.

This shift from subject knowledge is emphasised in the literature and as Bourner (1998) states, the critical function of HE is no longer seen as imparting subject knowledge but the development of higher level transferable skills which include critical thinking and reflection. The findings are also consistent with Srikanthan and Dalrymple (2004) who find

that employers give low ratings to academic knowledge while skills such as problem solving, team interaction, communication etc., are highly valued. Given the comparatively lower emphasis by students on skills, it is also worthwhile to consider Peronne & Vicker's (2003) recommendation that in order to prepare graduates for the transition to the workplace, HEPs and academic staff would need to put in greater effort to enhance student awareness of key transferable skills. The development of skills is quite possibly more difficult to achieve than imparting an understanding of the subject and any effort to enhance students' skills profile must be adequately supported by appropriate emphasis in all the integral aspects of HE including curriculum, teaching and learning methods and assessments. As Struyven et al. (2002) note surface approaches to learning that focus on and assess memorisation of facts should be discouraged; rather a deeper approach to learning that focuses on how students use, evaluate and interpret information must be encouraged.

Three statements addressed the importance of workplace skills, the first one focused on the role of curriculum, the second on the role of the lecturer in developing such skills and the third on assessing skills. While there was congruence on the importance of the role of the lecturer in this context, there was a lack of congruence on the emphasis of the curriculum and assessments on skills, particularly because a comparatively lower proportion of students considered the latter as very important. This shows that students value workplace skills as long as the lecturer is responsible for developing such skills but many are not that enthusiastic. The findings indicate a gap between students on the one hand and employers and academics on the other, regarding the importance of emphasising skills in all aspects of HE if indeed learner transformation is to be achieved. This is consistent with the interviews which showed that despite their emphasis on being prepared for success at the workplace, students' interpretation of what this preparation entailed was less clear and in some cases included only practical understanding of the subject.

Despite the importance placed on core transferable skills in the literature, it must also be considered whether the importance placed by employers and academics in this study is the result of local/regional contexts, given the particular problems faced by HEPs in the region in inculcating such skills in students. In a study on students in one of the countries in the Arabian Gulf, Bridger (2007) observes that students have to be guided very carefully at tertiary level to make the transition from a tutor-centred approach to a student centred-one, where they take responsibility for their own learning. Students in Oman do face similar

problems due to the existing practices in the public secondary school system (Goodliffe, 2004). This coupled with the fact that Omani secondary school leavers have to rapidly adjust to learning in a second language in HE, makes the development of core transferable skills such as critical thinking and reflection, learner independence, team interaction, effective communication skills etc., quite difficult. As such skills are particularly lacking, the effectiveness with which an HEP is able to inculcate the core transferable skills can largely determine the perception about its graduates in the workplace.

#### 5.7.6 Curriculum

The importance attached to the curriculum dimension by academic staff and employers underlines its relevance in achieving the desired outcomes of HE. This is consistent with Nabi and Bagley (1998) who stress the critical impact of curriculum on almost all aspects of HE; curriculum can encourage or discourage the development of subject and practical knowledge, the development of core skills, the choice of teaching and learning methods and assessment strategies. However, students' responses indicate that they are not fully aware of the primary role of curriculum in influencing their overall experience and the outcomes of their education. Student responses, therefore, are responsible for the lack of congruence between responses on most of the curriculum criteria.

The highest importance is attached by all three stakeholders to curriculum focus on transferable skills, which is consistent with the findings in the previous section. The importance of incorporating skills development in the curriculum is supported by many researchers, for instance, Thomas (2007) emphasises the need for more flexible curricula with greater emphasis on the skills required for lifelong learning, as a result of economic and demographic changes. The curriculum can also encourage the adoption of active teaching and learning methodologies by tutors and students which is fundamental in developing reflective abilities and critical thinking skills. The comparatively lower emphasis on skills by students maybe consistent with Dillon and Hodgkinson (2000) who find that some students find it irritating and demotivating when they are repeatedly faced with similar advice that focus on skills development. It is also consistent with students' reluctance to engage in interactive learning methods and non-traditional assessment methods as indicated in other sections.

An important value of this study is to provide an understanding of how subject knowledge is viewed by employers and academic staff. The study found significant differences between academic staff on one hand and employers and students on the other regarding the focus of the curriculum on subject knowledge. Strongly consistent with the interview findings, the survey indicates that while many academics rate curriculum focus on the subject and coverage as very important, employers do not share the same emphasis on subject knowledge. As employers revealed in the interviews, they felt that undergraduate programmes should focus on in-depth knowledge of a more narrow range of core fundamental topics rather than a broad range of topics. Employers and even some academics expressed their concern that an extensive coverage of subject topics may often result in the majority of students not really having an in-depth understanding of most topics. It was noted that with its focus on an extensive range of topics HE seem to be producing undergraduates who are not really fit for junior or middle level jobs for which they are recruited, nor for the higher level jobs which require considerable experience. Hence, consistent with Baruch and Leeming (1996), employers consider that more focused in-depth knowledge of the core and fundamental subject concepts may be needed in the early stages of a graduate's working career. This must be accompanied by an awareness of how to adapt theoretical understanding to practical work situations which was also valued very highly by both employers and academics but comparatively lower by students.

Middlehurst (2001) contends that a critical issue in effective curriculum development is whether curriculum developers are able to determine if the curriculum meets the ever changing needs of both students and potential employers. Otherwise, the content will not be fit for the purpose for which it is designed, either for the direct customer (student) or the indirect purchaser (state or employer). HEPs in Oman may consider it productive to evaluate whether the curriculum does provide adequate in-depth and focused subject knowledge required at this level or whether the curriculum is too ambitious in that it ultimately provides only a superficial knowledge of too many topics without sufficient emphasis on core skills. Kember (1997) notes that curriculum can influence tutors to focus on the subject matter rather than on the development of critical thinking. The challenge for HEPs lies in designing curriculum that provides a balance between subject content and context with adequate scope for students to enhance practical skills and a critical mind-set. Ultimately, curriculum should be also built around the needs and aspirations of the learners so that they are adequately motivated in order to become deep learners.

## 5.7.7 Teaching and Learning

The very high importance attached to the teaching and learning dimension in this study is consistent with the increasing focus in the recent literature on the resurgence of the importance attributed to the teaching function. Consistent with the interviews, faculty related factors and teaching methods have been given high importance by the three stakeholder groups. The importance attributed by the three stakeholders to the lecturers' teaching methods and styles and their ability to communicate and explain topics, reinforces the critical role of academic staff and the teaching function in enhancing the quality of HE. However, although factors such as accessibility of explanation and the lecturers' ability to stimulate thinking and behavioural skills are highly valued by all three stakeholders, a comparatively lower proportion of students attach the highest importance to these factors resulting in a significant lack of congruence. This reveals less emphasis by students on aspects which require them to understand or to undergo transformation.

The literature indicates that the focus on learner-centred methods has resulted in the critical role of academics being generally underplayed (Bradley, 1999). Lomas (2004) highlights the need to reward and recognise good quality teaching in an environment that does not adequately emphasise the teaching function. The ability of teaching staff to stimulate students' interest in the subject and to motivate them to participate in the learning process is considered extremely important by all groups. This is consistent with Anderson (2000) who stresses that a crucial factor in the complex interactions in HE includes students' engagement with the subject which is influenced in part by the enthusiasm and skill of the lecturer. Pennington and O'Neil (1994) also emphasise that the lecturer's personal commitment and enthusiasm for the subject is vital in shaping students' interest and thereby deep approaches to learning, which in this highly tutor-centred context is highly relevant to the development of critical skills. What is generally not sufficiently considered is that student-centred teaching and the complexities inherent in mass education require tutors to adopt more effective and innovative teaching and learning methods and assessment strategies.

Reflecting the emphasis on transferable skills, the survey finds that stakeholders greatly value the ability of teaching staff to be transformative i.e. to stimulate thinking and develop

behavioural skills in their students. Employers, particularly, value tutors' focus on skills and behavioural development more than most of the other teaching and learning criteria. This is supported by Hawawini (2005) who observes that employers of business graduates are increasingly demanding behavioral and societal skills and is also consistent with Stefani (2005) who argues that the traditional role of tutors in the transmission of knowledge is no longer adequate as today's knowledge economy requires tutors to develop a different skill set. However, the lack of congruence on the tutors' ability to stimulate thought and behavioural development showed that a comparatively smaller proportion of students attached the same level of importance. In comparison, a higher proportion of students consider development of workplace skills as important, which indicates that many students do not relate critical thinking and behavioural skills to the workplace. This is consistent with the findings in the interviews that students are not aware of what transferable or workplace skills entail.

However, the interviews also found that while the development of transferable skills was seen as vital by academic staff, there was less clarity on their own role in developing such skills. Academic staff were not really aware of what was required from them in terms of effective skills and personal development and seemed to perceive their role as that of imparting subject knowledge. Therefore, while the survey found that lecturers consider their ability to develop skills as very important, the interview findings indicate that lecturers require further guidance on how this can be achieved. The survey also found that while academic staff do acknowledge that they have a key role to play in inculcating core skills, they nevertheless consider that the curriculum plays a stronger role in this process in comparison to themselves. Hence, it appears that a degree of confusion does exist among academics on their own role in developing skills and whether the inclusion of skills development in the curriculum will naturally allow such skills to be developed without an equal focus by teaching staff. As the development of core skills would require extensive input from teaching staff, especially in the current context which is heavily tutor-centred, any lack of awareness on the part of academic staff on their role will have a negative influence on the ultimate outcome of HE.

Hill (1995) stresses the critical role that personal interaction between student and staff has in enhancing student motivation and learning which is exemplified through tutors' sensitivity to student needs. Accordingly, the survey found that empathy with students'

needs is valued by both students and lecturers as an important aspect of teaching and learning. Employers however, do not rate this as very critical. This may be due to the fact that employers are not directly involved in the process of teaching and learning and hence may not appreciate the more subtle factors. Employers may also consider that understanding individual student needs is not that important as they attach more value on the HE process being challenging enough to mould students' character and attitude, as revealed in the interviews. Surprisingly, the survey also finds that academic staff place comparatively less importance on their own experience in terms of number of years. The minimum experience of the academic staff in the sample is 6 years and, therefore, their lack of appreciation for teaching experience may suggest that they feel that experience by itself does not add much value.

A main finding of this study is that although students attach the highest importance to instructor related factors, they are not very enthusiastic about their own role in the teaching and learning process. This is exemplified by the lower importance they attach to all criteria that involve student participation, including reading and understanding the subject material, interactive teaching methods and contribution of other students to their learning. This strongly reflects a tendency on the part of many students to rely solely on lecturers and tutor-centred teaching methods. Many students also do not place much value on the positive impact that other students can have on their own teaching and learning experience, while this is considered important by both employers and academics.

The lack of appreciation for interactive student-centred teaching methods by students reflects the observations by academic staff in the interviews that many students resent student-centred and interactive methods of teaching, preferring to rely on lecturers for all the input. Considering that most students prefer traditional teacher-centred teaching methods to more innovative methods, together with the high importance they attach to teaching methods, we could conclude that many students perceive teacher-centred teaching methods to have the most positive impact on their learning. This view is not shared by most academics and employers who value interactive teaching methods, possibly to ensure students' participation in the teaching and learning process and stimulate and enhance skills. Hence, HEPs do not have the hard task of convincing their staff about the value of using innovative interactive teaching methods; instead they should educate students about the value of such methods and thus gradually overcome their resistance. Staff development is

also essential in order to develop awareness of practical ways to reduce student resistance and enhance motivation and participation.

Hill et al (2003) highlight the importance of support systems in student transition from school to HE and they find that students greatly value such support systems. However, although additional academic support systems are considered important by the majority of the respondents, it is ranked comparatively lower on the list of teaching and learning criteria. It must be noted that this criterion was not overtly emphasised in the interviews as well.

While the majority of respondents believe that student feedback must be an important focus of quality management and that students can evaluate good teaching, the proportion of academics who believe that students can indeed evaluate good teaching is lower than the other two groups. This is consistent with Harvey & Langley (1995) who write that academics believe that students are not the best judge of the education they receive, particularly in terms of identifying the knowledge and skills necessary for their future careers. There may be some truth in this particularly if teaching staff have the difficult job of weaning students away from traditional tutor-centred teaching approaches to a student-centred one by introducing non-traditional teaching methods that are strongly resisted by students. As the interviews revealed, academic staff do feel that students may resent tutors who attempt to do so and student feedback may be coloured by this lack of appreciation.

#### 5.7.8 Assessments

Consistent with the interviews, employers in particular, rank assessments and outcomes the least important among all the six dimensions of quality. While employers share the concern of academic staff for programme standards and the process and curriculum aspects, they do not agree to the same extent as academics on the importance of assessments. This can be considered as evidence that employers share the same doubts as Barnett (1994) as to whether the complex nature of HE can be easily and accurately reduced to a set of easily measurable competencies.

The results also indicate that high grades in assessments are not considered a reliable or important indicator of quality in HE by all three stakeholder groups. This must be

considered together with employers' emphasis in the interviews, that although the process of HE should be challenging enough to mould and develop students' skills and knowledge, they would not necessarily trust higher grades as indicative of better skills. This indicates that employers and to an extent academics do not really trust assessment strategies to induce the transformative process in students or to be a valid measure of the extent of that transformation. One may conclude that employers share Trow's (1996) opinion that attempts to measure educational outcomes are spurious as education is more of a process rather than an outcome. However, many organisations do rely on grades in the first instance to screen applicants, as revealed in the interviews. As Schray (2006) observes, although the most important evidence of quality in HE is considered to be performance, especially student achievement of learning outcomes, this must in fact be represented by a broad range of performance indicators including access, student learning, degree completion, and economic returns and not merely grades.

The survey findings are also consistent with comments of academic staff in the interviews that they place more value on the assessment process in influencing the quality of HE rather than high grades. However, while academics share the view that good grades in assessments are not a very important indicator of quality, they attach considerable importance to the role of assessments and assessment standards in influencing the overall quality of HE. This reveals a lack of consistency in their response, because if academic staff really believe that assessments and assessment standards are important and integral to learning and can affect student transformation, then they should in effect consider assessment performance to be very important. This lack of consistency implies that while academics acknowledge that assessments are a vital factor in driving HE quality, they also acknowledge, like employers, that current assessment strategies are not appropriate in terms of measuring important competencies. Employers and academics may also share Gibbs and Simpson (2005) belief that if the assessment strategy is not well designed or comprehensive it may become comparatively easier for students to provide the teacher with whatever s/he expects without really going through a transformative process or enhancing their capabilities. This is particularly true when we consider academics' views in the interviews that assessments may not really be assessing critical or key outcomes or that it may be easier to obtain higher grades in some institutions compared to others as assessment standards may not be consistent across institutions. In comparison, the fact that students rank assessment performance higher than the other two groups may be consistent with students' comments in the interviews that they feel good when they perform well in assessments. This can exert its own influence on student motivation.

Research indicates that the quality of student learning is higher in non-traditional and assignment-based courses rather than exam-based courses (Gibbs and Simpson, 2005). As students in this study prefer traditional tutor-centred teaching methods, it is logical to assume that they would prefer traditional and more prescribed assessment methods as well. This is indeed supported by the findings, as the large proportion of students consider examinations to be the best assessment method which is consistent with McHardy and Allen (2000) who find that students prefer and are more comfortable with prescriptive and traditional methods of learning and assessment. In comparison, it is found that employers do not consider examinations to be an appropriate test of students' skills and knowledge. Academics are comparatively more undecided on this aspect, although almost half of the sample disagrees with the appropriateness of exams. Another reason for this ambiguity could involve the concern expressed by a few academics in the interviews that some forms of assessments such as take-away assignments can be misused by the use of unfair means. Further analysis of academic responses by subject shows that while IT lecturers consider examinations to be the best assessment method, Business lecturers do not. This may indicate that examinations are more appropriate for the IT discipline or may also reflect the lack of skills focus by IT lecturers. Consistent with this response, academics (mainly IT lecturers) place slightly more importance on assessing subject knowledge, while employers and students attach more importance on assessing skills for the workplace. This is consistent with Eagle and Brennan (2007) who emphasise that assessing graduates for entry to the workplace requires much more than the passive reproduction of subject knowledge.

It is also found that the implementation of a continuous assessment strategy is supported by all stakeholders so as to provide rigour and challenge to the process of HE. The endorsement of continuous assessments by the majority of students is surprising, considering the workload that this entails. Feedback on assessments is an integral part of an effective assessment strategy and findings show that academics and employers clearly realise this; but students do not. Many student interviewees did not seem aware of the practice of tutors providing feedback on assessments. The relationship between feedback by tutors and the enhancement of students' capacity for critical reflection is highlighted by Harvey & Knight (1996). The significantly lower importance given to this criterion by

students indicates that some of the institutions in the study may not have assessment feedback systems for students to appreciate the value of feedback or that the feedback given may not be considered constructive or useful by the students. The challenge is to provide opportunities for students to receive regular constructive informal and formal feedback on their performance. Students focus on parts of the curriculum for which they are rewarded (Gordon, 2003), so if students receive constructive feedback on their skills and subject understanding they will be motivated to spend more time and effort in developing them.

#### 5.7.9 Resources

Students attach more importance to the resources dimension in comparison to employers and academic staff. Among the different resources, the highest importance is given to the range and quantity of library resources and quality of teaching facilities by students and academics, which is not very surprising. Students to staff ratio is valued by the majority of academic staff indicating the difficulties of managing larger student numbers and what will essentially be mixed ability students in the same class due to the diversity of student intake. This is consistent with the interviews, as larger lecture sessions were not seen as very conducive to learning, given the difficulties of learning in a second language. Closer personal interaction between students and staff and small class sizes may be required if tutors are to influence students personal and skills development. Sports and recreation facilities although ranked lower are also valued by the majority of respondents, particularly students which is not surprising as well as employers. Employers' responses are consistent with their view on the overall development of the student not only through teaching and learning but through recreational activities which provide opportunities for personal development.

# 5.7.10 Students as Co-Participants

Student transformation requires a very active and joint participation between students and HEPs (Hill, 1995). Harvey and Knight (1996) note that students' analytical and critical skills can be developed only by the joint participation of students and academic staff. The concept of students as co-producers of their own learning requires them to participate to the fullest possible extent, or the learning outcomes or objectives may not be met satisfactorily. Accordingly, both employers and academic staff in this study consider the attitude and

commitment of students and their participation in the teaching and learning process as extremely relevant to quality. Their overall response to student related criteria indicate that while it is very important for HEPs to focus on students and facilitate learning to the fullest possible extent, this does preclude students from taking responsibility for their own performance and learning.

However, a significant finding of the survey is the confirmation that majority of students have no awareness of the importance of the concept of student participation in the learning process. It was found that students clearly expect a lot from HEPs and their tutors with regard to their academic success, although they attach considerably lower importance to all factors that involve their input and participation. Students' responses consistently placed lower importance on all factors that involve their input while they attach the highest importance to what can be provided for them by tutors and the HEP. They are less emphatic about academic standards, student participation and their own contribution to learning inside and outside the classroom, although they value the skills and knowledge that will prepare them for the workplace. Although they share the same view as employers and academics that HE is about equipping students with the skills to think for themselves, they do not seem aware of their own role in this process. There is a definite lack of awareness among students that in order to learn effectively and develop relevant skills and knowledge, the process of learning has to be equally comprehensive and challenging and requires their commitment and participation. These findings are consistent with Telford and Masson (2005) who also find that students did not consider their own commitment to learning as important

Student responses on the question of institutional responsibility versus that of students' also show that almost half of the sample consider that it is the institution's rather than their own responsibility to ensure their academic success. This also confirms the interview findings that many students tend to expect tutors to impart knowledge and thereby ensure their success, without realising the importance of their own contribution towards the process if learning is to be indeed successful in the true sense of the word. The findings clearly support Mattick and Knight's (2007) view that undergraduate students may not be prepared for a learning environment which involves self-directed participation and that they may find the process of self-directed learning and participation daunting. In order to overcome such trepidation and encourage student participation, tutors and academic advisers should

provide a supportive learning environment with adequate guidance on self-directed learning and how its success could be evaluated. Hill (1995) makes the connection between student expectations and their prior educational experiences and Rolfe (2002) finds that students tend to adopt passive learning approaches and expect all information to be provided to them as a result of their secondary school experiences and time and societal constraints. As undergraduate students do not have any comparative frame of reference with regards to expectations in HE other than that of their schooling system, they may have unrealistic expectations, which is clearly the case of students in Oman who come from traditional tutorcentred schooling environments. Enhancing student participation will take careful management by HEPs because the interviews revealed that academic staff did not really have clear ideas of how they could motivate and support students, the majority of whom are comfortable only with traditional teaching and assessment methods, together with the difficulties of learning in a second language. These findings also lead us to consider the concerns raised by Clayson and Haley (2005) that students as fee paying customers will take less responsibility for their own learning and will place the responsibility for their failure or poor performance on the HEP or more precisely their tutors. It would be interesting to find out if this attitude is shared by students in government owned institutions that do not charge fees.

#### 5.7.11 Differences between Academic Staff

Significant differences in the responses of the two groups of academic staff (IT and Business) indicate that Business lecturers are more aware of the changing facets of HE in terms of responsiveness of programmes to the job market, practical relevance of education, transformative learning and the importance of transferable skills. Possibly as a result, Business lecturers are also more emphatic about the importance of faculty related factors such as understanding student needs, teaching methods and teaching styles and accessibility of explanation of subject related concepts. Quite significantly, Business lecturers place more importance on core transferable skills than IT lecturers, a fact which was also brought out in the interviews. Therefore, they are also more open to the use of interactive teaching methods and place more stress on the importance of students' own contribution to learning inside and outside the classroom.

IT lecturers account for the ambiguity present in the responses of academic staff on certain criteria as a comparatively smaller proportion attach high importance to academic standards, interactive teaching methods and student responsibility while a larger proportion value traditional assessment methods and meeting student expectations. A comparatively smaller proportion disagree that examinations are the most appropriate assessment method which is also consistent with the lower emphasis by IT lecturers on transferable skills, learner transformation and interactive teaching methods. Morley and Aynsley (2007) find that what constituted quality in HE differed according to the sector with scientific and technical employers placing greater emphasis on graduates' subject knowledge. Although Morley and Aynsley was referring to employers, it is plausible that lecturers of more technical subjects such as IT may also emphasise subject knowledge more than lecturers of less technical subjects such as Business. However, in the interviews employers were emphatic that IT graduates should possess better soft and transferable skills, and not just subject knowledge. A smaller proportion of IT staff also agree that the purpose of HE is to equip students with the skills required to think for themselves, which was ranked as the most important objective by all stakeholder groups, thus highlighting significant differences between IT lecturers and the other stakeholder groups. Even more surprisingly, a larger proportion of IT lecturers in comparison to Business lecturers consider that it is more important for HEPs to provide education and services that meets the needs and expectations of students and also place more responsibility on the institution rather than the student in ensuring students' success.

While demonstrating more empathy with students by emphasising the importance of understanding student needs, accessibility of explanation and other faculty related criteria, Business lecturers are nevertheless more emphatic about maintaining standards and not merely meeting student expectations. They, therefore, emphasise the importance of equipping students with the skills to think and learn for themselves as the most important purpose of HE. There is thus more congruence between the views of employers and Business lecturers, while IT lecturers are more congruent towards students' views in certain aspects. The latter is not a positive indication as the congruencies involve key criteria such as high academic standards, student effort and participation and the development of core transferable skills; hence, HEPs will have to ensure that IT lecturers understand the importance of these factors if they are to provide a more meaningful teaching and learning experience to their students.

# **5.8 CONCLUSION**

In order to answer the second research question, this chapter presented the congruencies and differences in the perceptions of the three stakeholder groups on the quality criteria identified from the interviews and the ensuing implications for HEPs. Further measures to be taken by HEPs are discussed in the next chapter which presents the framework of the stakeholder determinants of quality in HE.

#### **CHAPTER SIX**

# FRAMEWORK OF STAKEHOLDER DETERMINANTS OF QUALITY IN HIGHER EDUCATION

The third research question sought to propose a framework for managing quality based on the stakeholder determinants of quality (SDQ). This study reveals that although there are significant differences among stakeholders on the determinants of quality in HE, there are sufficient similarities among the majority of stakeholders to facilitate a framework for managing quality that addresses the management of the differences and builds on the congruencies. This chapter discusses the proposed framework for managing quality in HE for Business and IT undergraduate programmes based on the key SDQ identified from this study.

# 6.1 SDQ FRAMEWORK

The proposed framework for managing quality set out in Table 20 contains 28 key criteria under the six dimensions of HE for continuous monitoring and enhancement. It is intended as a framework for institutional self-assessment that can be applied by HEPs to support their continuous quality improvement efforts so that quality becomes a way of life rather than a distinct or separate aspect of the institution's activities. The framework identifies the key areas that must be addressed by HEPs, however, it does not suggest any processes and procedures that may be used in order to monitor and enhance these key criteria.

All the criteria identified in the framework are important in managing the quality of the provision; however, it is useful to note that each criterion may require different levels of attention, management and monitoring. Criteria which are congruent between the 3 groups or between 2 groups and which are considered most important by the majority of respondents in those groups are indicated as having 'strong congruence'. These criteria are extremely important and require constant monitoring and enhancement. Where a significant lack of congruence between the stakeholders was identified, this is indicated as a 'gap' in the framework which requires very careful and appropriate management by HEPs, particularly if the gap is on the part of students or academic staff. Those criteria on which there is congruence but has comparatively lower emphasis by respondents is indicated as

'congruence' and may require comparatively less attention; however this does not mean that they can be ignored.

Table 20: Framework of Stakeholder determinants of quality in HE (SDQ)

Quality	Quality Criteria	Congruence
Dimension		
	Ascertaining students'	Gap: Students unaware of their role as
Admission	commitment and attitude to	co-participants in the teaching and
Factors	learning at point of entry	learning process.
	Provision of a range of	Gap: Academics are not as concerned
	programmes on offer in	about programme variety as students
	accordance with industry and	and employers.
	societal requirements	
	Academic standards of	Strong congruence between
Institutional	programmes must be rigorous	employers and academic staff.
Factors	and challenging in order to	Gap: Students unaware of the
	ensure students' overall	importance of maintaining standards or
	development and transformation	the role of standards in their own
		development.
	Enhance institutional reputation	Strong congruence
	and industry links	
	Apply and monitor attendance	Congruence
	requirements	
	Balance curriculum focus	Strong congruence
Curriculum	between subject knowledge and	
	core transferable skills.	
	Focused rather than extensive	Gap: Academics emphasise wide
	subject knowledge	coverage of subject. They should aim
		for in-depth focused knowledge of key
		subject areas in the first instance.
		Coverage of further areas must not be
		at the expense of core fundamental
		subject knowledge and application of
		that knowledge which is greatly emphasised by employers.
	Emphasis on developing core	Strong congruence between
	generic skills for the work place	employers and academic staff.
	generic skills for the work place	Chiproyers and academic stair.

		Care Mara amenhacia an transferable
		Gap: More emphasis on transferable
		skills by IT curriculum is required.
		Gap: Students lack adequate awareness
		what skills for the workplace really
		entails.
	Practical components in the	Strong congruence between
	curriculum	employers and academic staff.
		Gap: Lesser emphasis by students as
		they may lack adequate awareness of
		the role of curriculum.
	Ensure relevance of the	Strong congruence between
	programme to the industry	employers and academic staff.
		Gap: Lesser emphasis by students as
		they may lack adequate awareness of
		the role of curriculum.
	Quality of teaching facilities	Strong congruence
Learning	Range and quality of library	Strong congruence
Resources	resources	
	Staff: student ratio	Congruence
	Sports and recreation facilities	Congruence
	Tutors' ability to provide	Strong congruence between
Teaching &	accessible explanation of subject	employers and academic staff.
Learning	concepts	Gap: Lesser emphasis by students.
8	Tutors using interesting teaching	Strong congruence
	methods and teaching styles	However, students' definition of
	including providing interactive	interesting teaching methods may differ
	student-centred learning	from that of the HEP/employers'.
	opportunities.	Students are also unaware of the
		importance of new and innovative
		learning experiences in enhancing their
		skills and learning. Students must be
		made aware of the benefits of particular
		methods and adequately supported
		during the process of familiarisation
		with non-traditional teaching methods.
	Tutors' role in motivating	Strong congruence
	Tracord role in indivading	chi drig corigi derice
	students' interest in the subject	

	Tutors' role in developing core	Strong congruence
	transferable skills for the	However, students may be unaware of
	workplace	what workplace skills really entail.
		Academics require considerable support
		and guidance in fulfilling this role.
	Tutors' role in stimulating	Strong congruence between
	thought and behavioural and	employers and academic staff.
	personal skills	Gap: Many students unaware of the
		relationship between behavioural skills
		and critical thinking and skills for the
		workplace.
		Academics require considerable support
		and guidance in fulfilling this role.
	Tutors' empathy towards	Strong congruence between
	individual student needs	academics and students.
	Tutors' skills of organisation and	Strong congruence between
	communication of subject of	employers and academic staff.
	teaching staff	Students do not attach the same level
		of importance.
	Efforts put in by students outside	Strong congruence between
	the classroom to read and	employers and academic staff.
	understand the subject material	Gap: Students unaware of their role as
		co-participants in the learning process.
	Enhance contribution of all	Congruence between employers and
	students to overall learning	academic staff.
	experience inside & outside the	Gap: Students unaware of their role as
	classroom	co-participants.
	Additional academic support	Emphasised by academic staff. However
	systems to enhance participation	the importance of such support systems
	of students in the learning	is not recognised by students essentially
	process and develop student	as they are unaware of their role as co-
	awareness	participants.
	Variety of assessment methods	Strong Congruence
Assessments		However, many students and IT
		academics tend to prefer traditional
		assessment methods such as
		examinations.

Emphasis of assessments on	Strong Congruence between academic
skills and not just subject	staff and employers.
knowledge	Gap: Students place less emphasis on
	this criterion. Moreover, employers and
	academics do not believe that
	assessments are currently a good
	indicator of students' critical/overall
	skills and this must be addressed.
Feedback provided to students	Strong Congruence between academic
	staff and employers.
	Gap: Students unaware of the
	importance of feedback to their learning
Use of continuous assessment	Congruence
strategy	

In accordance with Becket and Brooke (2005) who identify the requirements of an analytical quality framework, the following elements have been considered in proposing the framework:

- 1. The degree to which inputs, processes and outputs in HE are assessed: The SDQ comprise key criteria which fall under the input, process and output dimensions of HE. As all the criteria exert differing but considerable influence on the overall learning experience and ultimately the outcomes of HE, it is important to ensure a synergistic balance between all dimensions i.e. Admissions Factors, Resources, Curriculum, Institutional factors, Teaching and Learning and Assessment.
- 2. The degree to which different stakeholder perspectives are considered: the strength of the proposed framework is that it is based on the key determinants of quality as perceived by the three main stakeholders. Where there are gaps or differences in perceptions among the three groups it entails managing the differing expectations and deficiencies.
- **3.** The extent to which different quality dimensions are considered: All the key dimensions of quality in HE as identified in the literature are covered by the framework.

Academic support systems such as registration and finance are not included, as they were not identified by the stakeholders as having a direct impact on the quality of HE.

**4.** The balance of quality assurance versus quality enhancement practices and processes: An essential requirement of the proposed framework is that managing quality is a normal, integral and continuous aspect of all activities within HEPs. This necessarily involves quality assurance and enhancement as integral aspects of quality management and should not be seen as disparate functions. Quality thus becomes the responsibility of every member of staff, who must be empowered and supported so that they are able to monitor and enhance their own contribution towards addressing the SDQ and the objectives of the programme/institution.

In order to effectively apply the framework, the following areas must be addressed carefully by HEPs:

# 6.2 STUDENT INSTITUTIONAL SOCIALISATION

If students are to be effective co-participants in HE, they must develop adequate awareness and appreciation of institutional values and what is expected of them (Hill, 1995). The process of developing congruence between the quality values of students and those of academic staff is akin to the service sector equivalent known as 'customer organisational socialization' (Telford and Masson, 2005, p.108). Similar to service customers, students must be supported in acquiring the skills and knowledge required to interact effectively with the HEP and its staff so as to achieve planned outcomes. This process can be termed 'Student Institutional Socialisation' and should involve the following aspects:

# **6.2.1** Managing Student Expectations

The framework clearly identifies several gaps in students' awareness particularly their role as co-participants in HE. Closely linked to student participation are the expectations that students have of their tutors, HEPs and HE in general. Hence, an essential first step in enhancing student participation is for HEPs to effectively manage student expectations which may include those that are realistic and unrealistic. It is inevitable that students' perceptions of the quality of their experience will depend on their values and expectations which will in turn influence their involvement and commitment to the teaching and learning

process. However, while the importance of student expectations is acknowledged in the literature, there is not much evidence to show how they can be effectively managed.

.

Managing student expectations would require HEPs to identify and address even those expectations that it may not be possible to satisfy. As this study revealed, some students have unrealistic expectations in that they expect HEPs to be responsible for their success rather than accepting their own responsibility. It is clear that in many cases students' quality values are based on a view of HE in which their contribution is minimal. Despite the importance given to the concept of students as partial employees or co-producers in the process of HE, students clearly do not share in this understanding. However, the interviews did reveal that students can be made aware of the importance of their own contribution to learning and that it is possible for HEPs to create appropriate opportunities to raise students' awareness of the importance of participation and the ensuing benefits in relation to their future success. In particular, students must understand that teaching can facilitate learning but does not cause it.

While most HEPs tend to address issues such as teaching and learning strategies and student responsibilities during induction or orientation sessions, it is suggested that much more is required in order to deal effectively with this issue. Students must be informed of what is and what is not possible together with a rationale for the distinction. This may include creating opportunities for peer learning, more intensive student and staff interactions with a greater facilitative role on the part of the instructor. In doing so, HEPs must be aware that students tend to be strongly influenced by the perceptions of other students and negative word of mouth has the potential to compound problems. As recommended by Hill et al (1996) such dialogue between HEPs and students should encourage students to reflect on past learning experiences so that they can build on these experiences and thereby develop new skills, while at the same time allow negative approaches to be identified. It is also important to note that student expectations may change over time and, therefore, must be closely monitored. The interviews revealed that students were influenced by their peers and word of mouth communications. Hence, as suggested by Hill (1996) HEPs would do well to use their existing students in their communications with prospective students; however, they must first ensure that these students share the HEPs' values through a process of mutual dialogue.

#### **6.2.2** Student Motivation

Irrespective of the quality of curriculum, learning opportunities and resources provided by an HEP, key factors that determine student learning relates to their motivation and commitment. The study revealed that student motivation and interest in the subject depends to a large extent on the lecturer. However, many lecturers may tend to focus on the learning process rather than more aesthetic qualities such as enjoyment with the lesson and socialisation, in addition to developing empathy with students' needs. The enthusiasm and skill of the teaching staff are very important in transmitting a sense of enthusiasm for the subject matter which is instrumental in student motivation. This requires teaching staff to become reflective practitioners and ensure their skills of communication, presentation and organisation of class contact sessions are as effective as possible. Opportunities for student interaction must be introduced so that participants can develop new skills within a HE ambience and the friendly atmosphere of the classroom.

# 6.3 LEARNER TRANSFORMATION

Transformation of learners and the development of core transferable skills for the workplace have been identified as the primary objective of HE. Transferable skills that were identified as most critical for graduates of undergraduate programmes in Business and IT disciplines are a combination of personal behavioural skills, generic transferable skills, general awareness and focused subject specific and practical knowledge. Specifically, they include the following:

- Willingness to learn and learning skills
- Adaptability and flexibility
- Positive attitude to work and work ethics
- Self and time management skills
- Effective skills of communication and interaction
- Ability to think critically and solve problems
- Ability to work effectively in teams
- Good grasp of fundamental core subject related topics and their application
- Knowledge of practical components
- Creativity

While HE in the past was more concerned with the transmission of knowledge, it must now make the transition to developing a portfolio of skills and knowledge that will enable graduates to succeed after graduation and enable them to make use of the information explosion that marks today's society. In order to facilitate student engagement with the learning process and the meaning of the subject, HEPs would do well to provide a coherent and consistent approach to learner transformation. This process should also involve employers who can provide an external perspective to students through industry links and workshops. As highlighted by Srikanthan and Dalrymple (2007) learner transformation requires curriculum aimed at transformation, teaching for transformation and assessment for transformation.

# 6.3.1 Curriculum focus on core transferable skills

The role of curriculum in encouraging teaching staff to focus on the required portfolio of skills and subject matter cannot be over emphasised. It might also be appropriate for curriculum developers to assess whether the curriculum incorporates an unnecessarily wide range of subject content that may not be really be necessary for the workplace and, therefore, sets unrealistic goals for most students. This should not be interpreted as a sign of lowering academic standards; rather the goal of undergraduate education should be to produce graduates who have adequate core subject knowledge together with the higher level cognitive and transferable skills such as application, analysis, problem solving, communication and learning.

# 6.3.2 Balance between subject and skills by teaching staff

Srikanthan and Dalrymple (2007) recommend a fundamental paradigm shift in the notion of teaching as a routine and subsidiary task to a key performance indicator. Moreover, if the development of skills is to be effective, teaching staff will also have to make the transition from subject focus to skills focus in their teaching. They will essentially drive the development of core skills as most undergraduate Omani students will not have the necessary skills to make this transition on their own as their previous experience in school is largely tutor-centred. This entails motivating reluctant students to move from passive learning to active learning, which requires providing and supporting them with the tools required to cope with such transition. As revealed by this study, academic staff may not be

comfortable or familiar with their role in developing core transferable skills although they fully agree with their importance. In order to facilitate learner transformation, tutors would have to stimulate thinking and cross over to the areas of personal and behavioural development. This may be a strong culture shift for many staff who see their role as imparting subject knowledge and research, not students' personal and skills development. Academic staff may not be properly equipped to handle this transition; hence, HEPs will have to provide appropriate opportunities to develop their awareness of the methods and approaches which can contribute to skills development and student transformation. The emphasis should be on developing a proactive institutional culture where all staff engage in supporting student transformation. This then requires team work and creative approaches at tutor level, institutional level and also national level by redefining objectives and adopting innovative teaching and learning strategies aimed at enhancing students' engagement with the learning process.

# 6.3.3 Balance between subject and skills focus in assessments

Assessment of student learning can be the most significant factor affecting transformation (Srikanthan and Dalrymple, 2007). The study revealed that employers and to a lesser extent academics do not consider assessments an appropriate measure of core competencies. This indicates that assessment strategies must evaluate skills and overall abilities rather than focus only on subject understanding. Incorporating a variety of assessment methods that is appropriate for developing and measuring core transferable skills and which will encourage students to become active rather than passive learners is vital, however, students require careful support in managing the time demands involved.

#### 6.4 MANAGEMENT COMMITMENT

In order to build commitment to quality both academics and management must fully appreciate the reasons why managing quality is imperative. Ultimately, quality standards will provide the main competitive edge and ensure the longer-term survival of an institution. The reality may be that when HEPs are funded by the government, the criticality of achieving this competitive edge may be diminished to an extent. On the other hand, when HEPs are privately funded, economic considerations and shorter-term objectives of meeting student expectations may overrule the longer term objectives of meeting broader

stakeholder expectations. Nevertheless in the long run, both types of institutions would have to reconcile their longer and shorter term aspirations in order to remain successful.

No quality model will be effective or can be successfully implemented unless there is adequate management commitment to quality and an adequate understanding of the complexities and subtleties of the concept of quality. The success of a quality strategy depends on the acceptance and involvement of everybody involved in its implementation and this requires a conducive organisational culture. As Senge (2000) observes, HE institutions are very complex organisations where knowledge is fragmented into specialised areas and educators are engaged in a highly individual activity of teaching. Bringing about changes in such a complex system requires commitment and acceptance of a holistic, integrated approach to quality so that it permeates throughout the institution and becomes everyone's responsibility. Management aspects include adequate attention to the appropriateness and interdependence of leadership, strategies and internal structures that will enable the integration of quality management as a normal, integral and continuous function within the institution.

# CHAPTER SEVEN CONCLUSION

#### 7.1 SIGNIFICANCE AND STRENGTHS OF THE STUDY

By its very nature HE is a complex activity and managing the quality of that activity is complicated and multifaceted. Similar to any management model, the successful implementation of a quality management model requires an appropriate degree of consensus by key stakeholders, without which it will not be effective in the longer-term. Identifying quality values of stakeholders is, therefore, considered critical as they underpin the motivations, expectations and the resultant behaviour of the main participants in HE. However, there are very few studies that seek to determine what the primary stakeholders in HE, namely students, academics and employers view as critical to the quality of HE. Moreover, there is a lack of agreement on the dimensions of quality and quality models in HE. This study, therefore, sought to address a gap in the literature by identifying quality criteria considered to be most important by students, academics and employers. In doing so, it has added a new dimension to the existing literature by providing a framework for managing quality in HE that is based on stakeholder determinants of quality.

The study was located in Muscat, Oman and focused on students and academics on Business Studies and IT undergraduate programmes in five out of the seven private HEPs in Muscat offering these programmes. A sample of the larger employers from both the public and private sector were also included in the survey. Currently, there is a lack of research on HE in Oman and hence this study on quality criteria pertaining to private sector HEPs also provides important and hitherto unavailable insights into the local issues in HE. In order to develop the framework, the methodology adopted was one of sequential methodological triangulation which provided both depth and breadth to the findings which is considered to be one of the main strengths of the study. The quality values of each stakeholder group were first identified through semi-structured in-depth individual interviews with a range of stakeholders from each of the three groups. This data provided an in-depth and comprehensive understanding of stakeholder perspectives. The quality values identified from the interviews were termed the 'stakeholder determinants of quality' (SDQ). The knowledge thus gained of the SDQ together with the issues identified in the interviews provide a more comprehensive understanding than is currently available in the literature of

the quality values of the three stakeholder groups in HE. This understanding was further enhanced and validated by exploring the extent of differences and similarities in quality values among the three groups using quantitative methods. A quantitative survey among a wider population of each stakeholder group was also conducted using a structured questionnaire derived essentially from the stakeholder determinants of quality that emerged from the interviews. The survey identified the criteria on which the three groups of stakeholders were congruent in their values or were significantly different. In combining both qualitative and quantitative methods, the aim was to complement the findings from each method and produce results which highlight the contributions of both and provide an integrated understanding of relevant issues. Thus the qualitative methods informed the quantitative methods and the rich data and resultant findings from both methods informed the framework for managing quality based on stakeholder values. A significant strength of this study is therefore the methodological approach which can be replicated in any setting, and not only offers the advantages of methodological and sample triangulation but also systematically explores and establishes the quality values and similarities and differences thereof between a considerably large population of the primary stakeholders in higher education.

The study established that academic staff and employers place the greatest emphasis on the core academic factors that have a direct impact on HE i.e. teaching and learning, academic standards and the curriculum. They share the perception that enhancing standards, student motivation and commitment and rigour of the learning process and environment, thereby requiring students to be active participants in the educational process are most important. The survey findings showed congruence among the three stakeholders on the importance of the process aspects of HE. i.e. the teaching and learning function. Students' engagement with the learning process and the lecturers' ability to impart, facilitate subject understanding and develop behavioural and transferable skills are considered some of the most critical issues in managing quality by all three groups. There is strong evidence that understanding a static corpus of subject knowledge is no longer considered an indicator of quality by all three stakeholders, but particularly so by employers. Accordingly, a very significant finding of this study is that employers, academic staff and students, although to a lesser extent, consider the development of core transferable skills and behavioural skills as possibly more important than even subject specific knowledge and practical understanding of the subject. Employers attach the lowest importance to the outcomes and assessment dimension. While

the study also established that students and academic staff, in particular, attach comparatively more importance to assessments than employers, the survey responses of all three stakeholders' imply a lack of confidence on the ability of current assessments to accurately evaluate the overall skills and critical attributes of graduates.

The survey also found that the most significant purpose of HE is to develop qualities in students that will allow them to act with a high degree of autonomy by equipping them with the skills to think critically. Furthermore, consistent with this purpose, the transformative definition of quality in HE found considerable support from all stakeholders. However, it also revealed that it is undesirable and perhaps impossible to arrive at a single definition of quality or purpose for HE. While there is no overall consensus on defining quality in HE, all the definitions have considerable support, with some definitions finding more support according to each stakeholder perspective. This underlines the multidimensional character of quality as emphasised in the literature and requires HEPs to achieve an appropriate balance between the key objectives of HE as well as consider a range of qualities as correlates of quality in HE. All three stakeholders also identify with the value added concept of HE where the value added to students in terms of overall development, knowledge and skills, which may not necessarily be reflected in the grades obtained, is what constitutes quality in HE. Stakeholders do not consider that high admission standards and high grades at the point of entry are necessary indicators of the quality of subsequent educational provision.

However, although student transformation was strongly emphasised, a key finding of this study is that the majority of students have no awareness of the importance of student participation in the transformation process. These findings have significant implications for HEPs as lack of student participation will negatively affect the outcomes of HE. While students attach the most importance to the process of teaching and learning and the resource inputs that affect their learning, they attribute the least importance to their own input. They show strong resistance to student-centred teaching methods and the majority prefer traditional teaching and assessment methods that allow them to remain passive participants. The lack of emphasis by students on their own role in the learning process is best highlighted by the fact that only a small proportion accept responsibility for their own academic success while the larger proportion place the responsibility for ensuring students' academic success on the institution. The findings indicate a gap between students on the one

hand and employers and academics on the other, regarding the importance of student participation in all aspects of the learning process. The findings are also consistent with the literature that students may not really value high academic standards or may not value standards as highly as academic staff.

The study also revealed that while there are no significant differences between the quality values of IT and Business students, there are significant differences between IT and Business academic staff. IT academic staff attach less importance to maintaining standards, developing core transferable skills, student participation and are less open to the use of non-traditional methods of teaching and assessment in comparison to Business staff. There is thus more congruence between employers and Business staff on learner transformation and student participation, while IT staff are more congruent towards students' views in certain aspects.

Based on the findings from the interviews and the survey with three groups of stakeholders, the study offers a unique framework for managing quality in higher education, i.e. the framework of stakeholder determinants of quality in higher education. The uniqueness of this quality framework lies in its approach as it seeks to address the differences or gaps in quality values between the stakeholder groups while it builds on the congruencies in their views. It identifies the key areas that must be addressed, monitored and evaluated by HEPs in order to manage the quality of their provision. The SDQ framework recommends that in order to achieve the desired outcomes, HEPs must provide for what has been termed 'student institutional socialisation', which is similar to the service sector equivalent of customer socialization, whereby students are supported in acquiring the skills and knowledge required to interact effectively with the HEP so that they can achieve planned outcomes. A key objective of student institutional socialisation must be to develop congruence between the quality values of students and those of academic staff. Managing student expectations, motivation and thereby behaviour are an important part of this process and must be considered as a core issue for academic management. The framework also emphasises learner transformation as the ultimate purpose of higher education through a balanced approach to subject knowledge and transferable skills in all aspects of curriculum, teaching and learning and assessment.

With the increasing concern that quality assurance requirements are diverting too much of academics' attention from their primary role of teaching and research, what is needed is an approach to quality that is integrated into the normal routine of any HEP, rather than focusing solely on the requirements of external regulatory bodies. This study offers an insight into an approach to quality management that is based on an understanding of key values of the main participants in HE. It is held that such an approach will help develop shared values and quality consciousness within institutions which quality assurance regimes that are externally imposed have not been very successful in achieving.

#### 7.2 FUTURE IMPLICATIONS

The strength of the SDQ framework stems from the fact that it is based on a combined stakeholder perspective and in its ability to draw attention to the gaps and congruencies among stakeholder groups, thereby ensuring that gaps are monitored properly and managed effectively. Any HEP in Oman that falls within the parameters of this study can implement the framework by first conducting a detailed evaluation of their own individual position in relation to each of the criteria listed in the framework. This would entail a thorough and critical self-review on how the HEP is positioned in each of the five dimensions of quality in the framework, namely, admissions criteria, institutional, resources, teaching and learning and assessment, a process which will require the HEP to evaluate the comprehensiveness of their own existing provision. Following this evaluation, the HEP would need to plan how it can satisfactorily bridge the gap between its own current position and the key areas as highlighted in the framework. Adequate emphasis must be given to learner transformation and to critical and personal development, which has been identified as the ultimate purpose of higher education. In order to do so, the institution must ensure that its curriculum design, teaching and learning methods and opportunities and assessment strategies all facilitate learner transformation, and provide an adequate balance between subject knowledge and skills in all aspects of its provision. In this respect, the role of the lecturer in managing student expectations, needs and motivations is fundamental. It is vital that lecturers use appropriate teaching methods in order to stimulate active learning and maximise participation. Academic staff will need strong support from the management and the HE community in general if they are to fulfill the role described above. It will require a coordinated and consistent approach by the HEP in order to reinforce the message of student participation.

HEPs outside of the parameters of this study could replicate the methodological approach to determine if there are differences in the SDQ as a result of their differing subject disciplines, geographical location or other contextual factors. Nevertheless, the SDQ framework offers several areas that must be addressed by any HEP irrespective of its own individual position, or how much it differs from the HEPs in this study in terms of subject discipline, geographical area or institutional background and ownership. Regardless of these sorts of differences, the student population is diverse, and this lack of homogeneity must be addressed and managed carefully by all HEPs in order to maximise learning. It is therefore highly recommended that all HEPs invest sufficient time and resources in a continuous process of student institutional socialisaton, as this is the first and most critical step in managing students' expectations and motivation. Without it, students are unlikely to understand what is expected of them. With it, greater congruence between internal stakeholder values should be achieved, which is essential in order to enhance the quality of the teaching and learning experience, and the ultimate outcome of HE.

#### 7.3 LIMITATIONS

Given that the current study is limited to undergraduate programmes in Business Studies and IT disciplines, it is entirely possible that there are differences in quality values of students and academic staff from other subject disciplines. Furthermore, as the study focused on private HE in Oman, it may lack applicability to other HE environments. Although employers were included in the study as the primary external stakeholder, the study does not consider the views of other external stakeholders such as the Ministry of Higher Education who have a key regulatory role in the HE sector.

#### 7.4 SCOPE FOR FURTHER RESEARCH

This study was limited to private HEPs in Oman. Further research into quality values of stakeholders of government owned HEPs will provide an understanding of whether such quality values will differ according to the ownership orientation of the HEP. Research into

the quality values of students and academic staff in other fields of study and employers in these sectors would be valuable in determining if quality values are generalisable across other subject disciplines. Future studies could also explore quality values of key stakeholders in other countries and in other subject disciplines and with different types of tertiary institutions in order to test whether the results obtained are general and consistent across different samples. The study finds significant lack of awareness among students on their own role as co-participants in the HE process and also highlights the often unrealistic expectations that students may have for HEPs and their tutors. Bridger (2007) emphasises that students' values, beliefs and experiences, must be identified so that they can be incorporated and respected. However, there is very limited literature on how student expectations can be managed and students can be motivated to contribute effectively to the teaching and learning process. This study illuminates the need for further research into student expectations and ways in which such expectations can be managed effectively.

#### REFERENCES

Abdullah, F. (2006), "Measuring service quality in higher education: three instruments compared", *International Journal of Research & Method in Education*, Vol. 29, No. 1, pp. 71–89.

Aly, N. and Akpovi, J. (2001), "Total quality management in California public higher education", Quality *Assurance in Education*, Vol. 9, No. 3, pp. 127-131.

Amaral, A., 2007, "Higher education and quality assessment: The many rationales for quality", in Bollaert, L.; Brus, S.; Curvale, B.; Harvey, L.; Helle, E.; Jensen, H.T.; Komlejnovic, J.; Orphanides, A. and Sursock, A. (eds), *Embedding Quality Culture in Higher Education*, Brussels, EUA, pp. 6-10. http://www.fup.pt/admin/cipes/docs/B1.pdf accessed, 9 September, 2007.

Amaratunga, D. and Baldry, D. (2001), "Case study methodology as a means of theory building performance measurement in facilities management organisations", *Work Study*, Vol. 50, No.3, pp. 95-104.

Amin, M.R and Amin, N.A. (2003), "Benchmarking learning outcomes of undergraduate business education", *Benchmarking: An International Journal*, Vol. 10, No.6, pp. 538-558.

Anderson, E. (1995), "High tech v. high touch: a case study of TQM implementation in higher education", *Managing Service Quality*, Vol.5, No.2, pp.48-56.

Anderson, L. (2000), "Teaching development in higher education as scholarly practice: a reply to Rowland et al. turning academics into teachers", *Teaching in Higher Education*, Vol. 5 No.1, pp.23-31.

Anon (2005), "Planning and evaluating content, Implementing a new design for an undergraduate programme, *Development and Learning in Organizations*, Vol. 20, No 1, pp.32-34, available at http://www.emeraldinsight.com/10.1108/13660750510611189, (accessed 11 November, 2006).

Bailey, D. and Bennett, J. (1996), "The realistic model of higher education", *Quality Progress*, Vol. 29, pp. 77–83.

Barnet, R (1994), "The limits of competence, knowledge, higher education and society", *Society for Research into Higher Education*, Buckingham: Open University Press.

Baruch, Y. and Leeming, A. (1996)," Programming the MBA programme - the quest for curriculum", *Journal of Management Development*, Vol. 15, No. 7, pp. 27-36.

Basave, A. (1998), "Integral philosophy of education; a new paedeia", Available at: http://www.bu.edu/wcp/Papers/Educ/EducBasa.html (accessed 19 August, 2007).

Becket, N. and Brookes, M. (2005), "Analysing Quality Audits in Higher Education", *Promoting good practice in Learning, Teaching and Assessment in Higher Education*, Brookes eJournal of Learning and Teaching, Vol. 1, No. 2, Available at **www.brookes.** 

ac.uk/publications/bejlt/volume1issue2/academic/becket\_brookes.html! (accessed 12 September, 2006) .

Bernard, H.R. (2000), Social Research Methods: Qualitative and Quantities Approaches, California: Sage Publications.

Bourner, T. (1998), "More knowledge, new knowledge: the impact on education and training", *Education* + *Training*, Vol. 40, No.1, pp. 11-14.

Bradley, J. (1993), "Methodological issues and practices in qualitative research", *Library Quarterly*, Vol.63, pp.431-449.

Bridger, J. (2007), "From passive to active learners: The lived experience of nurses in a specialist nephrology nursing education programme", *Journal of Workplace Learning*, Vol. 19, No. 2, pp. 78-91.

Brigham, S.E. (1993), "TQM: lessons we can learn from industry", *Change*, Vol. 25, pp. 42–51.

Bruneau, W. and Savage, D.C. (2002), Counting Out the Scholars: The Case Against Performance Indicators in Higher Education, Toronto: James Lorimer & Company, Ltd.

Burton, L. (2001), "Confounding Methodology and Methods", *British Journal of Sociology of Education*, Vol. 22, No.1. pp. 171-175.

Bryman, A. and Cramer, D. (1995), *Quantitative data analysis for social scientists*, London: Routledge.

Cahill, D. J. (1996), "When to use qualitative methods: a new approach", *Marketing Intelligence and Planning*, Vol. 14, No. 6, pp. 16-20.

Calvo-Mora, A., Leal, A. and. Roldán, J. (2006), "Using enablers of the EFQM model to manage institutions of higher education", *Quality Assurance in Education*, Vol. 14, No. 2, pp. 99-122.

Campell, C. and Rozsnyani, C. (2002), "Quality Assurance and the Development of Course Programs", *Papers on Higher Education*, Bucharest, Unesco-CEPES, Available at www.emeraldinsight.com/.../published/emeraldfulltextarticle/pdf/120014010 3\_ref.html (accessed on 13 June, 2006).

Carlson, P.M. and Fleisher, M.S. (2002), "Shifting realities in higher education: today's business model threatens our academic excellence", *International Journal of Public Administration*, Vol. 25, No.9/10, pp.1097-111.

Carroll, M. (2006), "Plan for an Omani Higher Education Quality Management System: The Quality Plan," Draft v4 For Public Consultation, Oman Accreditation Council, Ministry of Higher Education, Oman.

Chadwick, P. (1995), "TQM at South Bank University: issues in teaching and learning", *Quality Assurance in Education*, Vol. 3, No.1, pp. 39-44.

Cheng, Y.C. and Tam, W.M. (1997), "Multi-models of quality in education", *Quality Assurance in Education*, Vol. 5, No.1, pp. 22-31.

Chua, C. (2004), "Perception of Quality in Higher Education", *Proceedings of the Australian Universities Quality Forum*, 2004, AUQA Occasional Publication, Available at http://www.auqa.edu.au/auqf/2004/program/papers/Chua.pdf (accessed on 11 April, 2007)

Clayson, D.E. and Haley, D.A. (2005), "Marketing models in education: students as customers, products, or partners", *Marketing Education Review*, Vol. 15, No.1, pp.1-10.

Creswell, J. W. (1994), *Research Design: Qualitative and Quantitative Approaches*, California: Sage Publications.

Cronin, J.J., Taylor, S.A. (1992), "Measuring service quality: a re-examination and extension", *Journal of Marketing*, Vol. 56, No. 7, pp.55-68.

Colling, C. and Harvey, L. (1996), "Quality control, assurance and assessment – the link to continuous improvement", *Quality Assurance in Education*, Vol. 3, No. 4. pp. 30-34.

Cook. M.J. (1997), "A student's perspective of service quality in education", *Total Quality Management*, Vol. 8, Nos 2&3, pp.120-125.

Cullen, J., Joyce, J., Hassall, T. and Broadbent, M. (2003), "Quality in higher education: from monitoring to management", *Quality Assurance in Education*, Vol. 11, No. 1, pp. 5-14.

Cuthbert, P.F. (1996.a.), "Managing service quality in HE: is SERVQUAL the answer? Part 1", *Managing Service Quality*, Vol. 6, No.2, pp. 11-16.

Cuthbert, P.F. (1996.b), "Managing service quality in HE: is SERVQUAL the answer? Part 1", *Managing Service Quality*, Vol. 6, No.3, pp. 31-35.

Cyert, R.M. (1993), "Universities, competitiveness and TQM: a plan of action", *Public Administration Quarterly*, Vol. 17, pp. 10–18.

Davies, A. and Fitchett, J.A. (2005), "Beyond incommensurability? Empirical expansion on diversity in research", *European Journal of Marketing*, Vol. 39, No. 3/4, pp. 272-293

Deem, R and Brehony, K.J. (2005), "Management as ideology: the case of 'new managerialism' in higher education", *Oxford Review of Education*, Vol. 31, No. 2, pp. 217-235

Doherty, G.D. (1997), "Quality, standards, the consumer paradigm and developments in higher education", *Quality Assurance in Education*, Vol. 5, No. 4, pp. 239-248.

De Rada, V. (2005), "Measure and control of non-response in a mail survey", *European Journal of Marketing*, Vol. 39, No. 1, pp. 16-32.

Díaz-Martínez, C and Navarro, P. (1997), "Meta-Analysis of Surveys from a Qualitative Perspective" Available at <a href="http://www.netcom.es/pnavarro/Publicaciones/Meta-analysis Surveys.html">http://www.netcom.es/pnavarro/Publicaciones/Meta-analysis Surveys.html</a> (accessed 4 March, 2006).

Dickson, K.D., Pollock, A.and Troy, J. (1995), "Perceptions of the value of quality assessment in Scottish education", *Assessment and Evaluation in Higher Education*, Vol. 20, No.1, pp.59-66.

Dillon, C. and Hodgkinson, L. (2000), "Programme specification in a flexible, multidisciplinary curriculum environment: an Open University perspective", *Quality Assurance in Education*, Vol. 8, No. 4, pp. 203-211.

Drucker, P., F. (1994), *Innovation and Entrepreneurship*, Oxford: Heinemann.

Eagle, L. and Brennan R, (2007), "Are students customers? TQM and marketing perspectives", *Quality Assurance in Education*, Vol. 15, No. 1, pp. 44-60.

Easterby-Smith, M. (1991), *Management Research: An Introduction*, London: Sage Publications.

Edvardsson, B. (2005), "Service quality: beyond cognitive assessment", *Managing Service Quality*, Vol.15, No. 2, pp. 127-131.

Ellis, R. (1995), "Quality Assurance for university teaching: issues and Approaches", in Ellis, R (ed) *Quality Assurance for University Teaching*, The Society for Research into Higher Education, Open University Press, pp. 3-15.

Elton, L (1998), "Are degree standards going up, down or sideways?", *Studies in Higher Education*, Vol. 23 No.1, pp.35-42.

Emery, C., Kramer, T. and Tian, R. (2001), "Customers vs. products: adopting an effective approach to business students", *Quality Assurance in Education*, Vol. 9, No.2, pp.110-15.

Emiliani, M.L. (2004), "Is management education beneficial to society?", *Management Decision*, Vol. 42, No.3/4, pp.481-98.

Emiliani, M.L. (2005), "Using kaizen to improve graduate business school degree programs", *Quality Assurance in Education*, Vol. 13, No.1, pp. 37-52.

Eriksen, S.D. (1995), "TQM and the transformation from an élite to a mass system of higher education in the UK", *Quality Assurance in Education*, Vol.3, No. 1, pp. 14-29.

Farrugia, C. (1996), "A continuing professional development model for quality assurance in higher education", *Quality Assurance in Education*, Vol. 4, No. 2, pp. 28-34.

Fogelman, K. (2002), "Surveys and Sampling" in Coleman, M and Briggs, A.(eds) *Research Methods in Educational Research*, London: Paul Chapman.

Fowler, F.J. Jr (1995), *Improving Survey Questions: Design and Evaluation*, Thousand Oaks: Sage.

Fullan, M. (2001), *The New Meaning of Educational Change*, London: Routledge Palmer.

Galloway, L. (1998), "Quality perceptions of internal and external customers: a case study in education administration", *The TQM Magazine*, Vol. 10, No.1, pp.20-6.

Garvin, D.A (1988), Managing Quality: The Strategic and Competitive Advantage, Free Press, New York.

Goodliffe, T. (2004), "Introducing Personal Development Planning: Educating Engineers for the Workplace of the Future", *Global Journal of Engineering Education*, Vol.8, No.2, pp. 147-152.

Gordon, J. (2003), "Personal and professional development: Fostering students' personal and professional development in medicine: a new framework for PPD", *Medical Education*, Vol. 37, No. 4, pp. 341-349.

Gouthro, P.A., Taber, N. and Brazil, A. (2006), "Universities as Inclusive Learning Organizations for Women", *Canadian Association for the Study of Adult Education* (CASAE), *National Conference On-Line Proceedings*, pp.114-119, Available at http://www.oise. utoronto.ca /CAS AE/ cnf2006 /2006online proceedings/ CAS2006 Patricia%20A% 20Gouthro.pdf (accessed 11 January, 2007).

Gibbs, G. and Simpson, C. (2005), "Conditions under which assessment supports students' learning", *Learning and Teaching in Higher Education*, Issue 1, pp. 3-31.

Gibbs, P. and Iacovidou, C. (2004), "Quality as pedagogy of confinement: is there an alternative?", *Quality Assurance in Education*, Vol.12, No. 3, pp. 113-119.

Gorard, S. (2006), "Re-analysing the value-added of primary schools", Paper presented at the British Educational Research Association Annual Conference, University of Glamorgan, Education-Line database, Available at <a href="http://www.york.ac.uk/depts/educ/Research PaperSeries/Paper%2015%20Valueadded%20in%20primary%20schools.pdf">http://www.york.ac.uk/depts/educ/Research PaperSeries/Paper%2015%20Valueadded%20in%20primary%20schools.pdf</a> (accessed 19 August, 2007).

Green, D. (1994), "What is higher education? Concepts, policy and practice", in Green, D. (ed), *What Is Quality in Higher Education*?, Buckingham: The SRHE and Open University Press, pp.3-20.

Greenbaum, T.L. (1998), *The Handbook for Focus Group Research*, Thousand Oaks: Sage.

Gregory, S. (1995), "Using qualitative research for the sociology of food", *British Food Journal*, Vol. 97, No. 7, pp. 32-35.

Grey, C. (2004), "Reinventing business schools: the contribution of critical management education", *Academy of Management Learning and Education*, Vol. 3, No.2, pp.178-86.

Guba, E.G. (1985), "The context of emergent paradigm research", in Lincoln, Y.S. (ed), *Organisational Theory and Inquiry*, United Kingdom, Sage.

Guidebook for Universities and Colleges, Ministry of Higher Education: 2006

Gupta, A. and Chen, I. (1995), "Service quality: implications for management development", *International Journal of Quality & Reliability Management*, Vol. 12, No. 7, pp. 28-35.

Hammersley, M. (1987), "Some notes on the terms 'validity' and 'reliability', *British Educational Research Journal*, Vol.13, No. 1, pp. 73-81.

Hammersley, M. (1998), "The relationship between qualitative and quantitative research: paradigm loyalty versus methodological eclecticism", in Richardson, J.T.E. (ed). *Handbook Of Qualitative Research Methods For Psychological And The Social Sciences*, United Kingdom: BPS Books.

Hargreaves, A. (1998), "The emotional practice of teaching", *Teaching and Teacher Education*, Vol. 14, No.8, pp.835-54.

Harvey, L. (2005), "A history and critique of quality evaluation in the UK", *Quality Assurance in Education*, Vol. 13, No. 4, pp. 263-276.

Harvey, L. and Green, D. (1993), "Assessing quality in higher education: a transbinary research project", *Assessment and Evaluation in Higher Education*, Vol. 18, No.2, pp.143-9.

Harvey, L. and Knight, P. (1996), *Transforming Higher Education*, Birmingham: SREH & Open University Press.

Hawawini, G. (2005), "The future of business schools", *Journal of Management Development*, Vol. 24, No. 9, pp.770-83.

Haworth, J.G., and Conrad, C.F. (1997), *Emblems of quality in higher education: developing and sustaining high quality programs*. Boston: Allyn and Bacon.

Healy, M. and Perry, C. (2000), "Comprehensive criteria to judge validity and reliability of qualitative research within the realism paradigm", *Qualitative Market Research: An International Journal*, Vol. 3, No. 3, pp. 118-126.

Hennig-Thrau, T., Langer, M.F. and Hansen, U. (2001), "Modeling and managing student loyalty: an approach based on the relationship quality", *Journal of Service Research*, Vol. 3, No.4, pp.331-44.

Hewitt, F and Clayton, M. (1999) "Quality and complexity – lessons from English higher education", *International Journal of Quality & Reliability Management*, Vol. 16, No. 9, pp. 838-858.

Heyneman, S. P. (2006), "Global Issues in Higher Education", *Journal U.S.A*, Available at http://usinfo.state.gov/journals/itgic/0206/ijge/heyneman.html (accessed 31 October, 2006).

Higher Education Funding Council (England) HEFCE (1993), *Assessments of the Quality of Education*, Bristol. Available at http://www.hefce.ac.uk/Pubs/hefce/1994/m1\_94.htm (accessed 2 June, 2006)

Hill, F.M. (1995), "Managing service quality in higher education: the role of the student as primary consumer", *Quality Assurance in Education*, Vol. 3, No. 3, pp. 10-21.

Hill, Y., MacGregor, J. and Dewar, K. (1996), "Nurses' access to higher education: a quality product", *Quality Assurance in Education*, Vol. 4, No. 2, pp. 21-27.

Hill, Y., Lomas, L. and MacGregor, J. (2003), "Students' perceptions of quality in higher education", *Quality Assurance in Education*, Vol.11, No.1, pp. 15-20.

Hoepfl, M.C. (1997), "Choosing Qualitative Research: A Primer for Technology Education Researchers", *Journal of Technology Education*, Vol. 9, No. 1, pp.47-63.

Holland, P. and Pyman, A. (2006), "Corporate universities: a catalyst for strategic human resource development, *Journal of European Industrial Training*, Vol. 30, No.1, pp. 19-31.

Holmes, G and McElwee, G. (1995), "Perspectives: Total quality management in higher education: how to approach human resource management", *The TQM Magazine*, Vol. 7, No. 6, pp. 5–10.

Hsieh, H. (2005), "Shaping universities as learning organisations in Taiwan: a comparative study on comprehensive and technology universities", *World Transactions on Engineering and Technology Education*, Vol.4, No.2, pp 245-248.

Hung, Y. H.. Huang, M. L and Cheng, K. S. (2003), "Service quality evaluation by service quality performance matrix", *Total Quality Management*, Vol. 14, No. 1, pp. 79-89.

Hyde, K.F. (2000), "Recognising deductive processes in qualitative research", *Qualitative Market Research: An International Journal*, Vol. 3, No. 2, pp. 82-90.

Iacobucci, D., Grayson, K.A. and Omstrom, O.L. (1994), "The calculus of service quality and customer satisfaction: theoretical and empirical differentiation and

integration", in Swartz, T.A., Bowen, D.E. and Brown, S.W. (eds), *Advances in Services Marketing and Management*, Greenwich: JAI Press, Vol. 3, pp.16-84.

Jackson, S. (2000), "Achieving clinical governance in Women's Services through the use of the EFQM Excellence Model", *International Journal of Health Care Quality Assurance*, Vol.13, No. 4, pp. 182-190.

Jameson, S.M and Holden, R. (2000), "Graduateness – who cares? Graduate identity in small hospitality firms, *Education* + *Training*, Vol. 42, No. 4/5, pp. 264-271.

Jamieson, S. (2004), "Likert scales: how to (ab)use them", *Medical Education*, Vol. 38, pp. 1212–1218.

Johnson, R.N. and Deem, R. (2003), "Talking of students: Tensions and contradictions for the manager-academic and the university in contemporary higher education", *Higher Education*, Vol. 46, pp. 289–314.

Johnson, R.B and Onwuegbuzie, A.J. (2004), "Mixed Methods Research: A Paradigm whose time has come", *Educational Researcher*, Vol. 33, No.7, pp. 14-26.

Johnson, C. and Spicer, D. (2006), "A case study of action learning in an MBA program", *Education* + *Training*, Vol. 48, No. 1, pp. 39-54.

Joseph, M. and Joseph, B. (1997), "Service quality in education: a student perspective", *Quality Assurance in Education*, Vol. 5, No. 1, pp. 15-21.

Kanji, G.K. and Malek, A. (1999), "Total quality management in UK higher education institutions", *Total Quality Management*, Vol. 10, pp. 129–153.

Kember, D. (1997), "A reconceptualisation of the research into university academics conceptions of teaching", *Learning and Instruction*, Vol. 7, No. 3, pp. 255-275.

Konidari, V and Abernot, Y. (2006), "From TQM to learning organisation: Another way for quality management in educational institutions", *International Journal of Quality & Reliability Management*, Vol. 23, No 1, pp. 8-26.

Lagrosen, S., Seyyed-Hashemi, R. and Leitner, M. (2004), "Examination of the dimensions of quality in higher education", *Quality Assurance in Education*, Vol.12, No. 2, pp. 61-69.

Lammers, H.B., Kiesler, T., Curren, M.T., Cours, D. and Connett, B. (2005), "How hard do I have to work? Student and faculty expectations regarding university work", *Journal of Education for Business*, Vol. 80, No.4, pp.210-13.

Lammers, W., and Murphy, J. (2002) "A profile of teaching techniques used in the university classroom", *Active Learning in Higher Education*, Vol. 3, No.1, pp.54-67 in Hill, Y., Lomas, L. and MacGregor, J. (2003), "Students' perceptions of quality in higher education", *Quality Assurance in Education*, Vol.11, No.1, pp. 15-20.

Lawrence, J.J and Pharr, S. (2003), "Evolution of admission standards in response to curriculum integration", *Quality Assurance in Education*, Vol. 11,No. 4, pp. 222-233.

LeBlanc, H.P. (1995), "Syncretism of Quantitative and Qualitative Research Paradigms: The case for methodological triangulation", Available at <a href="http://www.eric.ed.gov/ERICDocs/data/ericdocs2sql/content\_storage\_01/0000019b/80/14/62/39.pdf">http://www.eric.ed.gov/ERICDocs/data/ericdocs2sql/content\_storage\_01/0000019b/80/14/62/39.pdf</a> (accessed March, 2007)

Lewis, B. R. (1991) "Customer care in service organizations", *Management Decision*, Vol. 29, No.1, pp. 31-34.

Lind, D.A., Marchal, W.G. and Wathen, S.A. (2005), *Statistical Techniques in Business and Economics*, (Int.ed), NY: McGraw Hill, Irwin.

Lomas, L. (2002), "Does the development of mass education necessarily mean the end of quality?", *Quality in Higher Education*, Vol. 8 No.1, pp.71-80.

Lomas, L. (2004), "Embedding quality: the challenges for higher education", *Quality Assurance in Education*, Vol. 12, No 4, pp. 157-165.

Lomas, L. and Tomlinson, K. (2000), "Standards: the varying perceptions of senior staff in higher education institutions", *Quality Assurance in Education*, Vol.8, No.3, pp. 131-139

Macfarlane, B. and Lomas, L. (1995), "Client-based management education: values and quality", *Management Development Review*, Vol.8, No. 1, pp. 32-36.

Madu, C.N. (1998), "An empirical assessment of quality: research considerations" *International Journal of Quality Science*, Vol. 3, No. 4, pp. 348-355.

Madu, C.N. and Kuei, C.H. (1993), "Dimensions of quality teaching in higher institutions", *Total Quality Management*, Vol. 4, No.5, pp.325-38.

Maggs-Rapport, F. (2001), "Best research practice': in pursuit of methodological rigour", *Journal of Advanced Nursing*, Vol. 35, No. 3, pp. 373-383.

Mahapatra, S.S. and Khan M. S. (2007), "A framework for analysing quality in education settings", *European Journal of Engineering Education*, Vol. 32, No. 2, pp. 205–217.

Mangan, J., Lalwani, C. and Gardner, B. (2004), "Combining quantitative and qualitative methodologies in logistics research", *International Journal of Physical Distribution & Logistics Management*, Vol. 34, No. 7, pp. 565-578.

Mangold, W.G. and Babakus, B. (1991), "Service quality: the front stage vs. the back stage perspective", *Journal of Services Marketing*, Vol. 5, No.4, pp.59-70.

Martinsuo, M. (2001), "Balancing large scope and project manageability in a qualitative research design", *Management Decision*, Vol.39, No. 7, pp. 539-550.

Marsh, H.W. and Roche, L.A. (2000), "Effects of grading leniency and low workload on students' evaluation of teaching: popular myth, bias, validity, or innocent bystanders?", *Journal of Educational Psychology*, Vol. 92, No.1, pp.202-28.

Mattick, K. and Knight, L. (2007), "High-quality learning: harder to achieve than we think?", *Medical Education*, Vol. 41, No. 7, pp. 638-644.

McGregor, W. (2002), "Preparing for an uncertain future", *Facilities*, Vol. 18, No. 10/11/12, pp. 402-410.

McHardy, P. and Allan, T. (2000), "Closing the gap between what industry needs and what HE provides", *Education* + *Training*, Vol. 42, No. 9, pp. 496-508.

Michael, S (1997), "American higher education system: consumerism versus professorialism", *International Journal of Educational Management*, Vol. 11, No. 3, pp. 117-130.

Middlehurst, R. (1997), "Enhancing quality", in Coffield, F., Williamson, B. (eds), *Repositioning Higher Education*, Buckingham: SRHE/Open University Press.

Middlehurst, R. (2001), "Quality Assurance Implications of New Forms of Higher Education", Part 1: A Typology, *European Network for Quality Assurance in Higher Education*, Helsiniki, Occasional Papers 3, Available at http://www.enqa.eu/files/newforms.pdf (accessed January, 2006).

Miles, M.B. and Huberman, A.M. (1994), *An Expanded Sourcebook: Qualitative Data Analysis*, (2nd Ed.), Newbury Park, CA: Sage Publications.

Milliken, J. (2001), "Qualitative Research and Marketing Management", *Management Decision*, Vol. 39, No.1, pp. 71-77.

Mizikaci, F. (2006), "A systems approach to program evaluation model for quality in higher education", *Quality Assurance in Education*, Vol. 14, No.1, pp. 37-53.

Morgan, D. L. (1998), "Practical strategies for combining qualitative and quantitative methods: Applications to health research", *Qualitative Health Research*, Vol. 3, pp. 362–376.

Morley, L. and Aynsley, S. (2007), "Employers, Quality and Standards in Higher Education: Shared Values and Vocabularies or Elitism and Inequalities?" *Higher Education Quarterly*, Vol. 61, No. 3, pp. 229-249.

Muller, D. and Funnell, P. (1992), "Initiating change in further and vocational education: the quality approach", *Journal of Further and Higher Education*, Vol. 16, No.1, pp.41-9.

Nabi, G. and Bagley, G. (1999), "Graduates perceptions of transferable personal skills and future career preparation in the UK", *Education* + *Training*, Vol. 41, No.4, pp.184-93.

Narasimhan, K. (1997), "Improving teaching and learning: the perceptions minus expectations gap analysis approach", *Training for Quality*, Vol. 5, No. 3, pp. 121-125.

Nunnally, J. C. and Bernstein, I. H. (1994), *Psychometric theory*, (3<sup>rd</sup> ed.). New York: McGraw -Hill

Oldfield, B. M. and Baron, S. (2000), "Student perceptions of service quality in a UK university business and management faculty", *Quality Assurance in Education*, Vol. 8, No. 2, pp. 85-95.

O'Neill, M.A. and Palmer, A. (2004), "Importance-performance analysis: a useful tool for directing continuous quality improvement in higher education", *Quality Assurance in Education*, Vol. 12, No. 1, pp. 39-52.

Onweughuzie, A. J. (2002), "Why can't we all get along? Towards a framework for unifying research paradigms", *Education*, Vol. 122, Issue 3.

Owlia, M and Aspinwall, E (1996), "A framework for the dimensions of quality in higher education", *Quality Assurance in Education*, Vol. 4, No 2, pp. 12-20.

Paloniemi, S. (2006), "Experience, competence and workplace learning", *Journal of Workplace Learning*, Vol. 18, No. 7/8, pp. 439-450.

Parasuraman, A., Zeithaml, V.A. and Berry, L.L. (1994), "Reassessment of expectations as a comparison standard in measuring service quality: implications for further research", *Journal of Marketing*, Vol. 58, No. 1, pp. 111-24.

Parasuraman, A., Zeithaml, V.A. and Berry, L.L. (1988), "SERVQUAL: a multiple-item scale for measuring consumer perceptions of quality", *Journal of Retailing*, Vol. 64 No.1, pp.12-40.

Pennington, G. and O'Neil, M. (1994), "Enhancing the Quality of Teaching and Learning in Higher Education", *Quality Assurance in Education*, Vol. 2, No. 3, pp. 13-18.

Perrone, L. and Vickers, M.H. (2003), "Life after graduation as a "very uncomfortable world: an Australian case study", *Education + Training*, Vol. 45, No.2, pp. 69-78.

Policnao, A. (2007), "The rankings game: and the winner is", *Journal of Management Development*, Vol.26, No. 1, pp. 43-48.

Pounder, J. (1999), "Institutional performance in higher education: is quality a relevant concept?", *Quality Assurance in Education*, Vol. 7, No. 3, pp. 156-165.

Pounder, J.S. and Coleman, M. (2002), "Women-better leaders than men? In general and educational management it still "all depends", *Leadership and Organisational Development Journal*, Vol. 23, No. 3, pp.122-133.

Pupius, M. (2001), "Quality – the end or just the beginning?", paper presented at The End of Quality?, *Society for Research in Higher Education Conference*, 25-26 May, Birmingham. Available at http://www.qualityresearchinternational.com/papers/pupius.pdf (accessed June, 2006).

Pursglove, J. and Simpson, M. (2007), "Benchmarking the performance of English universities", *Benchmarking: An International Journal*, Vol.14, No. 1. pp. 102-122.

Quality Assurance Agency for Higher Education (2004), "A brief guide to quality assurance in UK higher education", Available at **www.qaa.ac.uk/aboutus/heGuide/guide.asp** (accessed 10 February, 2006).

Rad, A.M. (2005), "A survey of total quality management in Iran: Barriers to successful implementation in health care organisations", *Leadership in Health Services*, Vol.18, No.3, pp. 12-34

Ratcliffe, J. (2002), "Scenario planning: strategic interviews and conversations", *Foresight*, Vol. 4, No. 1, pp. 19-30.

Rao, S. and Perry, C. (2003), "Convergent interviewing to build a theory in under-researched areas: principles and an example investigation of Internet usage in inter-firm relationships", *Qualitative Market Research: An International Journal*, Vol.6, No. 4, pp. 236-247

Reichardt, C.S. and Rallis, S.F. (1994), "Qualitative and quantitative inquiries are not incompatible: A call for a new partnership", in Reichardt, C.S. and Rallis, S.F (eds), *The qualitative-quantitative debate: New perspectives*, pp.85-92, Jossey-Bass, San Francisco

Riege, A.M. (2003), "Validity and reliability tests in case study research: a literature review with "hands-on" applications for each research phase", *Qualitative Market Research: An International Journal*, Vol. 6, No.2, pp. 75-86.

Remenyi, D., Williams, B., Money, A. and Swartz, E. (1998), *Doing Research in Business and Management*, London: Sage Publications.

Richards, K. (2003), Qualitative Inquiry in TESOL, New York: Palgrave Macmillan.

Rhodes, G. and Shiel, G. (2007), "Meeting the needs of the workplace and the learner through work-based learning", *Journal of Workplace Learning*, Vol. 19, No. 3, pp. 173-187.

Robson, C. (1993), Real Word Research: a Resource for Social Scientists and Practitioners-Researchers, Oxford: Basil Blackwell.

Rocco, T.S, Bliss, L.A. and Gallagher, L. and Pérez-Prado, A. (2003), "Taking the next step: mixed methods research in organizational systems", *Information Technology, Learning, and Performance Journal*, Vol. 21, No. 1, pp. 19-29.

Roffe, I.M. (1998), "Conceptual problems of continuous quality improvement and innovation in higher education", *Quality Assurance in Education*, Vol.6, No. 2, pp. 74-82.

Rolfe, H. (2002), "Students' demands and expectations in an age of reduced financial support: the perspectives of lecturers in four English universities", *Journal of Higher Education Policy and Management*, Vol. 24 No.2, pp.171-82.

Rowland, S., Byron, C., Furedi, F., Padfield, N. and Smyth, T. (1998), "Turning academics into teachers", *Teaching in Higher Education*, Vol. 3, pp.133-41.

Saad, G.H and Siha, S.(2000), "Managing quality: critical links and a contingency model", *International Journal of Operations & Production Management*, Vol. 20, No. 10, pp. 1146-1164.

Sahney, S., Banwet, D.K. and Karunes, S. (2004), "Conceptualizing total quality management in higher education", *The TQM Magazine*, Vol. 16, No. 2, 145-159

Sahney, S., Banwet, D.K. and Karunes, S. (2006), "An Integrated Framework for Quality in Education: Application of Quality Function Deployment, Interpretive Structural Modelling and Path Analysis", *Total Quality Management*, Vol. 17, No. 2, pp. 265–285.

Sallis, E. (1993), Total Quality Management in Education, London: Kogan Page.

Santiago, R. and Carvalho, T. (2003), "Effects of Managerialism on the Perceptions of Higher Education in Portugal", *Higher Education Policy*, Vol. 17, No. 4, pp. 427-444.

Saunders, I.W. and Walker, M. (1991), "TQM in tertiary education", *The International Journal of Quality & Reliability Management*, Vol. 8, pp. 91–103.

Saunders, M., Lewis, P. and Thornhill, A. (2003), *Research Methods for Business Students*, (3rd ed), London: FT Prentice-Hall.

Schray, V. (2006), "Assuring Quality in Higher Education: Key Issues and Questions for Changing Accreditation in the United States", A National Dialogue: *The Secretary of Education's Commission on the Future of Higher Education*, Issue Paper, No.4. Available at http://www.ed.gov/about/bdscomm/list/hiedfuture/reports/schray.pdf (accessed 22 January, 2006).

Senge, P.M. (2000), Schools that Learn: A Fifth Discipline Fieldbook for Educators, Parents, and Everyone Who Cares about Education, New York: Doubleday.

Sitkin, S.B., Sutcliffe, K.M. and Shroeder, R.G. (1994) "Distinguishing control from learning in total quality management: A contingency perspective", *The Academy of Management Review*, Vol. 19, pp. 537–564.

Silvestro, R. (2001), "Towards a contingency theory of TQM in services - How implementation varies on the basis of volume and variety", *International Journal of Quality & Reliability Management*, Vol. 18, No. 3, pp. 254-288.

Sohail, M.S., Rajadurai, J., and Abdul Rahman, N.A. (2003), "Managing quality in higher education: a Malaysian case study", *International Journal of Educational Management* Vol. 1, No. 4, pp. 141-146.

Soutar, G. and McNeil, M. (1996), "Measuring service quality in a tertiary institution", *Journal of Educational Administration*, Vol. 34, No. 1, pp. 72–82.

Srikanthan, G. and Dalrymple, J. (2002), "Developing a Holistic Model for Quality in Higher Education", Available at http://www.cmqr.rmit.edu.au/publications/gsjdicit0 2.pdf (accessed 20 June, 2006)

Srikanthan, G. and Dalrymple, J. (2003), "Developing alternative perspectives for quality in higher education", *The International Journal of Educational Management*, Vol. 17, No. 3, pp. 126-136.

Srikanthan, G. and Dalrymple, J. (2007), "A conceptual overview of a holistic model for quality in higher education", *International Journal of Educational Management*, Vol. 21 No. 3, pp. 173-193.

Stefani, L. (2005), "Assessment of Student Learning: promoting a scholarly approach", *Learning and Teaching in Higher Education*, Issue 1, pp. 51-66.

Stenbacka, C. (2001), "Qualitative research requires quality concepts of its own", *Management Decision*, Vol. 39, No. 7, pp. 551-556.

Stokes, D. and Bergin, R. (2006), "Methodology or "methodolatry"? An evaluation of focus groups and depth interviews", *Qualitative Market Research: An International Journal*, Vol. 9, No. 1, pp. 26-37.

Strauss, A. and Corbin, J. (1990), *Basics of Qualitative Research: Grounded Theory Procedures and Techniques*, London: Sage Publications.

Strauss, A. and Corbin, J. (1994), "Grounded theory methodology", in Denzin, N.K, and Lincoln, Y (eds), *Handbook of Qualitative Research*, Newbury Park, Sage Publications, pp.273-85.

Struyven, K., Dochy, F., and Janssens, S., (2002), "Students' perceptions about assessment in higher education: a review", *Paper presented at the Joint Northumbria/Earli SIG Assessment and Evaluation Conference: Learning communities and assessment cultures*, University of Northumbria at Newcastle, Available at <a href="http://www.leeds.ac.uk/educol/documents/">http://www.leeds.ac.uk/educol/documents/</a> 00002255.html (accessed 12 March, 2006).

Svensson, G. (2003), "A generic conceptual framework of interactive service quality", *Managing Service Quality*, Vol.13, No.4, pp. 267-275.

Tan, K.C. and Kek, S. W. (2004), "Service Quality in Higher Education: Using an Enhanced SERVQUAL Approach", *Quality in Higher Education*, Vol. 10, No. 1, pp.17-24.

Teas, R.K. (1994), "Expectations as a comparison standard in measuring service quality: an assessment of a reassessment", *Journal of Marketing*, Vol. 58, No. 1, p.132-139.

Telford, R and Masson, R. (2005), "The congruence of quality values in higher education", *Quality Assurance in Education*, Vol. 13, No. 2, pp. 107-119.

Thomas, H. (2007), "An analysis of the environment and competitive dynamics of management education", *Journal of Management Development*, Vol. 26, No.1, pp. 9-21.

Tribus, M. (1994), "The application of quality principles in education at Mt. Edgecumbe High School, Alaska", in Doherty, G.D. (eds), *Developing Quality Systems in Education*, London: Routledge, pp. 273-88.

Tribus, M. (1996), "TQM in education: the theory and how to put it to work", in Tribus, M. and Langford, D. (eds), in *Quality Goes to School: Readings on Quality Management in Education*, Arlington: American Association of School Administrators, pp.37-40.

Trow, M. (1994), "Managerialism and the academic profession: the case of England", *Higher Education Policy*, 7, pp. 11–18.

Trow, M. (1996), "On the accountability of higher education in the United States" in Bowen, W. and Shapiro, H. (eds) *Universities and Their Leadership*, Princeton: Princetown University Press.

Vazzana, G. Bachmann, D and Elfrink, J. (1997), "Does higher education practice what it teaches?: A study of total quality management processes in academia answers this question", *Quality Progress*, Vol.30, pp. 67–74.

Watty, K. (2005), "Quality in accounting education: what say the academics?", *Quality Assurance in Education*, Vol. 13, No. 2, pp. 120-131.

Wals, A. and Jickling, B. (2002). "Sustainability in higher education: From doublethink and newspeak to critical thinking and meaningful learning", *International Journal of Sustainability in Higher Education*, Vol. 3, No. 3, pp. 221-232.

Welsh, J.F. and Dey, S. (2002), "Quality measurement and quality assurance in higher education", *Quality Assurance in Education*, Vol. 10, No.1, pp. 17-25.

Weinreich, N.K. (1996), "Integrating quantitative and qualitative methods in social marketing research", *Social Marketing Quarterly*, Available at http://www.socialmarketing.com/research.html (accessed 15 June, 2004).

Welsh Assembly Government, (2007), *Review of Secondary School Performance Measures: Consultation document*, Available at **www.wales.gov.uk/consultations** (accessed on 3 September, 2006).

Westerheijden, D.F. (2000), "Where are the quantum jumps in quality assurance? Developments of a decade of research on a heavy particle", *Higher Education Policy*, Vol. 38, No. 2, pp. 233-54.

West-Burnham, J., and Davies, B. (1994), "Quality management as a response to educational changes", *Studies in Educational Administration*, No.60, pp.49-60.

Wicks, A. and Whiteford, G. (2006), "Conceptual and practical issues in qualitative research: Reflections on a life-history study", *Scandinavian Journal of Occupational Therapy*, Vol.13, No.2, pp. 13-94.

Wiklund, H., Klefso, B., Wiklund, P. and Edvardsson, B. (2003), Innovation and TQM in Swedish Higher Education Institutions Possibilities and Pitfalls, *The TQM Magazine*, Vol. 15, No. 2, pp. 97 -107.

Williams, G. (1993), "Total quality management in higher education: panacea or placebo?" *Higher Education: the International Journal of Higher Education and Educational Planning*, Vol. 25, No.3, pp.229-37.

Williams, G. (1997) "The market route to mass higher education: British experience 1979-1996", *Higher Education Policy*, Vol. 10, No.3/4, pp.275-89.

Williams, P. (2002), "Anyone for enhancement?", *QAA Higher Quality, The Bulletin Of The Quality Assurance Agency For Higher Education*, No.11, pp.1-2, Available at http://www.qaa.ac.uk/news/higherquality/hq11/HQ11.pdf (accessed 24 September, 2004).

Wright, C. and O'Neill, M. (2002) "Service quality evaluation in the higher education sector: an empirical investigation of students' perceptions", *Higher Education Research & Development*, Vol. 21, No. 1, pp.23-39.

Yin, K. (1994), Case Study Research Design and Methods, CA: Sage Publications.

Yorke, M.(1997), "A good league table guide?", *Quality Assurance in Education*, Vol. 5, No. 2, pp. 61-72.

Yorke, M. (1999), "Assuring quality and standards in globalised higher education", *Quality Assurance in Education*, Vol. 7, No.1, pp. 14-24.

Zeithaml, V.A., Parasuraman, A., and Berry, L.L. (1990), *Delivering Quality Service: Balancing Customer Perceptions and Expectations*, New York: The Free Press.

### **APPENDIX: 1**

Table 19: Differences between academic staff on subject area: Mann Whitney Test

		%	%	%		Sum of	Asym Sig. (2-
	Subject	>3	=3	<3	Rank	Ranks	tailed
High academic standards	IT	82.1%	17.9%	.0%	31.78	1239.50	0.024*
	Business	84.8%	15.2%	.0%	42.08	1388.50	
Curriculum practical focus	IT	82.1%	15.4%	2.6	32.55	1269.50	0.048*
'	Business	100%	.0%	.0%	41.17	1358.50	
Job relevance of programmes	IT	97.4	2.6%	.0%	31.81	1240.50	0.012*
	Business	100%	.0%	.0%	42.05	1387.50	
Understanding student needs	IT	89.7%	7.7%	2.6%	31.73	1237.50	0.018*
	Business	100%	.0%	.0%	42.14	1390.50	
Teaching methods	IT	89.7%	7.7%	2.6%	31.99	1247.50	0.025*
	Business	97.0%	3.0%	.0%	41.83	1380.50	
Accessible explanation	IT	84.6%	12.8%	2.6%	31.85	1242.00	0.006**
-	Business	100%	.0%	.0%	42.00	1386.00	
Interactive learning methods	IT	82.1%	17.9%	.0%	32.45	1265.50	0.044*
	Business	100%	.0%	.0%	41.29	1362.50	
Student efforts	IT	84.6%	12.8%	2.6%	31.42	1225.50	0.013*
	Business	100%	.0%	.0%	42.50	1402.50	
Faculty focus on workplace skills	IT	84.6%	12.8%	2.6%	31.27	1219.50	0.012*
	Business	93.9%	6.1%	.0%	42.68	1408.50	0.011*
Overall teaching & learning	IT	92.3%	5.1%	2.6%	31.35	1222.50	0.011*
criteria	Business	93.9%	6.1%	.0%	42.59	1405.50	
Overall assessment criteria	IT	94.9%	5.1%	.0%	32.40	1263.50	0.029*
	Business	93.9%	6.1%	.0%	41.35	1364.50	
Suitability of examinations	IT	35.9%	30.8%	33.3%	41.29	1610.50	0.023*
	Business	18.2%	18.2%	63.6%	30.83	1017.50	
Institutional responsibility	IT	12.8%	33.3%	53.8%	42.50	1657.50	0.005**
	Business	12.1%	12.1%	75.8%	29.41	970.50	
Meeting specified objectives	IT	97.4%	2.6%	.0%	31.58	1231.50	0.014*
	Business	90.9%	9.1%	.0%	42.32	1396.50	
Transformative ability	IT	82.1%	17.9%	.0%	31.92	1245.00	0.027*
	Business	93.9%	6.1%	.0%	41.91	1383.00	
Meeting student needs	IT	89.7%	2.6%	7.7%	41.23	1608.00	0.025*
expectations	Business	54.5%	6.1%	6.1%	30.91	1020.00	
Learning for themselves	IT	94.9%	5.1%	.0%	32.36	1262.00	0.035*
	Business	100%	.0%	.0%	41.39	1366.00	

<sup>&</sup>gt;3 shows % of respondents who selected 4 (very important/ agree ) or 5 (most important/ strongly agree )

<sup>&</sup>gt;4 shows % of respondents who selected 5 (most important/strongly agree)

<sup>\*</sup>Significant at 0.05 level of significance \*\* Significant at 0.01 level of significance

#### **APPENDIX 2**

### SAMPLE OF INTERVIEW SCHEDULE

Stakeholder: Student/Academic/Employer	Gender: Male/Female	
Name:		
Institution:		
Department (academics/employers)		
Programme/Subject (students/academics):		
Designation (academics/employers)		
Level/Year (students)		

- 1. What do you think of first when I say the words "quality higher education"?
- 2. What according to you are the most important issues that affect the quality of higher education? (at least three factors)
- 3. Why are each of these important?/. What is it that about these factors that affect the quality of the education?
- 4. Have you had any negative experiences regarding any of the factors identified by you
- 5. What is the role of the tutor in higher education?
- 6. Is the quality of student intake a key indicator in determining the quality of higher education?
- 7. Do you think the assessment and grades obtained by students are important in evaluating quality? Why?
- 8. What kind of skills should higher education be developing?
- 9. What do you think is more important, assessing skills or subject knowledge?
- 10. How often are you (only students) influenced by your friends' negative comments and opinions?
- 11. Which of the following would you consider more important in higher education (with examples) and why:
  - The inputs (students, resources, curriculum, faculty)
  - The process (of teaching & learning, teaching methods, assessments)
  - The outcome of education: grades, knowledge, skills

# APPENDIX 3 STUDENT PERCEPTION OF QUALITY IN HIGHER EDUCATION: QUESTIONNAIRE

The objective of this questionnaire is to determine what students think is important in managing quality in higher education. The information that is provided will be used for research purposes only and will be kept <u>entirely confidential</u>.

A. Level/Year				
B. Programme	/ specialization:			
C. Male □ Fen	nale □			
D. Name of univ	versity/college			
E. Nationality				
Circle a numbe important.		s most important to y 1 to 5 where 1=least		=most
		entering a programme	· ·	
1) Tilgir adırılıs	2	3	<del>.</del> . 4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		r ability before entry	to a course:	·
1	2	3	4	5
		Average Importance	Quite Important	Most Important
3) <b>Students o</b>	n the programme	have good attitude a	nd commitment to	studies:
l east important	Low Importance	Average Importance	Quite Important	Most Important
		courses on offer:	2dite important	Most Important
1	2	3	4	5
		Average Importance	Quite Important	Most Important
Γ\ <b>Ο</b>				
5) Overall no	w important do yo	ou consider all the abo	ove admission criti	
1	2	3	4	5
1	2	3 Average Importance	ove admission crite 4 Quite Important	
1 Least important  G. INSTIT	2 Low Importance UTIONAL FACTOR	3 Average Importance	4 Quite Important	5 Most Important
1 Least important  G. INSTIT	2 Low Importance UTIONAL FACTOR	3 Average Importance	4 Quite Important	5 Most Important
1 Least important  G. INSTIT  6) Reputation  1	2 Low Importance SUTIONAL FACTOR of the institution 2	3 Average Importance  RS n among the general s	4 Quite Important student population 4	5 Most Important  : 5
1 Least important  G. INSTIT 6) Reputation 1 Least important	2 Low Importance  OUTIONAL FACTOR Of the institution 2 Low Importance	3 Average Importance  RS n among the general so 3 Average Importance	4 Quite Important  student population 4 Quite Important	5 Most Important  : 5 Most Important
1 Least important  G. INSTIT 6) Reputation 1 Least important	2 Low Importance  OUTIONAL FACTOR Of the institution 2 Low Importance	3 Average Importance  RS n among the general s	4 Quite Important  student population 4 Quite Important	5 Most Important  : 5 Most Important
1 Least important  G. INSTIT  6) Reputation 1 Least important 7) Reputation 1	2 Low Importance  Of the institution 2 Low Importance of the institution 2	3 Average Importance 2S n among the general s 3 Average Importance n among employers & 3	4 Quite Important  student population 4 Quite Important general communit 4	5 Most Important  : 5 Most Important  fy: 5
1 Least important  6. INSTIT 6) Reputation 1 Least important 7) Reputation 1 Least important	2 Low Importance  TUTIONAL FACTOR  To of the institution 2 Low Importance To of the institution 2 Low Importance Low Importance	3 Average Importance  2S n among the general s 3 Average Importance n among employers & 3 Average Importance	4 Quite Important  student population 4 Quite Important	5 Most Important  : 5 Most Important
1 Least important  6. INSTIT 6) Reputation 1 Least important 7) Reputation 1 Least important	2 Low Importance  Of the institution 2 Low Importance of the institution 2	3 Average Importance  2S n among the general s 3 Average Importance n among employers & 3 Average Importance	4 Quite Important  student population 4 Quite Important general communit 4 Quite Important 4	5 Most Important  : 5 Most Important  fy: 5
1 Least important  6. INSTIT 6) Reputation 1 Least important 7) Reputation 1 Least important 8) Strict attent 1 Least important	2 Low Importance 2 Low Importance 5 of the institution 2 Low Importance 2 Low Importance 2 Low Importance 1 Low Importance 1 Low Importance 2 Low Importance 1 Low Importance	3 Average Importance  2S Average Importance 3 Average Importance 4 Average Importance 5 Average Importance 6 Average Importance	4 Quite Important  student population 4 Quite Important general communit 4 Quite Important  4 Quite Important	5 Most Important  : 5 Most Important  :y: 5 Most Important  5 Most Important
1 Least important  6. INSTIT 6) Reputation 1 Least important 7) Reputation 1 Least important 8) Strict attent 1 Least important	2 Low Importance 2 Low Importance 5 of the institution 2 Low Importance 2 Low Importance 2 Low Importance 1 Low Importance 1 Low Importance 2 Low Importance 1 Low Importance	3 Average Importance  2S Average Importance 3 Average Importance 4 Average Importance 5 Average Importance 6 Average Importance 7 Avera	4 Quite Important  Student population 4 Quite Important general communit 4 Quite Important  4 Quite Important  4 Quite Important allenges for student	5 Most Important  : 5 Most Important  ty: 5 Most Important  5 Most Important  Most Important  to achieve:
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atten 1 Least important  9) Programm 1	2 Low Importance  TUTIONAL FACTOR  To of the institution 2 Low Importance 2 Low Importance Indance requirement 2 Low Importance 2	3 Average Importance  RS  n among the general s 3 Average Importance n among employers & 3 Average Importance ents for classes: 3 Average Importance mic standards and ch	Quite Important  Student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  4 Quite Important  allenges for student 4	5 Most Important  : 5 Most Important  ty: 5 Most Important  5 Most Important  Most Important  to achieve: 5
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atten 1 Least important  9) Programm 1 Least important	2 Low Importance  TUTIONAL FACTOR TO of the institution 2 Low Importance To of the institution 2 Low Importance Todance requirement 2 Low Importance Es set high acade 2 Low Importance	Average Importance  among the general s  among the general s  Average Importance among employers &  Average Importance ents for classes:  Average Importance mic standards and ch  Average Importance	4 Quite Important  student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important	5 Most Important  : 5 Most Important  ty: 5 Most Important  5 Most Important  to achieve: 5 Most Important
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atten 1 Least important  9) Programm 1 Least important	2 Low Importance  TUTIONAL FACTOR TO of the institution 2 Low Importance To of the institution 2 Low Importance Todance requirement 2 Low Importance Es set high acade 2 Low Importance	3 Average Importance  2S Average Importance 3 Average Importance 4 Average Importance 5 Average Importance 6 Average Importance 7 Average Importance	4 Quite Important  student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important  tivities for student	5 Most Important  : 5 Most Important  fy: 5 Most Important  5 Most Important  nt to achieve: 5 Most Important  s:
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atten 1 Least important  9) Programm 1 Least important  10) The opport 1	2 Low Importance  TUTIONAL FACTOR TO of the institution 2 Low Importance To of the institution 2 Low Importance Todance requirement 2 Low Importance Es set high acade 2 Low Importance	Average Importance  among the general s  among the general s  Average Importance among employers &  Average Importance ents for classes:  Average Importance mic standards and ch  Average Importance	4 Quite Important  student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important	5 Most Important  : 5 Most Important  ty: 5 Most Important  5 Most Important  to achieve: 5 Most Important
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atten 1 Least important  9) Programm 1 Least important  10) The opport 1 Least important	2 Low Importance  TUTIONAL FACTOR TO of the institution 2 Low Importance To of the institution 2 Low Importance To dance requirement 2 Low Importance Es set high acade 2 Low Importance tunities provided 2 Low Importance	3 Average Importance  2S Average Importance 3 Average Importance 4 Average Importance 5 Average Importance 6 Average Importance 7 Average Importance 8	4 Quite Important  Student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important  tivities for student  4 Quite Important	5 Most Important  5 Most Important  79: 5 Most Important  5 Most Important  nt to achieve: 5 Most Important  s: 5 Most Important  Most Important  5 Most Important  5 Most Important
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atter 1 Least important  9) Programm 1 Least important  10) The opport 1 Least important  11) Overall hou	2 Low Importance 2 Low Importance 2 Low Importance 3 Low Importance 4 Low Importance 6 Low Importance 7 Low Importance 8 Low Importance 9 Low Importance 1 Low	Average Importance  among the general s  among the general s  Average Importance among employers &  Average Importance ants for classes:  Average Importance mic standards and ch  Average Importance for extracurricular ac  Average Importance for extracurricular ac  Average Importance  average Importance for extracurricular ac  average Importance ou consider all the about	4 Quite Important  student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important  tivities for student 4 Quite Important  4 Quite Important  4 Quite Important  4 Quite Important  6 4 4 4 4 4 4 4 4 4 6 6 6 7 7 8 7 8 8 7 8 8 8 8 8 8 8 8 8 8	5 Most Important  : 5 Most Important  :y: 5 Most Important  5 Most Important  nt to achieve: 5 Most Important  s: 5 Most Important
1 Least important  G. INSTIT  6) Reputation 1 Least important  7) Reputation 1 Least important  8) Strict atter 1 Least important  9) Programm 1 Least important  10) The opport 1 Least important  11) Overall hou	2 Low Importance  TUTIONAL FACTOR TO of the institution 2 Low Importance To of the institution 2 Low Importance To dance requirement 2 Low Importance Es set high acade 2 Low Importance tunities provided 2 Low Importance	Average Importance  among the general s  among the general s  Average Importance among employers &  Average Importance ents for classes:  Average Importance mic standards and ch  Average Importance for extracurricular ac  Average Importance	4 Quite Important  Student population 4 Quite Important  general communit 4 Quite Important  4 Quite Important  allenges for student 4 Quite Important  tivities for student  4 Quite Important	Most Important  S Most Important  S Most Important  S Most Important  S Most Important  At to achieve:  Most Important  S Most Important

Most Important

12) The coverage and currency of the curriculum and content of the course:

1 2 3 4

**CURRICULUM AND CONTENT** 

Least important Low Importance

13) The emphas	is of the curricu	lum on subject speci	fic knowledge:	
1 13) Trie emprias	or the curricu	am on subject specif	nc Knowledge.	5
Least important	Low Importance	Average Importance	Ouite Important	Most Important
		curriculum & links to		Wost Important
1	niponents in the	2	$\Lambda$	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		e to the job market:	Quite Important	Wost Important
1	r the programm	e to the Job Market.	4	5
Least important	Low Importance	Average Importance	Ouite Important	Most Important
		g skills for the work p		
		team work, analysis,		en management,
1	)	3	4	5
Least important	Low Importance	Average Importance	Ouite Important	Most Important
		ou consider all the ab		
be:				
1	2	3	4	5
Least important	Low Importance	Average Importance	Ouite Important	Most Important
H. RESOUR				
		such as classrooms,	labs and lecture h	alls:
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	nd quality of lib			
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	out and appeara			
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		recreation facilities:	,	
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		classroom or the sta		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		ou consider all the ab		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	•		•	•
I. TEACHIN	IG AND LEARNIN	NG EXPERIENCE		
24) Ability of led	cturers to under	stand individual stud	ent needs:	
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
25) The teaching	a mothade and t			most important
	y memous and t	eaching styles of the	•	Wost Important
1	g memous and t 2	eaching styles of the	•	5
1 Least important	2 Low Importance	3 Average Importance	lecturers: 4 Quite Important	5 Most Important
	2 Low Importance	3	lecturers: 4 Quite Important	5 Most Important
	2 Low Importance	3 Average Importance	lecturers: 4 Quite Important	5 Most Important
26) <i>The ability o</i> 1 Least important	2 Low Importance of the lecturer to 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance	lecturers: 4 Quite Important nanner accessible t 4 Quite Important	5 <u>Most Important</u> to all students; 5 Most Important
26) <i>The ability o</i> 1 Least important	2 Low Importance of the lecturer to 2 Low Importance	3 Average Importance explain topics in a n 3	lecturers: 4 Quite Important nanner accessible t 4 Quite Important	5 <u>Most Important</u> to all students; 5 Most Important
26) The ability of 1 Least important 27) Teachers tea	2 Low Importance of the lecturer to 2 Low Importance aching on a prog	3 Average Importance o explain topics in a m 3 Average Importance	lecturers: 4 Quite Important nanner accessible t 4 Quite Important	5 <u>Most Important</u> to all students; 5 Most Important
26) The ability of 1 Least important 27) Teachers teach 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance	lecturers:  4 Quite Important hanner accessible to 4 Quite Important teaching styles/m 4 Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important
26) The ability of 1 Least important 27) Teachers teach 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3	lecturers:  4 Quite Important hanner accessible to 4 Quite Important teaching styles/m 4 Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important
26) The ability of 1 Least important 27) Teachers teach 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance	lecturers:  4 Quite Important hanner accessible to 4 Quite Important teaching styles/m 4 Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance	3 Average Importance  o explain topics in a m 3 Average Importance  gramme have similar 3 Average Importance in terms of number of 3 Average Importance	lecturers:  4 Quite Important  4 Quite Important  teaching styles/m  4 Quite Important  f years and level of  Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance	3 Average Importance  o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of	lecturers:  4 Quite Important  4 Quite Important  teaching styles/m  4 Quite Important  f years and level of  Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important of teaching 5
26) The ability of 1 Least important 27) Teachers tea 1 Least important 28) The experien 1 Least important 29) Ability of lead	2 Low Importance 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motive 2	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest	lecturers:  4 Quite Important 4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4	Most Important fo all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of lead 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motiva 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance	lecturers:  4 Quite Important  4 Quite Important  teaching styles/m  4 Quite Important  f years and level of  Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important of teaching 5
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of lead 1 Least important	2 Low Importance 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motive 2	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance	lecturers:  4 Quite Important 4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4	Most Important fo all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of leat 1 Least important 30) The qualification	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motiva 2 Low Importance ations of lecture 2	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance ors: 3	lecturers:  4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4 Quite Important 4 Quite Important 4 Quite Important 6 in the subject: 4 Quite Important	Most Important fo all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of least important 30) The qualification 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motiva 2 Low Importance ations of lecture 2 Low Importance	3 Average Importance of explain topics in a m 3 Average Importance of armme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance ors: 3 Average Importance ors: 3 Average Importance	lecturers:  4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4 Quite Important 4 Quite Important 4 Quite Important	Most Important fo all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of least important 30) The qualification 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motiva 2 Low Importance ations of lecture 2 Low Importance	3 Average Importance o explain topics in a m 3 Average Importance gramme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance ors: 3	lecturers:  4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4 Quite Important 4 Quite Important 4 Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important  5 Most Important
26) The ability of 1 Least important 27) Teachers teat 1 Least important 28) The experient 1 Least important 29) Ability of least important 30) The qualification 1 Least important	2 Low Importance of the lecturer to 2 Low Importance aching on a prog 2 Low Importance nce of lecturers 2 Low Importance cturers to motiva 2 Low Importance ations of lecture ations of lecture 2 Low Importance ations importance ations of lecture 2 Low Importance ations of lecture 2 Low Importance ations of lecture 2 Low Importance ations an impressiv 2	3 Average Importance of explain topics in a m 3 Average Importance of armme have similar 3 Average Importance in terms of number of 3 Average Importance ate students' interest 3 Average Importance ors: 3 Average Importance ors: 3 Average Importance	lecturers:  4 Quite Important 4 Quite Important teaching styles/m 4 Quite Important of years and level of 4 Quite Important in the subject: 4 Quite Important 4 Quite Important 4 Quite Important	5 Most Important to all students; 5 Most Important ethods: 5 Most Important of teaching 5 Most Important  5 Most Important

behavioura	l skills:			
1	2	3	4	5
		Average Importance	Quite Important	Most Important
		udents to a student's	overall learning	experience inside &
outside the	classroom:	_		_
1 1	2	3	4	5
		Average Importance	Quite Important	Most Important
	es provided for l	interactive student ce	entred learning ex	periences in the
classroom:	0	0	4	-
		3 A	4	Most Important
		Average Importance	Quite Important	Most Important
subject ma		s outside the classro	om to read and ur	iderstand the
1	າ eriai.	2	1	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		t systems available to		Wost Important
1	ncademic support	3 systems available to	Δ	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		anise, communicate		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		ping skills required for		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
•	•	5 1	,	•
39) Overall hov	v important do yo	ou consider the teachi	ing & learning crit	eria to be:
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	•	-		•
J. OUTCOM	MES AND ASSESSI	MENT		
40) Achieving h	igh grades in ass	sessments:		
1	2	3	4	5
		Average Importance	Quite Important	Most Important
41) <b>The variety</b>	of assessment m	nethods:		
1 1	2	3	4	5
		Average Importance	Quite Important	Most Important
42) <b>Continuous</b>	assessments in a	a module requiring st	udents to work co	ntinuously:
l coot imam =t = - '	L out I have a mt =	Avorage Imame t	Quito Imama at a card	Moot Immont
		Average Importance	Quite Important	Most Important
	· -	udents on assessmen	ι perτormance:	E
<u> </u>	Low Importance	3 Average Importance	4 Quite Important	O Most Important
	•	ards for student perfo		Most Important
1 147 A33C33IIIEII	ıs serriiyii Stailüi ?	arus ror student pent ર	ormance: 4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		nents on developing s		
1	) 2	nems on developing s	A	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		ments on subject spe		most important
1	) 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2	эттэ эт э <b>аы</b> јест эре 3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		ou consider the outco		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		oragoiportanoo	_a.toportant	octportant
K. To what exte	ent do vou agree.	disagree with the fol	llowing statement	s?
		disagree with the fol		
48) It is more in	mportant for stud	disagree with the foldents to cope with the		
	mportant for stud			
48) It is more in	mportant for stud ord. 2	dents to cope with the	eir courses easily l	rather than setting

32) Ability of lecturers to make students think, to change attitudes and develop

49) Examination is	s the best method	d to test student	ts knowledge and sl	kills.
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
50) Students have	the capacity to e	evaluate the tea	ching they receive.	
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
51) It is the institu	ution's responsibl	ility rather than	the students' to en	sure students are
successful in t	heir studies	-		
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
52) Attracting high	h quality student	input is a "nece	essary" condition fo	r ensuring quality in
higher educati	ion.	•	•	
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
53) The quality of	the student intak	ce is not importa	ant; the focus shoul	d be on the "value
added" to the	student (compari	ison of before a	nd after).	
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
54) The profession	nal and self devel	opment of stude	ents during the prod	ess of higher
education is m	nore important th	an the grades a	chieved by the stud	lent.
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
55) Using perform	ance indicators (	e.g. number of	graduates with high	marks, number of
computers to	students, student	to staff ratio) i	s an effective mean	s of measuring the
quality of an i				_
1	2	3	4	5
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree

### L. Which of the following statements best reflects your view of quality in higher education?

education?						
56) Providing education that is exceptional and has high standards.						
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
57) Achieving con	sistency in i	nternal processes.				
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
58) <b>Providing edu</b>	cation that r	neets specified objectiv	es or standar	ds.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
59) The ability of	59) The ability of the institution to be efficient and effective.					
1	2	3	4	5		
		Neutral	Agree	Strongly Agree		
		ution to be transformat	ive and to con	ntinually learn through		
empowermen	t and enhand	cement of all involved.				
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
61) <b>Providing edu</b>	cation and s	ervices that meets the	needs and exp	pectations of students.		
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
62) <b>Developing gr</b>	aduates who	o meet the expectations	s of the society	y.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
63) <b>Developing gr</b>	aduates who	o meet the requirement	s of employer	S.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
64) <b>Equipping stu</b>	dents with ti	he skills to learn and th	ink for thems	elves.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		

### THANK YOU

# APPENDIX 4 ACADEMIC STAFF PERCEPTION OF QUALITY IN HIGHER EDUCATION: QUESTIONNAIRE

The objective of this questionnaire is to determine what academics consider to be important in managing quality in higher education. The information that is provided will be used for research purposes only and will be kept <a href="entirely confidential">entirely confidential</a>.

A. Please indicate your formal role in the Institution

	Lecturer/Professor	Progra	amme Manager	Head of F	aculty/Depar	tment
	Any other: please spe	cify				
В.	Subject areas of teac	hing				
C.	Male □ Female □					
D.	Nationality					
E.	Number of years expe	rience in I	nigher education			
Cir	hich of the following rcle a number on a soportant.	cale from				
<b>F</b> . 1)	ADMISSION CR High admission crit		enterina a progra	mme:		
')	1 2	eria ioi e	3	11111 <del>6</del> .	ļ	5
	ast important Low Imp				Important	Most Important
2)	Testing students ap	otitude oi	r ability before en	try to a cou	ırse:	E
Le	2 ast important Low Imp	oortance	3 Average Importar	4 nce Quite	ı Important	5 Most Important
	Students on the pro					
	1 2		3	4		5
	ast important Low Imp		Average Importar		Important	Most Important
4)	The variety of prog	rammes/	/courses on oााer: २	Δ	Ĺ	5
Le	ast important Low Imp	ortance	Average Importar	nce Quite	r Important	Most Important
5)	Overall how import					
١.	1 2		3	4		5
Le	ast important Low Imp	ortance	Average Importar	ice Quite	Important	Most Important
G.	INSTITUTIONAL	LFACTOR	es			
6)	Reputation of the in	nstitutior	among the gene	ral student	population:	
10	1 2	ortanco	3 Avorago Importar	4 Ouito	l Important	5 Most Important
	ast important Low Imp Reputation of the in		Average Importar			Most Important
'	1 2	istitutioi	3	4 <b>geriera</b>	  -	5
	ast important Low Imp		Average Importar	ice Quite	Important	Most Important
8)	Strict attendance re	equireme	ents for classes:	,	•	F
Le	ے ast important Low Imp	oortance	Average Importar	4 nce Ouite	Important	5 Most Important
	Programmes set hi					
	1 2		3	4	ļ	5
	ast important Low Imp		Average Importar		Important	Most Important
10	) The opportunities p	provided	for extracurricula	r activities	for students	s: 5
Le	ے ast important Low Imp	ortance	Average Importar	nce Ouite	Important	Most Important
	) Overall how import					
	1 2		3	4		5
Le	ast important Low Imp	ortance	Average Importar	ice Quite	Important	Most Important
G.	CURRICULUM A	ND CONT	ENT			
12	) The coverage and	currency	of the curriculum	and conte	nt of the cou	
l e	1 2 ast important Low Imp	oortance	3 Average Importar	4 nce Ouite	l Important	5 Most Important
	ast amportant Low min	or tarioo	, wordgo importar	ioo Quito	portant	ost important

10) 71			<del> </del>	
13) <i>The empha</i>	isis of the curricu	lum on subject specif	ric Knowledge:	F
Least important	Low Importance	Average Importance	4 Quite Important	อ Most Important
		curriculum & links to		wost important
14) 11 11 11	omponents in the 2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		e to the job market:		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		g skills for the work p		elf management,
communica	ation, creativity, t	team work, analysis,	application:	_
1	2	3	4	5
		Average Importance		
be:	w important do ye	ou consider all the ab	ove curriculum & c	ontent criteria to
<i>De.</i>	2	3	1	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
Loust important	Low Importance	Average importance	Quite Important	Wost Important
H. RESOU	RCES			
		s such as classrooms,	labs and lecture h	alls:
1	2	3	4	5
		Average Importance	Quite Important	Most Important
19) The range	and quality of lib	orary resources:		
1	2	3	4	5
		Average Importance	Quite Important	Most Important
20) Campus lay	yout and appeara	nce:		-
Locat important	Low Importance	J Average Impertance	4 Ouita Important	5 Most Important
		Average Importance recreation facilities:	Quite Important	wost important
1	isive sports and i	ecreation racinties.	1	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		a classroom or the sta		Wost Important
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
23) Overall hou	พ important do yo	ou consider all the ab	ove resources crite	eria to be:
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
TEACHI	NC AND LEADNIN	NO EVDEDIENCE		
1. TEACHI	NG AND LEARNIN	Stand individual stud	ont needs.	
24) <b>ADIIILY OF 16</b>	ecturers to unaer.	Stand maividuai Studi	ent neeas: 4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		reaching styles of the		Wost Important
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		explain topics in a m		
1	2	3	4	5
	Low Importance	Average Importance	Quite Important	Most Important
27) Teachers to	eaching on a prog	gramme have similar	teaching styles/m	ethods:
1	. 2	3	4	5
	Low Importance	Average Importance	Quite Important	Most Important
28) Ine experie	ence of lecturers	in terms of number o	ा years and level o	τ teaching -
Loast important	Low Importance	J Avorago Importance	Ouito Important	Most Important
	Low Importance	Average Importance ate students' interest	Quite Important	Most Important
1	) 2	3	A	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	cations of lecture			
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		e and warm personal		-
11	2	3	4	5

I Least important Low Importance		<b>~</b>	
	Average Importance		
32) Ability of lecturers to make	e students think, to ch	ange attitudes and	l develop
behavioural skills:			
1 2	3	4	5
Loast important Low Importance	Average Importance	Quito Important	Most Important
Least important Low Importance	Average Importance	Quite Important	
33) The contribution of other s	students to a student's	s overall learning (	experience inside &
outside the classroom:			
1 2	3	4	5
Least important Low Importance	Average Importance	Ouite Important	Most Important
34) Opportunities provided for			
	interactive student co	entred learning exp	beriefices in the
classroom:			_
1 2	3	4	5
Least important Low Importance	Average Importance	Quite Important	Most Important
35) The effort put in by studen	ts outside the classro	oom to read and ur	nderstand the
subject material:			
3ubject material.	3	4	F
	3	4	5
Least important Low Importance		Quite Important	Most Important
36) Additional academic support	rt systems available to	o students:	
1 2	3	4	5
Least important Low Importance	Average Importance	Quite Important	Most Important
37) Ability of the lecturer to or	yanise, communicate	assess their sub	yeci areas: _
1 2	3	4	5
Least important Low Importance	Average Importance	Quite Important	Most Important
38) Focus of lecturers on devel	opina skills reauired f	or the workplace:	
1 2	3	Δ	5
Loost important Low Importance	Average Importance	Ouita Important	Most Important
Least important Low Importance	Average Importance	Quite Important	wost important
39) Overall how important do y	ou consider the teach	ing & learning crite	eria to be:
1 2	3	4	5
Least important Low Importance	Average Importance	Quite Important	Most Important
Least important Low importance	Average importance	Quite important	Wost Important
J. OUTCOMES AND ASSESS	SMENT		
40) Achieving high grades in as		4	5
40) Achieving high grades in as	ssessments:	4	5 Most Important
40) Achieving high grades in as 1 2 Least important Low Importance	ssessments: 3 Average Importance	4 Quite Important	5 Most Important
40) Achieving high grades in as	ssessments: 3 Average Importance	4 Quite Important	·
40) Achieving high grades in as 1 2 Least important Low Importance	ssessments: 3 Average Importance	4 Quite Important 4	5 Most Important 5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment in the second se	Average Importance methods:	4 Quite Important  4 Quite Important	5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the second se	Average Importance methods: 3 Average Importance	4 Quite Important	5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in	Average Importance methods: 3 Average Importance	4 Quite Important	5 Most Important Intinuously:
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 1 2 Least important Low Importance 42) Continuous assessments in 1 2	Average Importance  a Average Importance  methods:  3 Average Importance  a a module requiring so	4 Quite Important tudents to work co 4	5 Most Important Intinuously: 5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance	Average Importance  3 Average Importance  3 Average Importance  4 a module requiring somethics  3 Average Importance	4 <u>Quite Important</u> tudents to work co 4 Quite Important	5 Most Important Intinuously:
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 1 2 Least important Low Importance 42) Continuous assessments in 1 2	Average Importance  3 Average Importance  3 Average Importance  4 a module requiring somethics  3 Average Importance	4 <u>Quite Important</u> tudents to work co 4 Quite Important	5 Most Important Intinuously: 5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance	Average Importance  3 Average Importance  3 Average Importance  4 a module requiring somethics  3 Average Importance	4 <u>Quite Important</u> tudents to work co 4 Quite Important	5 Most Important Intinuously: 5
40) Achieving high grades in as  1 2  Least important Low Importance 41) The variety of assessment in 1 2  Least important Low Importance 42) Continuous assessments in 1 2  Least important Low Importance 43) The feedback provided to s 1 2	Average Importance  3 Average Importance 3 Average Importance 4 a module requiring si 3 Average Importance tudents on assessmen	4 Quite Important tudents to work co 4 Quite Important of performance: 4	5 Most Important intinuously: 5 Most Important
40) Achieving high grades in as  1 2  Least important Low Importance  41) The variety of assessment in 2  Least important Low Importance  42) Continuous assessments in 1 2  Least important Low Importance  43) The feedback provided to s  1 2  Least important Low Importance	Average Importance  Average Importance  Average Importance  Average Importance  A module requiring so  Average Importance  tudents on assessment  Average Importance	4 Quite Important tudents to work co 4 Quite Important nt performance: 4 Quite Important	5  Most Important intinuously: 5  Most Important
40) Achieving high grades in as  1 2  Least important Low Importance 41) The variety of assessment in 1 2  Least important Low Importance 42) Continuous assessments in 1 2  Least important Low Importance 43) The feedback provided to s 1 2	Average Importance  Average Importance  Average Importance  Average Importance  A module requiring so  Average Importance  tudents on assessment  Average Importance	4 Quite Important tudents to work co 4 Quite Important nt performance: 4 Quite Important	5 Most Important Intinuously: 5 Most Important 5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the season of	Average Importance  methods:  3 Average Importance a module requiring si 3 Average Importance tudents on assessment 3 Average Importance tudents on assessment 3 Average Importance dards for student perfect 3	4 Quite Important fudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4	5 Most Important Intinuously: 5 Most Important 5 Most Important 5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the second of	Average Importance  methods:  3 Average Importance a module requiring si 3 Average Importance tudents on assessmen 3 Average Importance dards for student performance 3 Average Importance	4 Quite Important tudents to work co 4 Quite Important nt performance: 4 Quite Important ormance: 4 Quite Important	5 Most Important ontinuously: 5 Most Important  5 Most Important  5 Most Important  5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the seast important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to some 1 2 Least important Low Importance 44) Assessments set high standard 1 2	Average Importance  methods:  3 Average Importance a module requiring si 3 Average Importance tudents on assessmen 3 Average Importance dards for student performance 3 Average Importance	4 Quite Important tudents to work co 4 Quite Important nt performance: 4 Quite Important ormance: 4 Quite Important	5 Most Important ontinuously: 5 Most Important  5 Most Important  5 Most Important  5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the second of	Average Importance  methods:  3 Average Importance a module requiring si 3 Average Importance tudents on assessmen 3 Average Importance dards for student performance 3 Average Importance	4 Quite Important tudents to work co 4 Quite Important nt performance: 4 Quite Important ormance: 4 Quite Important	5 Most Important ontinuously: 5 Most Important  5 Most Important  5 Most Important  5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of varie	Average Importance  methods:  3 Average Importance a module requiring single a module requirements on developing single a module requirements on developing single a module requirements single a module requirement single a module	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work	Most Important  Intinuously:  5  Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of variety	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer 3 Average Importance  dards for student performance  sments on developing  3 Average Importance	4 Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important	5 Most Important place:
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of the variety of assessments in the variety of variety of assessments in the variety of variety of assessments of the variety of variety of variety of assessments of the variety of	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer 3 Average Importance  dards for student performance  sments on developing  3 Average Importance	4 Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important	5 Most Important place: 5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer 3 Average Importance  dards for student perf 3 Average Importance sments on developing 3 Average Importance sments on subject specifications	Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4	5 Most Important place: 5 Most Important 5
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of variety	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer 3 Average Importance  dards for student performance  sments on developing  3 Average Importance	4 Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important	5 Most Important place: 5 Most Important
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment of the variety of assessments in the variety of the variety of assessments in the variety of	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer  3 Average Importance  dards for student perf  3 Average Importance  sments on developing  3 Average Importance  sments on subject specifications  Average Importance	Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important	Most Important  formalise of the second seco
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer  3 Average Importance  dards for student perf  3 Average Importance  sments on developing  3 Average Importance  sments on subject specifications  Average Importance	Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important	Most Important  formalise of the second seco
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1	Average Importance  a Module requiring so a module requiring so a Module sees sees sees a Average Importance  Average Importance tudents on assessment a Average Importance dards for student performance sees sees a Average Importance sees sees and average Importance sees and average sees and average Importance sees and average Importance	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important ormance: 4 Quite Important comportant comportant comportant comportant desiric knowledge: 4 Quite Important comportant desiric knowledge: 4 Quite Important comportant desiric knowledge: 4 Quite Important comportant	Most Important  formal street of the street
40) Achieving high grades in as  1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance	Average Importance  methods:  3 Average Importance  a module requiring si 3 Average Importance  tudents on assessmer  3 Average Importance  dards for student perf  3 Average Importance  sments on developing  3 Average Importance  sments on subject specifications  3 Average Importance	Quite Important tudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important	Most Important  South I
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance	Average Importance Table 1 Average Importance Table 2 Average Importance Table 3 Average Importance	Quite Important Audents to work co 4 Quite Important At performance: 4 Quite Important At Quite Important	Most Important  Most Important  Most Important   Most Important   Most Important   Most Important   Most Important    Most Important    Most Important    Most Important    Most Important   Most Important   Most Important   Most Important   Most Important   Most Important   Most Important
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance 47) Coverall how importance	Average Importance Table 1 Average Importance Table 2 Average Importance Table 3 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance	Quite Important fudents to work co 4 Quite Important at performance: 4 Quite Important formance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important formes & assessment 4 Quite Important Important	Most Important Intinuously:  5 Most Important  5 Most Important  5 Most Important  place:  5 Most Important  5 Most Important  t criteria to be:  5 Most Important  st criteria to be:  5 Most Important
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance	Average Importance Table 1 Average Importance Table 2 Average Importance Table 3 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance	Quite Important fudents to work co 4 Quite Important at performance: 4 Quite Important formance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important formes & assessment 4 Quite Important Important	Most Important Intinuously:  5 Most Important  5 Most Important  5 Most Important  place:  5 Most Important  5 Most Important  t criteria to be:  5 Most Important  st criteria to be:  5 Most Important
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance 48) It is more important for sta	Average Importance Table 1 Average Importance Table 2 Average Importance Table 3 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance	Quite Important fudents to work co 4 Quite Important at performance: 4 Quite Important formance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important formes & assessment 4 Quite Important Important	Most Important Intinuously:  5 Most Important  5 Most Important  5 Most Important  place:  5 Most Important  5 Most Important  t criteria to be:  5 Most Important  st criteria to be:  5 Most Important
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance 48) It is more important for standigh standard.	Average Importance Tamethods:  3 Average Importance Tamethods:  3 Average Importance Tamethods:  3 Average Importance Tudents on assessment  3 Average Importance Tudents for student perform  3 Average Importance Tamethod Importance Tamethod Importance Tamethod Tam	Quite Important fudents to work co 4 Quite Important of performance: 4 Quite Important ormance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important omes & assessment 4 Quite Important Illowing statement eir courses easily in	Most Important  Most Important  Most Important   Most Important   Most Important   Most Important   Most Important    Most Important    Most Important    Most Important    Most Important    Most Important
40) Achieving high grades in as 1 2 Least important Low Importance 41) The variety of assessment 1 2 Least important Low Importance 42) Continuous assessments in 1 2 Least important Low Importance 43) The feedback provided to s 1 2 Least important Low Importance 44) Assessments set high stand 1 2 Least important Low Importance 45) The emphasis of the assess 1 2 Least important Low Importance 46) The emphasis of the assess 1 2 Least important Low Importance 47) Overall how important do y 1 2 Least important Low Importance 48) It is more important for sta	Average Importance Table 1 Average Importance Table 2 Average Importance Table 3 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance Table 4 Average Importance	Quite Important fudents to work co 4 Quite Important at performance: 4 Quite Important formance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important formes & assessment 4 Quite Important Important	Most Important Intinuously:  5 Most Important  5 Most Important  5 Most Important  place:  5 Most Important  5 Most Important  t criteria to be:  5 Most Important  st criteria to be:  5 Most Important

49) Examination is	49) Examination is the best method to test students knowledge and skills.					
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
50) Students have	the capacity to	o evaluate the teac	hing they receive			
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
51) It is the institu	ution's respons	ibility rather than t	he students' to e	nsure students are		
successful in t	heir studies					
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
52) Attracting high	h quality stude	nt input is a "neces	sary" condition f	or ensuring quality in		
higher educati	ion.					
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
				ıld be on the "value		
added" to the	student (comp	arison of before an	d after).			
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
		elopment of studer				
education is m	nore important	than the grades ac	hieved by the stu	dent.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
55) Using perform	ance indicators	s (e.g. number of g	raduates with hig	h marks, number of		
computers to	students, stude	ent to staff ratio) is	an effective mea	ns of measuring the		
quality of an i	nstitution.					
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		

# L. Which of the following statements best reflects your view of quality in higher education?

education?						
56) Providing education that is exceptional and has high standards.						
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
57) Achieving con	sistency in ii	nternal processes.				
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		
58) <b>Providing edu</b>	cation that n	neets specified objectiv	es or standaı	rds.		
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
59) The ability of	the institutio	on to be efficient and ef	fective.			
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
	60) The capacity of the institution to be transformative and to continually learn through					
empowermen	t and enhand	cement of all involved.				
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
61) <b>Providing edu</b>	cation and s	ervices that meets the i	needs and ex	pectations of students.		
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
62) <b>Developing gr</b>	aduates who	meet the expectations	of the societ	ty.		
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
63) <b>Developing gr</b>	aduates who	meet the requirements	s of employe	rs.		
1	2	3	4	5		
Strongly Disagree		Neutral	Agree	Strongly Agree		
64) Equipping stu	dents with th	he skills to learn and the	ink for thems	selves.		
1	2	3	4	5		
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree		

THANK YOU

# APPENDIX 5 EMPLOYER PERCEPTION OF QUALITY IN HIGHER EDUCATION: QUESTIONNAIRE

The objective of this questionnaire is to determine what employers think is important in managing quality in higher education. The information that is provided will be used for research purposes only and will be kept <u>entirely confidential</u>.

A. Please indic	cate your official	designation		
	-			
	_			
·				
D. Male $\square$ Fe	emale $\square$			
E. Nationality				
Which of the	following criter	ria is most importar	nt to you as an er	mployer of
<b>Business and</b>	IT graduates?	Circle a number or	a scale from 1 t	o 5 where
1=least impo	ortant and 5 = m	ost important.		
		_		
	SION CRITERIA		_	
1) High admis	SSION CRITERIA TOR (	entering a programme	e: 1	5
	Low Importance	Average Importance	Quite Important	Most Important
		r ability before entry		wost important
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		e have good attitude a		
1	2	3	4	5
		Average Importance	Quite Important	Most Important
4) The variety	। of programmes	/courses on offer:		
1	2	3	4	5
		Average Importance		
5) Overall not	N important do y∈ ≏	ou consider all the ab	ove admission criti	<b>егіа то be:</b> 5
Least important	Low Importance	Average Importance	Quite Important	Most Important
Zodot III.portaiit	2011	, was ago mportaneo	Zuno mportam	octportaint
	UTIONAL FACTOR			
6) <b>Reputation</b>	of the institution	n among the general s	student population	<i>:</i>
1		3	4	5
		Average Importance	Quite Important	Most Important
/) <b>Reputation</b>	or the institution	n among employers &	general communit	t <b>y:</b> 5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	ndance requireme	<u> </u>	Quito Important	West Important
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
9) <b>Programm</b>	es set high acade	emic standards and ch	allenges for stude	nt to achieve:
1	2	3	4	5
	Low Importance	Average Importance	Quite Important	Most Important
10) The opport	unities provided	for extracurricular ac	tivities for student	'S <i>:</i>
Loost important	Low Importance	Avorago Importanco	4 Ouito Important	่ง Most Important
		Average Importance ou consider all the about	Quite Important	
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
<u> </u>	•	1 1	1	•
	CULUM AND CONT			
12) The covera	age and currency	of the curriculum and	d content of the co	
I 1	2	3	4	5

	Low Importance	Average Importance	Quite Important	Most Important
13) <b>The empha</b>	asis of the curricu	ılum on subject specii	fic knowledge:	
1	2	3	4	5
		Average Importance	Quite Important	Most Important
14) Practical c	omponents in the	curriculum & links to	the industry:	г
Loast important	Low Importance	Average Importance	Quite Important	b Most Important
		e to the job market:	Quite important	wost important
13) Kelevance	or the programm	2	Δ	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		g skills for the work p		
communic	ation, creativity,	team work, analysis,	application:	-
1	2	3	4	5
		Average Importance		Most Important
	w important do y	ou consider all the ab	ove curriculum & d	content criteria to
be:	2	2	4	E
Least important	Low Importance	Average Importance	Quite Important	b Most Important
Least important	Low Importance	Average importance	Quite Important	wost important
H. RESOU	RCES			
		s such as classrooms,	labs and lecture h	alls:
1	2	3	4	5
		Average Importance	Quite Important	Most Important
19) <i>The range</i>	and quality of lib	orary resources:		
1	. 2	3	4	5
		Average Importance	Quite Important	Most Important
20) Campus Ia	yout and appeara	ince:	4	F
Loast important	Low Importance	Average Importance	4 Ouita Important	5 Most Important
·	Low Importance	recreation facilities:	Quite Important	wost important
1	risive sports and i	que la contraction de la contr	Δ	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		a classroom or the sta		
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
23) Overall ho	w important do y	ou consider all the ab	ove resources crite	eria to be:
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
I TEACHI	INC AND LEADALL	NO EVDEDIENCE		
	ING AND LEARNII		ont noods:	
24) <b>ADITICY OF I</b>	ectarers to unider ?	stand individual stud 3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
		teaching styles of the		octportant
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
26) The ability	of the lecturer to	explain topics in a n	nanner accessible t	
1	2	3	4	5
	Low Importance	Average Importance	Quite Important	Most Important
27) <b>Teachers t</b>	eaching on a prog	gramme have similar	teaching styles/m	_
Loost important	2	Avorago Impartance	4	5 Most Important
	Low Importance	Average Importance	Quite Important	Most Important
20) The experi	ence of lecturers	in terms of number of	ı years and level O	r reacrining
Least important	Low Importance	Average Importance	Quite Important	Most Important
		ate students' interest		oct important
1	2	3	4	5
Least important	Low Importance	Average Importance	Quite Important	Most Important
	cations of lecture		•	
1 4	2	3	4	5
l I	2	3	7	· ·
	Low Importance	Average Importance	Quite Important	Most Important
		Average Importance ve and warm personal		

'	2	3	4	5
		Average Importance		
		students think, to cha	ange attitudes and	develop
behaviour	al skills:			_
l cost improvement	2	3	4	5
		Average Importance		Most Important
	bution of other si e classroom:	tudents to a student's	overali learning e	experience inside &
outside in	e ciassi oom:	2	4	5
Least important	t Low Importance	Average Importance	Ouite Important	Most Important
		interactive student ce		
classroom		merative stadem of	introd todrining exp	ocricioes in the
1	2	3	4	5
Least important	t Low Importance	Average Importance	Quite Important	Most Important
		ts outside the classro		
subject ma				
1	2	3	4	5
		Average Importance		Most Important
36) <b>Additional</b>	academic suppor	t systems available to	students:	
1	2	3	4	5
		Average Importance		Most Important
37) Ability of t	the lecturer to org	ganise, communicate	& assess their sub	ject areas:
	L Low Importance	Average Importance	4 Ouita Important	Most Important
		Average Importance oping skills required for		Most Important
1	on develo	pping skins required to	or the workplace:	F
Least important	t Low Importance	Average Importance	Quite Important	Most Important
Loust important	Low Importance	Average importance	Quite important	wost important
39) Overall ho	w important do ve	ou consider the teach	ina & learnina crit	eria to be:
1	2	3	4	5
Least important	t Low Importance	Average Importance	Quite Important	Most Important
	MES AND ASSESS			
	MES AND ASSESS high grades in as			_
40) <b>Achieving</b> 1	high grades in as	sessments: 3	4	5
40) <b>Achieving</b> 1 Least important	high grades in as: 2 t Low Importance	sessments: 3 Average Importance	4 Quite Important	5 Most Important
40) <b>Achieving</b> 1 Least important	high grades in as	sessments: 3 Average Importance nethods:	4 Quite Important	· ·
40) Achieving 1 Least important 41) The variet	high grades in as: 2 t Low Importance y of assessment n 2	sessments: 3 Average Importance nethods: 3	4	Most Important 5
40) Achieving 1 Least important 41) The variet 1 Least important	high grades in as: 2 t Low Importance y of assessment n 2 t Low Importance	Sessments: 3 Average Importance nethods: 3 Average Importance	4 Quite Important	Most Important  5  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important	high grades in as: 2 t Low Importance y of assessment n 2 t Low Importance	sessments: 3 Average Importance nethods: 3	4 Quite Important	Most Important  5  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1	high grades in as: 2 t Low Importance y of assessment n 2 t Low Importance s assessments in 2	Average Importance nethods: 3 Average Importance a module requiring st	4 Quite Important tudents to work co 4	Most Important  5 Most Important  ntinuously: 5
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important	high grades in as: 2 t Low Importance y of assessment n 2 t Low Importance is assessments in 2 t Low Importance	Average Importance  a Average Importance  nethods:  3 Average Importance  a module requiring st  3 Average Importance	4 Quite Important tudents to work co 4 Quite Important	Most Important  5  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important	high grades in as: 2 t Low Importance y of assessment n 2 t Low Importance is assessments in 2 t Low Importance	Average Importance nethods: 3 Average Importance a module requiring st	4 Quite Important tudents to work co 4 Quite Important	Most Important  5 Most Important  ntinuously: 5
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedboom 1	high grades in as:  2 t Low Importance y of assessment n 2 t Low Importance s assessments in 2 t Low Importance ack provided to st	Average Importance  a Average Importance  nethods:  3 Average Importance  a module requiring st  3 Average Importance	4 Quite Important tudents to work co 4 Quite Important	Most Important  5  Most Important  ntinuously: 5  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedba 1 Least important	high grades in as:  2 t Low Importance y of assessment n 2 t Low Importance s assessments in 2 t Low Importance ack provided to st 2 t Low Importance	Average Importance  a Average Importance  Average Importance  a module requiring st  3  Average Importance  4  Average Importance  5  4  4  4  5  4  5  6  6  6  6  6  6  6  6  6  6  6  6	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedba 1 Least important	high grades in as:  2 t Low Importance y of assessment n 2 t Low Importance s assessments in 2 t Low Importance ack provided to st 2 t Low Importance	Average Importance  a Average Importance  Average Importance  a module requiring st  3  Average Importance  fudents on assessment  3  Average Importance	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessment 1 Least important	high grades in as:  2 t Low Importance y of assessment in 2 t Low Importance s assessments in 2 t Low Importance ack provided to st 2 t Low Importance ints set high stand 2 t Low Importance	Average Importance	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important ormance: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessment 1 Least important	high grades in as:  2 t Low Importance y of assessment in 2 t Low Importance s assessments in 2 t Low Importance ack provided to st 2 t Low Importance ints set high stand 2 t Low Importance	Average Importance  3  Average Importance  3  Average Importance  3  Average Importance  3  Average Importance	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important ormance: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho	high grades in as:  2 t Low Importance y of assessment in 2 t Low Importance is assessments in 2 t Low Importance ack provided to st 2 t Low Importance ints set high stand 2 t Low Importance assis of the assessi	Average Importance  a Module requiring standards on assessment  Average Importance  a Module requiring standards on assessment  Average Importance  a Module requiring standards on assessment  Average Importance  ards for student performance  ards for student performance  ments on developing  3	4 Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  5 Most Important  place: 5
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important	t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance asis of the assessi 2 t Low Importance	Average Importance  a Module requiring standards on assessment  Average Importance  a module requiring standards on assessment  Average Importance  a Average Importance  ments on developing  Average Importance	4 Quite Important rudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important	t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance asis of the assessi 2 t Low Importance	Average Importance  a Module requiring standards on assessment  Average Importance  a Module requiring standards on assessment  Average Importance  a Module requiring standards on assessment  Average Importance  ards for student performance  ards for student performance  ments on developing  3	4 Quite Important rudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important 46) The empho 1	t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assessi 2	Average Importance a module requiring statements on assessment a Average Importance a worage Importance a module requiring statements on assessment a Average Importance ards for student performance a Average Importance	4 Quite Important rudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbox 1 Least important 44) Assessmen 1 Least important 45) The emphox 1 Least important 46) The emphox 1 Least important 46) The emphox 1 Least important	t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 2 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 1 t Low Importance	Average Importance methods:  Average Importance a module requiring st  Average Importance a module requiring st  Average Importance a Average Importance a Average Importance brack for student performance a Average Importance ments on developing  Average Importance ments on subject speciments on subject spec	4 Quite Important rudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbox 1 Least important 44) Assessmen 1 Least important 45) The emphox 1 Least important 46) The emphox 1 Least important 46) The emphox 1 Least important	t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 2 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 1 t Low Importance	Average Importance a module requiring statements on assessment a Average Importance a worage Importance a module requiring statements on assessment a Average Importance ards for student performance a Average Importance	4 Quite Important rudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbox 1 Least important 44) Assessmen 1 Least important 45) The emphox 1 Least important 46) The emphox 1 Least important 47) Overall ho	t Low Importance	Average Importance  a module requiring st  Average Importance  a module requiring st  Average Importance  a widents on assessment  Average Importance  ards for student perfor  Average Importance  ments on developing  Average Importance  ments on subject speciments	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important crific knowledge: 4 Quite Important ormance: 4 Quite Important crific knowledge: 4 Quite Important ormance: 4 Quite Important ormance: 4 Quite Important ormance: 4 Quite Important ormance: 4	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  criteria to be: 5
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbox 1 Least important 44) Assessmen 1 Least important 45) The emphox 1 Least important 46) The emphox 1 Least important 47) Overall ho 1 Least important	t Low Importance	Average Importance  a module requiring st  a Average Importance  a module requiring st  3  Average Importance  a definition of the students on assessment  3  Average Importance  a verage Importance  b ments on developing  3  Average Importance  ments on subject specification  3  Average Importance  ments on subject specification  3  Average Importance  con consider the outcome  3  Average Importance	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important ormance: 4 Quite Important skills for the work 4 Quite Important ecific knowledge: 4 Quite Important ormes & assessmen 4 Quite Important	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  traciteria to be: 5 Most Important
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important 46) The empho 1 Least important 47) Overall ho 1 Least important 47) Overall ho 1 Least important	t Low Importance t Low Importance t sassessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assess 1 t Low Importance tasis of the assess 2 t Low Importance the importance tent do you agree	Average Importance  a module requiring st  a Average Importance  a module requiring st  3  Average Importance  a definition of the state of the stat	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important cormance: 4 Quite Important skills for the work, 4 Quite Important ecific knowledge: 4 Quite Important comes & assessment 4 Quite Important Illowing statement	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  cit criteria to be: 5 Most Important  services and services are services and services are services are services and services are servic
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important 47) The empho 1 Least important 47) Overall ho 1 Least important 48) It is more	t Low Importance t Low Importance t sassessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assess 2 t Low Importance the important for study	Average Importance  a module requiring st  a Average Importance  a module requiring st  3  Average Importance  a definition of the students on assessment  3  Average Importance  a verage Importance  b ments on developing  3  Average Importance  ments on subject specification  3  Average Importance  ments on subject specification  3  Average Importance  con consider the outcome  3  Average Importance	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important cormance: 4 Quite Important skills for the work, 4 Quite Important ecific knowledge: 4 Quite Important comes & assessment 4 Quite Important Illowing statement	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  cit criteria to be: 5 Most Important  services and services are services and services are services are services and services are servic
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important 46) The empho 1 Least important 47) Overall ho 1 Least important 47) Overall ho 1 Least important	t Low Importance t Low Importance t sassessments in 2 t Low Importance ts assessments in 2 t Low Importance tack provided to st 2 t Low Importance this set high stand 2 t Low Importance tasis of the assess 2 t Low Importance the important for study	Average Importance  a module requiring st  a Average Importance  a module requiring st  3  Average Importance  a definition of the state of the stat	Quite Important tudents to work co 4 Quite Important at performance: 4 Quite Important cormance: 4 Quite Important skills for the work, 4 Quite Important ecific knowledge: 4 Quite Important comes & assessment 4 Quite Important Illowing statement	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  cit criteria to be: 5 Most Important  services and services are services and services are services are services and services are servic
40) Achieving 1 Least important 41) The variet 1 Least important 42) Continuou 1 Least important 43) The feedbo 1 Least important 44) Assessmen 1 Least important 45) The empho 1 Least important 47) The empho 1 Least important 47) Overall ho 1 Least important 48) It is more	t Low Importance t Low Importance t sassessments in 2 t Low Importance t sassessments in 2 t Low Importance	Average Importance a module requiring st a Average Importance a module requiring st a Average Importance a widents on assessment a Average Importance but a student performance a average Importance a student performance a	Quite Important at performance: 4 Quite Important at performance: 4 Quite Important at performance: 4 Quite Important askills for the work 4 Quite Important acific knowledge: 4 Quite Important ames & assessment 4 Quite Important Illowing statement acific courses easily in	Most Important  5 Most Important  ntinuously: 5 Most Important  5 Most Important  5 Most Important  place: 5 Most Important  5 Most Important  t criteria to be: 5 Most Important  seriather than setting

49) Examination is	s the best me	ethod to test students	knowledge and	l skills.	
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
50) Students have	e the capacity	v to evaluate the teach	ing they receiv	re.	
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
51) It is the institu	ution's respo	nsibility rather than th	he students' to	ensure students are	
successful in t	their studies.				
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
52) Attracting high	h quality stud	dent input is a "necess	sary" condition	for ensuring quality in	
higher educati	ion.				
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
53) The quality of	the student	intake is not importan	t; the focus sho	ould be on the "value	
added" to the	student (con	nparison of before and	l after).		
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
		levelopment of studen			
education is more important than the grades achieved by the student.					
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
55) Using perform	nance indicate	ors (e.g. number of gr	aduates with h	igh marks, number of	
computers to	students, stu	ident to staff ratio) is	an effective me	eans of measuring the	
quality of an i	nstitution.				
1	2	3	4	5	
Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	

### L. Which of the following statements best reflects your view of quality in higher education?

56) Providing education that is exceptional and has high standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  57) Achieving consistency in internal processes.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 4 5	education?					
57) Achieving consistency in internal processes.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree	56) Providing education that is exceptional and has high standards.					
57) Achieving consistency in internal processes.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 4 5	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 59) The ability of the institution to be efficient and effective. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 4 5	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Developing straduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	57) Achieving con	sistency in ir	nternal processes.			
58) Providing education that meets specified objectives or standards.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Developing straduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 59) The ability of the institution to be efficient and effective. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	
59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree	58) <b>Providing edu</b>	cation that n	neets specified objectiv	es or standar	ds.	
59) The ability of the institution to be efficient and effective.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5					Strongly Agree	
60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	59) The ability of	the institutio	n to be efficient and el	fective.		
60) The capacity of the institution to be transformative and to continually learn through empowerment and enhancement of all involved.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 61) Providing education and services that meets the needs and expectations of students. 1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Developing graduates who meet the requirements of employers. 1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	1	2	3	4	5	
empowerment and enhancement of all involved.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5					03 0	
Strongly Disagree Disagree Neutral Agree Strongly Agree  61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  62) Developing graduates who meet the expectations of the society.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  63) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree  64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5				ve and to cont	inually learn through	
61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5	empowerment and enhancement of all involved.					
61) Providing education and services that meets the needs and expectations of students.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Peveloping graduates who meet the expectations of the society.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Peveloping graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 62) Developing graduates who meet the expectations of the society. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5					03 0	
62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	61) <b>Providing edu</b>	cation and se	ervices that meets the	needs and exp	ectations of students.	
62) Developing graduates who meet the expectations of the society.  1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 63) Developing graduates who meet the requirements of employers. 1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5				J	0 3 0	
63) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	62) <b>Developing gr</b>	aduates who	meet the expectations	s of the society	<b>y</b> .	
63) Developing graduates who meet the requirements of employers.  1 2 3 4 5  Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5	1	2	3	4	5	
1 2 3 4 5 Strongly Disagree Disagree Neutral Agree Strongly Agree 64) Equipping students with the skills to learn and think for themselves. 1 2 3 4 5				J		
64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5	63) Developing graduates who meet the requirements of employers.					
64) Equipping students with the skills to learn and think for themselves.  1 2 3 4 5	1	2	3	4	5	
1 2 3 4 5	7	.,				
1 2 3 4 5	64) Equipping students with the skills to learn and think for themselves.					
	1	2	3	4	5	
Strongly Disagree Disagree Neutral Agree Strongly Agree	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree	

THANK YOU